Rational and non-rational two-dimensional conformal field theories arising from lattices

Maria Stella Adamo*1, Luca Giorgetti^{†2}, and Yoh Tanimoto^{‡2}

¹Department Mathematik, Friedrich-Alexander-Universität Erlangen-Nürnberg Cauerstrasse 11, 91058 Erlangen, Germany ²Dipartimento di Matematica, Università di Roma Tor Vergata, Via della Ricerca Scientifica 1, I-00133 Roma, Italy

Dedicated to the memory of Huzihiro Araki.

Abstract

For a (finite-dimensional) real Hilbert space \mathfrak{h} and an orthogonal projection p, we consider the associated Heisenberg Lie algebra and the two-dimensional Heisenberg conformal net. Given an even lattice Q in \mathfrak{h} with respect to the indefinite bilinear form on \mathfrak{h} defined by p, we construct a two-dimensional conformal net \mathcal{A}_Q extending the Heisenberg conformal net. Moreover, with a certain discreteness assumption on the spectrum of the extension, we show that any two-dimensional extension of the Heisenberg conformal net is of the form \mathcal{A}_Q up to unitary equivalence.

We consider explicit examples of even lattices where \mathfrak{h} is two-dimensional and p is one-dimensional, and we show that the extended net may have completely rational or non-completely rational chiral (i.e. one-dimensional lightray) components, depending on the choice of lattice. In the non-rational case, we exhibit the braided equivalence of a certain subcategory of the representation category of the chiral Heisenberg net corresponding to the two-dimensional lattice extension.

Inspired by the charge and braiding structures of these nets, we construct two-dimensional conformal Wightman fields on the same Hilbert spaces. We show that, in some cases, these Wightman fields generate the corresponding extended nets.

1 Introduction

Two-dimensional Conformal Field Theories (CFT) are nowadays studied in various frameworks. Some of them are specifically developed to describe and encompass conformal covariance, such as Vertex Operator Algebras (VOA) [FLM88, HK07], or the Segal axioms [Seg04]. Others have been introduced to deal with more general Quantum Field Theories and they can be restricted to the conformal setting: such as conformal nets [KL04a, KL04b] (fulfilling the Araki–Haag–Kastler axioms [Haa96, Ara99]), or the Gårding–Wightman axioms [LM75, SW00]. There is a

^{*}maria.stella.adamo@fau.de

[†]giorgett@mat.uniroma2.it

[‡]hoyt@mat.uniroma2.it

subclass of conformal field theories that is relatively well-understood, especially in two spacetime dimensions: they are called rational CFTs. These theories have well-behaved chiral components (described by rational VOAs or chiral conformal nets), and their representation categories have rigid structures (they are typically fusion, braided and modular). In the rational setting, the full two-dimensional CFTs extending the tensor product of chiral components can be described purely algebraically and categorically using certain commutative algebra objects in the tensor product of their representation categories [DMNO13], or equivalently, using module categories [Ost03], or braided equivalences of the chiral components [DNO13], see e.g. [FRS02, SFR06, RFFS07, Kon07, BKL15] and references therein. On the other hand, there is a huge (indeed uncountable) family of non-rational CFTs (e.g. whenever the central charge is positive and not a rational number), see e.g. [Car04], [Xu06], and their structure remains to be understood.

A representative family of non-rational (chiral or two-dimensional) CFTs is given by realizations of the canonical commutation relations: the so-called Heisenberg algebras. Recently, an algebraic-analytic framework for two-dimensional (full) CFT (called full VOA) has been proposed [Mor23] and the extensions of the Heisenberg algebras have been studied. In [AMT24], it has been proved that the correlation functions in a full VOA (assuming unitarity and some technical conditions) define Euclidean correlation functions satisfying the Osterwalder–Schrader axioms, and thus there are corresponding two-dimensional conformal Gårding–Wightman fields [AGT23], by the celebrated Osterwalder-Schrader reconstruction theorem. Therefore, it is natural to address a more direct construction of Gårding–Wightman fields and conformal nets. In this work, we construct (and classify) two-dimensional extensions of the Heisenberg models, both in the conformal net and Gårding–Wightman framework.

The Heisenberg algebra associated with a finite-dimensional real Hilbert space \mathfrak{h} is the Lie algebra, denoted by \mathfrak{h} , generated by $\{\boldsymbol{\alpha}(m)\}_{\boldsymbol{\alpha}\in\mathfrak{h},m\in\mathbb{Z}}$ and a central element K with commutation relations $[\boldsymbol{\alpha}(m),\boldsymbol{\beta}(n)] = \delta_{m,-n}m(\boldsymbol{\alpha},\boldsymbol{\beta})_{\mathfrak{h}}K$. By choosing an orthogonal projection p, we can declare that $\hat{p}\hat{h}$ corresponds to the (left) chiral component, and $\hat{p}\hat{h}$ to the (right) chiral (i.e. antichiral) component, where $\bar{p} = 1 - p$ is the projection onto the orthogonal complement. The algebra ph has irreducible modules parametrized by $\lambda \in ph$, thus they are naturally endowed with an additive group structure. In analogy with the rational case, in order to construct a two-dimensional CFT as an extension of the two-dimensional tensor product of ph with $\bar{p}h$, one might consider the trivial braided equivalence from the category of all $p\mathfrak{h}$ -modules to that of $\bar{p}\mathfrak{h}$ -modules (if $\dim p = \dim \bar{p}$ and ph and $\bar{p}h$ are identified). It turns out that this is not the only option. In addition to these "diagonal" extensions, one can choose a (not necessarily N-dimensional) even lattice Q in the N-dimensional real space \mathfrak{h} with respect to the indefinite bilinear form defined by p, $(\cdot|\cdot) := (p\cdot|p\cdot)_{\mathfrak{h}} - (\bar{p}\cdot|\bar{p}\cdot)_{\mathfrak{h}}$, where only a subset of simple modules appear, and then construct "nondiagonal" CFT extensions. Given an arbitrary even lattice Q in \mathfrak{h} , we perform this construction first in the two-dimensional conformal net setting, and then in the Gårding-Wightman setting. We do this by considering the 2-cocycle ϵ of the lattice Q [Kac98] and the twisted shift operators. The interplay between ϵ and the braiding of the chiral representation category guarantees the locality of the extended CFT.

By looking at special cases, we observe that these families of models accommodate both rational and non-rational cases. For example, when \mathfrak{h} is two-dimensional ($\mathfrak{h} \cong \mathbb{R}^2$) and we divide it into one-dimensional ($\cong \mathbb{R}$) orthogonal subspaces corresponding the chiral and antichiral parts, there are even lattices $Q \subset \mathfrak{h}$ generated by $\frac{R}{\sqrt{2}} \oplus \frac{R}{\sqrt{2}}$ and $\frac{R^{-1}}{\sqrt{2}} \oplus -\frac{R^{-1}}{\sqrt{2}}$ for any $R \in \mathbb{R}, R \neq 0$, and the corresponding two-dimensional conformal net, denoted by \mathcal{A}_Q , has rational chiral components properly extending the chiral Heisenberg nets if and only if $R^2 \in \mathbb{Q}$. If $R^2 \in \mathbb{R} \setminus \mathbb{Q}$, the chiral components of \mathcal{A}_Q remain the chiral Heisenberg nets $\mathcal{A}_{\mathbb{R}}$, which are not rational.

This non-rational case is of particular interest. Differently from the rational case [DNO13, BKL15], we are not adding bulk (i.e. non-chiral) fields associated with all the irreducible objects in the representation categories of the chiral components. Yet, the resulting two-dimensional conformal net \mathcal{A}_Q appears to be maximally extended (cf. our classification result in Section 5 which holds under a certain technical condition, see below). This suggests that, even considering proper subcategories of the chiral components, one may obtain maximally extended CFTs. In the examples above with $R^2 \in \mathbb{R} \setminus \mathbb{Q}$, we exhibit the braided equivalence corresponding to the (infinite index) extension $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}} \subset \mathcal{A}_Q$. The equivalence is defined on the (strict) tensor subcategory generated by the irreducible representations with charge $\frac{R}{\sqrt{2}}$ and $\frac{R^{-1}}{\sqrt{2}}$, it is a non-strictly tensor functor, and its non-trivial tensorator is the 2-cocycle ϵ . The analysis of these concrete examples opens the way to the study of more general two-dimensional non-rational conformal nets and extensions.

For an arbitrary N-dimensional \mathfrak{h} , under the condition that the additive subgroup of charges appearing in the extension is discrete, we classify all possible two-dimensional conformal net extensions of the Heisenberg nets: they are all associated with an even lattice. We follow the strategy of [BMT88]. That is, we first extract charge-carrying operators and we put them in a canonical form, then we examine the possible 2-cocycles arising. By the discreteness assumption, the set of possible charges forms a lattice. Locality of the extension imposes that the lattice is even and that the 2-cocycles are symmetric. As the symmetric 2-cocycles of an abelian group are necessarily 2-coboundaries, we obtain the classification result.

The models constructed here are relatively simple, yet of interest from various points of view. For example, in the physics literature, models corresponding to different Q (in the same \mathfrak{h}) are believed to be related by the "current-current" deformation: assuming that they are associated with a Lagrangian, different models should be associated with the Lagrangian perturbed by the current. On the other hand, when \mathfrak{h} is 2-dimensional and p is 1-dimensional as in the examples mentioned earlier, the models constructed here seem to be related with the massless free field with the compactified target spaces. See [Mor23, Section 6] and references therein for the physics literature. It is an interesting problem to connect these models with the conformal field theories arising from certain Lagrangians, see [GKR24] for recent developments.

Once we have clear charge structures, we proceed to construct Wightman fields. The fields generating the two-dimensional extensions are labelled by the elements of the lattice, and the smeared operator creates states with that charge. In order to show that they generate a two-dimensional conformal net, we need to verify that they satisfy strong enough bounds, cf. [AGT23, Lemma 5.2]. Such bounds are proved in some cases.

Heisenberg algebras and their extensions are associated with the loop group of the groups \mathbb{R}^N or $(S^1)^N$, respectively (see [TL97, Chapter V]), and from their vacuum representations one can construct conformal nets. In this regard, it is important to consider their reflection positive representations (cf. [ANS22, ANS25] for finite-dimensional groups), which should give rise to their Euclidean counterparts. Such representations will be the subject of future investigations.

This paper is organized as follows. In Section 2.1, we recall the definitions of two-dimensional conformal nets, chiral conformal nets on S^1 , and their representations. In Section 2.2, we consider the basic case where $\mathfrak{h} \cong \mathbb{R}$, and we recall the definition of the associated Heisenberg algebra $\widehat{\mathfrak{h}}$ (the U(1)-current algebra) and the corresponding conformal net on S^1 . In Section 2.3, we generalize it to the finite-dimensional case and recall the (braided and tensor) structure of its modules (representations). In Section 3, we construct the two-dimensional conformal net \mathcal{A}_Q associated with an even lattice $Q \subset \mathfrak{h}$ with respect to the bilinear form $(\cdot|\cdot)$, twisted by p. In Section 4, we study the braided equivalence associated with the extension in a class of examples,

where $Q \subset \mathfrak{h} \cong \mathbb{R}^2$. In Section 5, we classify two-dimensional conformal nets extending the Heisenberg nets, under the assumption that the representations of the Heisenberg net appearing in the Hilbert space of the extension form a discrete subset of \mathfrak{h} . In Section 6, we construct the corresponding Wightman fields on the Hilbert space of \mathcal{A}_Q .

2 Preliminaries

2.1 Conformal nets

2.1.1 Two-dimensional conformal nets

We consider the Araki–Haag–Kastler framework [Haa96, Ara99] for two-dimensional conformal field theory [KL04b]. In general, an Araki-Haag-Kastler net is defined on the Minkowski space \mathbb{R}^{1+1} , but in the conformal case, one can take more convenient coordinates: as we did in [AGT23, Section 2.1], we map the two-dimensional Minkowski space to $(-\pi,\pi) \times (-\pi,\pi)$ by $(a_+,a_-) \mapsto (2 \arctan a_+, 2 \arctan a_-)$, where (a_+,a_-) are the lightcone coordinates on \mathbb{R}^{1+1} (cf. [KL04b]). With this identification, $\overline{\mathrm{Diff}}_+(S^1) \times \overline{\mathrm{Diff}}_+(S^1)$ and $\overline{\mathrm{M\"ob}} \times \overline{\mathrm{M\"ob}}$ act locally on \mathbb{R}^{1+1} , where $\overline{\mathrm{M\"ob}} = \overline{\mathrm{PSL}_2(\mathbb{R})}$. Recall that $\overline{\mathrm{M\"ob}} \times \overline{\mathrm{M\"ob}}$, we may assume that it is a true (non-projective) representation.

A conformal net on \mathbb{R}^{1+1} , or two-dimensional conformal net, on a Hilbert space \mathcal{H} is a triple (\mathcal{A}, U, Ω) where \mathcal{A} is an assignment to each open bounded region $O \subset \mathbb{R}^{1+1}$ of a von Neumann algebra $\mathcal{A}(O)$ on \mathcal{H} , U is a unitary projective representation of $\overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ on \mathcal{H} , $\Omega \in \mathcal{H}$ such that (we assume a stronger condition that assures that the net extends to the Einstein cylinder [AGT23, Section 2.3, Proposition 2.5], so this is the definition of [KL04b])

- (2dCN1) **Isotony:** if $O_1 \subset O_2$, then $\mathcal{A}(O_1) \subset \mathcal{A}(O_2)$.
- (2dCN2) Locality: if O_1 and O_2 are spacelike separated, then $\mathcal{A}(O_1) \subset \mathcal{A}(O_2)'$.
- (2dCN3) **Diffeomorphism covariance:** for a bounded region $O \subset \mathbb{R}^{1+1}$, there is a neighborhood \mathcal{U} of the unit element of $\overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ such that if $\gamma \in \mathcal{U}$ then $\gamma O \subset \mathbb{R}^{1+1}$ and

$$U(\gamma)\mathcal{A}(O)U(\gamma)^* = \mathcal{A}(\gamma O).$$

Furthermore, if supp γ is spacelike from O, then $\operatorname{Ad} U(\gamma)(x) = x$ for all $x \in \mathcal{A}(O)$.

- (2dCN4) **Positivity of energy:** the restriction of U to the translation subgroup $\mathbb{R}^2 \subset \overline{\text{M\"ob}} \times \overline{\text{M\"ob}}$ has the joint spectrum contained in the closed forward light cone $\overline{V_+} = \{(a_0, a_1) \in \mathbb{R}^{1+1} : a_0^2 a_1^2 \ge 0, a_0 \ge 0\}$ (in the original coordinates) $= \{(a_+, a_-) \in \mathbb{R}^{1+1} : a_{\pm} \ge 0\}$ (in the lightcone coordinates).
- (2dCN5) Vacuum and the Reeh-Schlieder property: there exists a unique (up to a phase) vector $\Omega \in \mathcal{H}$ such that $U(g)\Omega = \Omega$ for $g \in \overline{\text{M\"ob}} \times \overline{\text{M\"ob}}$ and is cyclic for any local algebra, namely $\overline{\mathcal{A}(O)\Omega} = \mathcal{H}$.
- (2dCN6) Extension to the cylinder: $U(R_{2\pi} \times R_{-2\pi}) = 1$, where $R_{\bar{t}}$ is the lift of rotation by t in $\overline{\text{M\"ob}}$.

One of the main results of this paper is to construct a family of examples of two-dimensional conformal nets on \mathbb{R}^{1+1} . For this purpose, we note that it is enough to construct $\mathcal{A}(O)$ first for **double cones**, which correspond to a set of the form $I_+ \times I_- \subset (-\pi, \pi) \times (-\pi, \pi)$, where I_+, I_- are relatively compact intervals in $(-\pi, \pi)$, For such \mathcal{A} , we require the same axioms except that locality means that $\mathcal{A}(I_{+,1} \times I_{-,1})$ commutes with $\mathcal{A}(I_{+,2} \times I_{-,2})$ whenever one of the following conditions holds:

- $I_{+,1} < I_{+,2}$ and $I_{-,2} < I_{-,1}$
- $I_{+,2} < I_{+,1}$ and $I_{-,1} < I_{-,2}$

Note that any of these conditions correspond to spacelike-separated double cones in \mathbb{R}^{1+1} , and $\overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ map (locally) any double cone to a double cone.

From such \mathcal{A} for double cones, for any other open set $O \subset (-\pi, \pi) \times (-\pi, \pi)$ with non-trivial spacelike complement, we can extend \mathcal{A} by $\mathcal{A}(O) = \bigvee_{I_+ \times I_- \subset O} \mathcal{A}(I_+ \times I_-)$, where $\bigvee_{\bullet} \mathcal{M}_{\bullet}$ denotes the von Neumann algebra generated by $\{\mathcal{M}_{\bullet}\}$. With same U, Ω , the triple (\mathcal{A}, U, Ω) with the extended \mathcal{A} satisfies the axioms of two-dimensional conformal net on \mathbb{R}^{1+1} .

When (\mathcal{A}, U, Ω) is a conformal net on \mathbb{R}^{1+1} , the net \mathcal{A} contains nontrivial chiral components [Reh00]: they are conformal nets $\mathcal{A}_+, \mathcal{A}_-$ on S^1 (see Section 2.1.2) such that \mathcal{A} is an extension of $\mathcal{A}_+ \otimes \mathcal{A}_-$ (see Section 2.1.3). In this work, we do not assume that they are the maximal ones given in [Reh00]. Instead, we start with concrete examples of $\mathcal{A}_+, \mathcal{A}_-$ (the Heisenberg nets, Section 2.3.2) and construct (and classify) the extensions \mathcal{A} .

2.1.2 Conformal nets on S^1

Let \mathcal{I} be the set of non-empty, open, non-dense, proper intervals of S^1 . For $I \in \mathcal{I}$, we denote by I' the interior of $S^1 \setminus I$. A **conformal net on** S^1 on a Hilbert space \mathcal{H}_{κ} is a triple $(\mathcal{A}_{\kappa}, U_{\kappa}, \Omega_{\kappa})$, where \mathcal{A}_{κ} (the subscript κ is to remind the reader that the net is chiral, i.e. defined on S^1) is an assignment to each $I \in \mathcal{I}$ of a von Neumann algebra $\mathcal{A}_{\kappa}(I)$ on \mathcal{H}_{κ} , U_{κ} is a unitary projective representation of $\mathrm{Diff}_+(S^1)$ on \mathcal{H}_{κ} , $\Omega_{\kappa} \in \mathcal{H}_{\kappa}$ such that

- (1dCN1) **Isotony:** if $I_1 \subset I_2$, then $\mathcal{A}_{\kappa}(I_1) \subset \mathcal{A}_{\kappa}(I_2)$.
- (1dCN2) **Locality:** if I_1 and I_2 are disjoint, then $\mathcal{A}_{\kappa}(I_1) \subset \mathcal{A}_{\kappa}(I_2)'$.
- (1dCN3) **Diffeomorphism covariance:** For $I \in \mathcal{I}, \gamma \in \text{Diff}_+(S^1)$,

$$U_{\kappa}(\gamma)\mathcal{A}_{\kappa}(I)U_{\kappa}(\gamma)^* = \mathcal{A}_{\kappa}(\gamma I).$$

Furthermore, if supp γ is disjoint from I, then $\operatorname{Ad} U_{\kappa}(\gamma)(x) = x$ for $x \in \mathcal{A}_{\kappa}(I)$.

- (1dCN4) **Positivity of energy:** the restriction of U_{κ} to the rotation subgroup has the spectrum contained in \mathbb{N} .
- (1dCN5) Vacuum and the Reeh-Schlieder property: there exists a unique (up to a phase) vector $\Omega_{\kappa} \in \mathcal{H}_{\kappa}$ such that $U_{\kappa}(g)\Omega_{\kappa} = \Omega_{\kappa}$ for $g \in \text{M\"ob}$ and $\overline{\mathcal{A}_{\kappa}(I)\Omega_{\kappa}} = \mathcal{H}_{\kappa}$.

As we assume (1dCN4), U_{κ} is a projective representation of Diff₊(S¹) rather than of $\overline{\text{Diff}_{+}(S^1)}$, see [AGT23, Proposition 2.1].

For a conformal net \mathcal{A}_{κ} on S^1 , the following properties are automatic:

- $\mathcal{A}_{\kappa}(I)$ are type III₁ factors [GF93, Lemma 2.9]
- Haag duality $\mathcal{A}_{\kappa}(I)' = \mathcal{A}_{\kappa}(I')$ [GF93, Theorem 2.19]
- Additivity $\mathcal{A}_{\kappa}(I) = \overline{\bigcup_{I_{\alpha} \in I} \mathcal{A}_{\kappa}(I_{\alpha})}$ [FJ96, Section 3]

Let us recall some useful notions regarding representations of conformal nets (see [AGT23, Section 2.2.2]). A (locally normal) representation of a conformal net $(\mathcal{A}_{\kappa}, U_{\kappa}, \Omega_{\kappa})$ on S^1 is a family of (normal) representations $\rho = \{\rho_I\}$ of $\{\mathcal{A}_{\kappa}(I)\}_{I \in \mathcal{I}}$ on \mathcal{H}_{ρ} such that if $I_1 \subset I_2$, then $\rho_{I_2}|_{\mathcal{A}_{\kappa}(I_1)} = \rho_{I_1}$ (compatibility). For any such irreducible representation, there is a multiplier representation U_{ρ} of $\overline{\mathrm{Diff}_{+}(S^1)}$, defined by $U_{\rho}(\gamma) = \rho(U_{\kappa}(\gamma))$ if $\mathrm{supp}\,\gamma \subset I_1$ for some $I_1 \in \mathcal{I}$, such that for any $I \in \mathcal{I}$ and $x \in \mathcal{A}_{\kappa}(I)$ it holds that $\rho_{\gamma I}(\mathrm{Ad}\,U(\gamma)(x)) = \mathrm{Ad}\,U_{\rho}(\gamma)(\rho_I(x))$, by [AGT25, Section 5.2]¹. In particular, if $\mathcal{H}_{\rho} = \mathcal{H}_{\kappa}$ (the vacuum Hilbert space), and $\rho_I(\mathcal{A}_{\kappa}(I)) = \mathcal{A}_{\kappa}(I)$ for each I, we call ρ an **automorphism**. Note that an automorphism is irreducible. A representation on the same Hilbert space $\mathcal{H}_{\rho} = \mathcal{H}_{\kappa}$ is said to be **localized in** I if $\rho_{I'} = \mathrm{id}$. where I' is the interior of $S^1 \setminus I$. In this case, ρ_I maps $\mathcal{A}_{\kappa}(I)$ to itself by Haag duality. Given ρ_1 localized in I_1 , one can always find a unitarily equivalent automorphism ρ_2 localized in another interval I_2 , as local algebras are of type III. If $I_1 \cup I_2 \subset I \in \mathcal{I}$, there is a unitary operator $V \in \mathcal{A}_{\kappa}(I)$ implementing the equivalence between ρ_1 and ρ_2 by Haag duality. Such a V is called a **charge transporter**.

For an automorphism ρ and $\gamma \in \overline{\mathrm{Diff}_+(S^1)}$, $\rho^{\gamma} := \mathrm{Ad}\,U_{\kappa}(\gamma) \circ \rho \circ \mathrm{Ad}\,U_{\kappa}(\gamma^{-1})$ is localized in γI and is equivalent to ρ . We can take $z_{\kappa,\rho}(\gamma) := U_{\kappa}(\gamma)U_{\kappa}^{\rho}(\gamma)^*$ as a charge transporter such that $\mathrm{Ad}\,z_{\kappa,\rho} \circ \rho = \rho^{\gamma}$. We call $z_{\kappa,\rho}(\gamma)$ the **covariance cocycle** of ρ .

Let us remove -1 from S^1 and identify $S^1 \setminus \{-1\}$ with $(-\pi, \pi)$. Let ρ_1, ρ_2 be two automorphisms localized in I such that $\overline{I} \subset (-\pi, \pi)$. Take $\tilde{\rho}_1, \tilde{\rho}_2$ localized in $I_1, I_2 \in \mathcal{I}$, respectively, such that $I_1 \cap I_2 = \emptyset$, $I_1, I_2 \subset (-\pi, \pi)$ and $\tilde{\rho}_j$ is equivalent to $\rho_j, j = 1, 2$. For charge transporters V_1, V_2 between ρ_1 and $\tilde{\rho}_1$, ρ_2 and $\tilde{\rho}_2$, respectively, $\varepsilon_{\rho_1,\rho_2}^{\pm} := \rho_2(V_1^*)V_2^*V_1\rho_1(V_2)$, where \pm depends on whether I_1 is on the future (right, +) or the past (left, -) of I_2 , is called the (**DHR**) braiding between ρ_1, ρ_2 , after [DHR71, FRS89, GF93]. The braiding does not depend on the choice of $\tilde{\rho}_j$ or the charge transporters under these conditions.

If we take γ such that γI is on the future or the past of I, $I \cap \gamma I = \emptyset$ and $V_1 = \mathbb{1}$, $V_2 = z_{\kappa,\rho_2}(\gamma)$, then the braiding is (+ corresponds to γI being in the past of I, – corresponds to the future)

$$\varepsilon_{\rho_1,\rho_2}^{\pm} = z_{\kappa,\rho_2}(\gamma)^* \rho_1(z_{\kappa,\rho_2}(\gamma)). \tag{2.1}$$

Let us consider a pair of conformal nets $(\mathcal{A}_j, U_j, \Omega_j)$ on S^1 on the Hilbert spaces \mathcal{H}_j , j=1,2. We can consider its tensor product $\mathcal{A}_1 \otimes \mathcal{A}_2(I) := \mathcal{A}_1(I) \otimes \mathcal{A}_2(I)$ and it is again a conformal net² on S^1 with respect to $U_1 \otimes U_2$ and $\Omega_1 \otimes \Omega_2$ on $\mathcal{H}_1 \otimes \mathcal{H}_2$. For two representations σ_j of \mathcal{A}_j , respectively j=1,2, the tensor product $\sigma_1 \otimes \sigma_2$ is a representation of $\mathcal{A}_1 \otimes \mathcal{A}_2$. Let $\rho_j, \sigma_j, j=1,2$ be localized representations of \mathcal{A}_j , respectively. It is easy to show, if $V_j, W_j, j=1,2$ are charge transporters of ρ_j, σ_j as before, then $V_j \otimes W_j$ are a charge transporter for $\rho_j \otimes \sigma_j$, thus $\varepsilon_{\rho_1, \rho_2, \sigma_1 \otimes \sigma_2}^{\pm} = \varepsilon_{\rho_1, \sigma_1}^{\pm} \otimes \varepsilon_{\rho_2, \sigma_2}^{\pm}$.

¹It is important that, while it is known that any representation of a conformal net is associated with a positive-energy projective representation (defined up to a phase) of $\overline{\text{Diff}_{+}(S^1)}$ [DFK04, Wei06, Gui21], we have a global multiplier representation (defined including the phase) of $\overline{\text{Diff}_{+}(S^1)}$ with the same cocycle, as shown in [AGT25], because we will take the direct sum of multiplier representations with the same cocycle in (3.7) (which is again a multiplier representation with the same cocycle), while direct sum of projective representations (when appropriately defined) is not assured to be a projective representation.

²This should be distinguished from the two-dimensional conformal net with A_1 as the chiral component, A_2 as the antichiral component, see Section 2.1.3.

2.1.3 Two-dimensional tensor product nets and extensions

Let $(\mathcal{A}_{\pm}, U_{\pm}, \Omega_{\pm})$ be two conformal nets on S^1 on the Hilbert space \mathcal{H}_{\pm} , respectively. From them, we can construct a two-dimensional conformal net $(\mathcal{A}_{+} \otimes \mathcal{A}_{-}, U_{+} \otimes U_{-}, \Omega_{+} \otimes \Omega_{-})$ on the Hilbert space $\mathcal{H}_{+} \otimes \mathcal{H}_{-}$ by setting (first for double cones, then extending to arbitrary bounded regions as explained in Section 2.1.1)

- $\mathcal{A}_+ \otimes \mathcal{A}_-(I_+ \times I_-) := \mathcal{A}_+(I_+) \otimes \mathcal{A}_-(I_-)$
- $U_+ \otimes U_-(\gamma_+ \times \gamma_-) := U_+(\gamma_+) \otimes U_-(\gamma_-)$
- $\Omega_+ \otimes \Omega_- \in \mathcal{H}_+ \otimes \mathcal{H}_-$

The \pm -components are referred to as the **chiral** and **antichiral** components, respectively. Such a net *a priori* extends to the Einstein cylinder (see [AGT23]), but by (1dCN4), it reduces to the torus.

One can also consider representations of such nets, requiring the compatibility condition as we did in Section 2.1.2. We say that $\tau = \{\tau_O\}$, where τ_O is a representation of $\mathcal{A}_+ \otimes \mathcal{A}_-(O)$, where O is a region on the Einstein cylinder, if $\tau_{\tilde{O}}|_{\mathcal{A}(O)} = \tau_O$ for any $O \subset \tilde{O}$, where \tilde{O} is contained in a copy of the Minkowski space. Such τ can be restricted to $\mathcal{A}_+ \otimes \mathbb{1}$, $\mathbb{1} \otimes \mathcal{A}_-$ and gives representations of $\mathcal{A}_+, \mathcal{A}_-$ as conformal nets on S^1 , by considering the time-zero circle of the cylinder.

A generic two-dimensional conformal net (\mathcal{A}, U, Ω) is an extension of such net in the following sense: There are (two-dimensional) conformal nets of von Neumann algebras $\underline{\mathcal{A}}_+$, $\underline{\mathcal{A}}_-$ acting on the Hilbert space \mathcal{H} of \mathcal{A} , which satisfy

- all the axioms (2dCN1)-(2dCN4) except that Ω is not required to be cyclic for $\underline{\mathcal{A}}_{\pm}$,
- if $x \in \underline{\mathcal{A}}_+(I_+ \times I_-)$, then $\operatorname{Ad} U(\iota \times \gamma_-)(x) = x$ for all $\gamma_- \in \overline{\operatorname{Diff}_+(S^1)}$ and ι the identity diffeomorphism, and similarly, if $x \in \underline{\mathcal{A}}_-(I_+ \times I_-)$, then $\operatorname{Ad} U(\gamma_+ \times \iota)(x) = x$ for all $\gamma_+ \in \overline{\operatorname{Diff}_+(S^1)}$,
- $\underline{\mathcal{A}}_{\pm}(O) \subset \mathcal{A}(O)$ for all O.

In this case, we can define (by a slight abuse of notations) $\underline{\mathcal{A}}_{\pm}(I_{\pm}) := \underline{\mathcal{A}}_{\pm}(I_{+} \times I_{-})$, where $\mathcal{A}_{\pm}(I_{\pm})$ does not depend on I_{\mp} , and $\mathcal{H}_{\pm} := \overline{\bigcup_{I \in \mathcal{I}} \mathcal{A}_{\pm}(I)\Omega}$ (which is equal to $\overline{\mathcal{A}}_{\pm}(I)\Omega$ for any $I \in \mathcal{I}$ by the Reeh-Schlieder argument, see [BLM11, Lemma 5.1] and [GLW98, Theorem 1.4]) and define the restrictions $\mathcal{A}_{\pm}(I) := \underline{\mathcal{A}}_{\pm}(I)|_{\mathcal{H}_{\pm}}$. They are conformal nets on S^{1} , as we assume (2dCN6). In this case, we say that \mathcal{A} is an **extension** $\mathcal{A}_{+} \otimes \mathcal{A}_{-}$ and we denote it as $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{A}$. We say that $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{A}$ is a **conformal extension** if $U(\gamma_{+} \times \iota) \in \underline{\mathcal{A}}_{+}(I_{+})$ if supp $\gamma_{+} \subset I_{+}$ and $U(\iota \times \gamma_{-}) \in \underline{\mathcal{A}}_{-}(I_{-})$ if supp $\gamma_{-} \subset I_{-}$. In this case, it follows that the inclusion $\underline{\mathcal{A}}_{+}(I_{+}) \otimes \underline{\mathcal{A}}_{-}(I_{-}) \subset \mathcal{A}(I_{+} \times I_{-})$ is irreducible for each pair of I_{+}, I_{-} .

If \mathcal{A} is an extension of $\mathcal{A}_{+}\otimes\mathcal{A}_{-}$, then the map $\underline{\mathcal{A}}_{\pm}(I)\ni x\mapsto x|_{\mathcal{H}_{\pm}}\in\mathcal{A}_{\pm}(I)$ is an injective homomorphism (as Ω is separating). The inverse map is a representation of $\mathcal{A}_{\pm}(I)$. Using this inverse map, we can consider the embedding $\mathcal{A}_{+}\otimes\mathcal{A}_{-}\to\mathcal{A}$, and it can be seen as a representation of $\mathcal{A}_{+}\otimes\mathcal{A}_{-}$ on \mathcal{H} .

2.2 The U(1)-current algebra

2.2.1 The U(1)-current algebra and its modules

We will consider the infinite-dimensional Lie algebras known as the Heisenberg algebras. As a special case with rank 1, the Lie algebra is also known as the U(1)-current algebra, especially in

the context of algebraic QFT [BMT88]. For the convenience of the reader familiar with the latter, we will briefly summarize the structures. See also [AGT23, Section 6].

The U(1)-current algebra is the infinite-dimensional Lie algebra generated by the symbols $\{J_m\}_{m\in\mathbb{Z}}$ and K, with the following commutation relations

$$[J_m, J_n] = \delta_{m,-n} mK$$

and K is a central element.

For each pair of values $c, \lambda \in \mathbb{C}$, we can consider the so-called Verma module of the U(1)-current algebra: it is spanned by a distinguished vector Ω_{λ} and vectors of the form

$$J_{-m_1}\cdots J_{-m_k}\Omega_{\lambda}, \qquad m_j \in \mathbb{Z}, m_j > 0$$

with the properties that $K = c\mathbb{1}, J_0 = \lambda\mathbb{1}$ and

$$J_m \Omega_{\lambda} = 0, \qquad m > 0.$$

This representation (module) of the U(1)-current algebra is denoted by $M_{\mathfrak{h}}(c,\lambda)$ (where \mathfrak{h} is a 1-dimensional \mathbb{R} -vector space, cf. Section 2.3.1). This can be seen as the algebraic symmetric Fock space based on $\{J_m\Omega_{\lambda}, m < 0\}$. On $M_{\mathfrak{h}}(c,\lambda)$ with $c \neq 0$, $\{c^{-\frac{1}{2}}J_m\}_{m\in\mathbb{Z}}$ satisfy the same commutation relations as the $\{J_m\}_{m\in\mathbb{Z}}$ with c=1, so we may and do assume that c=1.

If $\lambda \in \mathbb{R}$, the representation is **unitary**, in the sense that there is a (positive definite) invariant scalar product $\langle \cdot, \cdot \rangle$ with respect to which $(J_m)^* = J_{-m}$ and $\langle \Omega_{\lambda}, \Omega_{\lambda} \rangle = 1$. Note that, by the invariance property, the value of $\langle \cdot, \cdot \rangle$ is uniquely determined by λ and c = 1. Moreover, $M_{\mathfrak{h}}(1,\lambda)$'s are isomorphic to each other as vector spaces with a scalar product, by identifying the vectors $J_{-m_1} \cdots J_{-m_k} \Omega_{\lambda}$ (the values λ do not appear here, nor in the scalar product). The module $M_{\mathfrak{h}}(1,0)$ with $\lambda = 0$ is called the **vacuum module** and its completion accommodates the chiral conformal net.

2.2.2 The U(1)-current net and its representations

Let us take the vacuum module $M_{\mathfrak{h}}(1,0)$ of the U(1)-current algebra and its Hilbert space completion $\mathcal{H}_{\mathfrak{h},0}$ with respect to the invariant scalar product $\langle \cdot, \cdot \rangle$. We introduce the operators L_m on $M_{\mathfrak{h}}(1,0)$ (or on $M_{\mathfrak{h}}(1,\lambda)$) by the Sugawara formula

$$L_m := \frac{1}{2} \sum_{k \in \mathbb{Z}} : J_{m-k} J_k :, \quad m \in \mathbb{Z}.$$

$$(2.2)$$

They satisfy the commutation relations $[L_m, J_n] = -nJ_{m+n}$, and the Virasoro algebra relations $[L_m, L_n] = (m-n)L_{m+n} + \frac{c}{12}m(m^2-1)\delta_{m,-n}$ with central charge c=1. For $m_j > 0$

$$L_0 \cdot J_{-m_1} \cdots J_{-m_k} \Omega_0 = \left(\sum_{j=1}^k m_j \right) J_{-m_1} \cdots J_{-m_k} \Omega_0.$$

The operator L_0 is already diagonalized, essentially self-adjoint and positive on $M_{\mathfrak{h}}(1,0) \subset \mathcal{H}_{\mathfrak{h},0}$. We denote its closure by the same symbol L_0 , and refer to it as the conformal Hamiltonian. Let $C^{\infty}(L_0) := \bigcap_{j=1}^{\infty} \text{Dom}(L_0^j)$. It holds that $||J_m\Psi|| \leq (m+1)||(L_0+1)\Psi||$ for any $\Psi \in C^{\infty}(L_0)$ [BSM90, (2.23)], which we call the **linear energy bounds** for the U(1)-current $\{J_m\}_{m\in\mathbb{Z}}$ in the vacuum Hilbert space $\mathcal{H}_{\mathfrak{h},0}$.

Let $f \in C^{\infty}(S^1)$ (without further remark, $C^{\infty}(S^1)$ is real valued). We regard f as a 2π -periodic smooth function on \mathbb{R} . Let us write its Fourier components by $f_n = \frac{1}{2\pi} \int_{-\pi}^{\pi} e^{-int} f(t) dt$. On the domain $C^{\infty}(L_0)$, we define

$$J(f)\Psi := \sum_{m \in \mathbb{Z}} f_m J_m \Psi,$$

which is convergent in $\mathcal{H}_{\mathfrak{h},0}$ by the linear energy bounds, and $J(f)\Psi \in C^{\infty}(L_0)$. It defines an essentially self-adjoint operator whose closure we denote again by J(f), and $C^{\infty}(L_0)$ is an invariant core for J(f) for every $f \in C^{\infty}(S^1)$, cf. [AGT23, Appendix A]. We denote also $W(f) := e^{iJ(f)}$.

For $I \in \mathcal{I}$ (see Section 2.1.1), we define

$$\mathcal{A}_{\mathfrak{h}}(I) := \{W(f) : \operatorname{supp} f \subset I\}''.$$

The representation of the Virasoro algebra $\{L_m\}_{m\in\mathbb{Z}}$ integrates to a positive-energy projective unitary representation U_0 of $\mathrm{Diff}_+(S^1)$ such that

$$\operatorname{Ad} U_0(\gamma)(J(f)) = J(f \circ \gamma^{-1}), \quad \operatorname{Ad} U_0(\gamma)(W(f)) = W(f \circ \gamma^{-1}),$$

where the first equality is as unbounded operators (including the domains) [CKLW18, Proposition 6.4]. The triple $(\mathcal{A}_{\mathfrak{h}}, U_0, \Omega_0)$ satisfies the axioms of conformal net on S^1 , and we call it the U(1)-current net.

Let $h \in C^{\infty}(S^1) = C^{\infty}(S^1, \mathbb{R})$. For $I_1 \in \mathcal{I}$, we define (note that we changed convention from [AGT23], and the integral is normalized by $\frac{1}{2\pi}$)

$$\sigma_{h,I_1}(W(f)) = e^{\frac{i}{2\pi} \int f(t)h(t)dt}W(f).$$

Then this is an automorphism of $\mathcal{A}_{\mathfrak{h}}$. Any two such automorphisms $\sigma_{h_1,I_1},\sigma_{h_2,I_1}$ are unitarily equivalent if and only if $\frac{1}{2\pi} \int_{-\pi}^{\pi} h_1(t) dt = \frac{1}{2\pi} \int_{-\pi}^{\pi} h_2(t) dt$. We know that any such automorphism (representation) is diffeomorphism covariant, in the sense that there is a positive-energy unitary multiplier representation U_h of $\overline{\mathrm{Diff}_+(S^1)}$, see [AGT25, Section 5.2], such that

$$\operatorname{Ad} U_h(\gamma)(\sigma_h(W(f))) = \sigma_h(\operatorname{Ad} U_0(\gamma)(W(f))) = \sigma_h(W(f \circ \gamma^{-1})).$$

Let $I \in \mathcal{I}$. If we take $h \in C^{\infty}(S^1)$, supp $h \subset I$, then σ_h is an automorphism of $\mathcal{A}_{\mathfrak{h}}$ localized in I. Let us fix h such that $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$. For $\alpha \in \mathbb{R}$, we say that the automorphism $\sigma_{\alpha h}$ has charge α . If h is localized in I, then by the calculations of [AGT23]³, for $\alpha, \beta \in \mathbb{R}$, the braiding is

$$\varepsilon_{\alpha h.\beta h}^{\pm} = e^{\pm i\pi\alpha\beta}.$$

2.3 Heisenberg algebras

2.3.1 The Heisenberg Lie algebras and their modules

Let \mathfrak{h} be a finite-dimensional \mathbb{R} -vector space, $(\cdot, \cdot)_{\mathfrak{h}}$ be a *positive-definite* symmetric bilinear form on \mathfrak{h} . If \mathfrak{h} is 1-dimensional with the standard scalar product, the following construction gives the U(1)-current algebra of Section 2.2.1. See also [Kac98, Section 3.5].

³In [AGT23, Section 6.3], we took the convention $\int h(t)dt = 1$ and the automorphism was given by $W(f) \mapsto e^{i\int f(t)h(t)dt}W(f)$. In this paper we have $\frac{1}{2\pi}\int h(t)dt = 1$, thus h gets multiplied by 2π . On the other hand, in the calculations of braiding, we had integral such as $\int h_j(t)H_{j'}(t)dt$, where $h_j = \alpha_j h$ and $H_{j'}$ is a primitive of $h_{j'}$. We replace the integral by $\frac{1}{2\pi}\int h_j(t)H_{j'}(t)dt$, thus altogether the value has a factor of 2π compared with [AGT23]. In [AGT23] we had $\varepsilon_{\alpha h, \beta h}^{\pm} = e^{\pm \frac{\alpha \beta}{2}}$, thus it translates to the formula below.

We consider the Heisenberg Lie algebra $\widehat{\mathfrak{h}}$, which is spanned by elements of the form $\alpha(m)$, where $\alpha \in \mathfrak{h}, m \in \mathbb{Z}$ (with the relations $(\alpha + \beta)(m) = \alpha(m) + \beta(m)$ and $(\lambda \alpha)(m) = \lambda \alpha(m)$) and a central element K (that is, as a linear space, $\widehat{\mathfrak{h}} \cong (\bigoplus_{\mathbb{Z}} \mathfrak{h}) \oplus \mathbb{C}K$ and $\alpha(m)$ has the element $\alpha \in \mathfrak{h}$ in the m-th summand), with the commutation relations

$$[\boldsymbol{\alpha}(m), \boldsymbol{\beta}(n)] = \delta_{m,-n} m \cdot (\boldsymbol{\alpha}, \boldsymbol{\beta})_{\mathfrak{h}} K.$$

This is related with the U(1)-current algebra as follows.

- If \mathfrak{h} is 1-dimensional, then by taking a vector $\mathbf{v} \in \mathfrak{h}$ with $(\mathbf{v}, \mathbf{v})_{\mathfrak{h}} = 1$, we obtain $[\mathbf{v}(m), \mathbf{v}(n)] = \delta_{m,-n}mK$, which is exactly the commutation relation of the U(1)-current algebra, by identifying $\mathbf{v}(m) = J_m$.
- If \mathfrak{h} is N-dimensional, we can take an orthonormal basis. For orthogonal vectors $\boldsymbol{\alpha}, \boldsymbol{\beta} \in \mathfrak{h}$, one has $[\boldsymbol{\alpha}(m), \boldsymbol{\beta}(n)] = 0$ also for m = -n. For each $\boldsymbol{\alpha} \in \mathfrak{h}$ with $(\boldsymbol{\alpha}, \boldsymbol{\alpha})_{\mathfrak{h}} = 1$, the elements $\{\boldsymbol{\alpha}(m)\}_{m \in \mathbb{Z}}, K$ span a subalgebra isomorphic to the one-dimensional case. Note that, the (commutative) subalgebra generated by $\boldsymbol{\alpha}(m), m < 0$ is isomorphic to the direct sum of N copies of the subalgebra generated by $J_m, m < 0$ of the U(1)-current algebra, that is, if $\boldsymbol{\alpha} = (\alpha_1, \dots, \alpha_n) \in \mathfrak{h}$, each $\alpha_j(m)$ can be seen as a copy of $\alpha_j J_m$.

As in the 1-dimensional case, there are irreducible modules of this algebra parametrized by $\lambda \in \mathfrak{h}$ as follows. For each $\lambda \in \mathfrak{h}$, we denote $M_{\mathfrak{h}}(1,\lambda) = U(\widehat{\mathfrak{h}})/J_{\lambda}$, where $U(\widehat{\mathfrak{h}})$ is the universal enveloping algebra of $\widehat{\mathfrak{h}}$, J_{λ} is the ideal generated by $\alpha(m)$ for m > 0, $\alpha(0) - (\lambda, \alpha)_{\mathfrak{h}}$, and by K - 1. More explicitly, let $\{\boldsymbol{v}_j\}_{j=1,\dots,N}$ be a basis of \mathfrak{h} , If $\lambda = \sum_j \lambda_j \boldsymbol{v}_j, \lambda_j \in \mathbb{R}$, then this module is isomorphic to the tensor product $M_{\mathbb{R}\boldsymbol{v}_1}(1,\lambda_1) \otimes \cdots \otimes M_{\mathbb{R}\boldsymbol{v}_N}(1,\lambda_N)$ of modules in Section 2.2.1, where $\mathbb{R}\boldsymbol{v}_j$ are the 1-dimensional vector spaces spanned by \boldsymbol{v}_j (see [DG13, Theorem 3.43]) and each $\boldsymbol{v}_j(m)$ acts on the j-th component as J_m . If $\lambda = \mathbf{0}$, the module $M_{\mathfrak{h}}(1,\mathbf{0})$ is called the **vacuum module** here as well.

2.3.2 The Heisenberg conformal net and its representations

Let \mathfrak{h} be N-dimensional with $(\cdot, \cdot)_{\mathfrak{h}}$ as in Section 2.3.1. The following construction gives the N-fold tensor product of the U(1)-current net of Section 2.2.2.

We consider the Heisenberg Lie algebra $\widehat{\mathfrak{h}}$ as above. Let $M_{\mathfrak{h}}(1, \lambda)$ be one of the representations above. We denote the Hilbert space completion with respect to the invariant scalar product $\langle \cdot, \cdot \rangle$ by \mathcal{H}_{λ} , or when we need to stress \mathfrak{h} , by $\mathcal{H}_{\mathfrak{h},\lambda}$.

Fix an orthonormal basis $\{v_j\}_{j=1,\dots,N}$ of \mathfrak{h} . On each module $M_{\mathfrak{h}}(1,\boldsymbol{\lambda})$ (we will be mainly interested in $\boldsymbol{\lambda}=\mathbf{0}$ for the following construction) we introduce the representation of the Virasoro algebra given by the N-dimensional analogue of the Sugawara formula

$$L_m := \frac{1}{2} \sum_{j=1}^{N} \sum_{k \in \mathbb{Z}} : \boldsymbol{v}_j(m-k) \boldsymbol{v}_j(k) :$$

Actually, it is easy to check that this does not depend on the choice of the orthonormal basis $\{v_j\}$. The operators $\{L_m\}$ satisfy the Virasoro relations with central charge c = N. Indeed, each $v_j(m)$ acts on the j-th tensor component as J_m . In this way, L_m can be seen as the N-fold tensor product $\bigoplus_{j=1}^N \mathbb{1} \otimes \cdots \otimes L_m^{\mathrm{U}(1)} \otimes \cdots \otimes \mathbb{1}$ of the representation of the Virasoro algebra for the U(1)-current algebra.

As in the 1-dimensional case, for each $f \in C^{\infty}(S^1)$ with $f_m = \frac{1}{2\pi} \int_{-\pi}^{\pi} e^{-imt} f(t) dt$ and $\alpha \in \mathfrak{h}$, we introduce the densely defined operator on \mathcal{H}_0 :

$$\boldsymbol{\alpha}(f) := \sum_{m \in \mathbb{Z}} f_m \boldsymbol{\alpha}(m)$$

(with a slight abuse of notation " $\boldsymbol{\alpha}(\cdot)$ ": $\boldsymbol{\alpha} \in \mathfrak{h}$, $\boldsymbol{\alpha}(m) \in \operatorname{End}(M_{\mathfrak{h}}(1,\mathbf{0}))$ for $m \in \mathbb{Z}$, but here $\boldsymbol{\alpha}(\cdot)$ will define a Wightman field on S^1). For $\boldsymbol{\alpha}, \boldsymbol{\beta} \in \mathfrak{h}$, $f, g \in C^{\infty}(S^1)$, it holds that

$$[\boldsymbol{\alpha}(f), \boldsymbol{\beta}(g)] = (\boldsymbol{\alpha}, \boldsymbol{\beta})_{\mathfrak{h}} \cdot \frac{1}{2\pi} \int_{-\pi}^{\pi} f'(t)g(t)dt.$$

In particular, $\alpha(f)$ and $\beta(g)$ commute when supp f and supp g are disjoint.

Let us consider the vacuum module $M_{\mathfrak{h}}(1,\mathbf{0})$. For each $\boldsymbol{\alpha} \in \mathfrak{h}$, the components $\{\boldsymbol{\alpha}(m)\}_{m \in \mathbb{Z}}$ satisfy the linear energy bounds with respect to L_0 . To see this, one can take an orthonormal basis of \mathfrak{h} whose first vector is a scalar multiple of $\boldsymbol{\alpha}$, then the $\boldsymbol{\alpha}(m)$ are bounded by the first component of L_0 , thus by the whole operator L_0 : $\|\boldsymbol{\alpha}(m)\Psi\| \leq \|\boldsymbol{\alpha}\| \cdot (m+1)\|(L_0+1)\Psi\|$ for every $\Psi \in C^{\infty}(L_0) = \bigcap_{i=1}^{\infty} \mathrm{Dom}(L_0^j)$.

As in the 1-dimensional case, we obtain a positive-energy projective unitary representation U_0 of Diff₊ (S^1) , which satisfies for each $\alpha \in \mathfrak{h}$

$$\operatorname{Ad} U_{\mathbf{0}}(\gamma)(\boldsymbol{\alpha}(f)) = \boldsymbol{\alpha}(f \circ \gamma^{-1}),$$

including the domain $C^{\infty}(L_0)$. Let us define

$$\mathcal{A}_{\mathfrak{h}}(I) := \{ e^{i\boldsymbol{\alpha}(f)} : \boldsymbol{\alpha} \in \mathfrak{h}, f \in C^{\infty}(S^1), \operatorname{supp} f \subset I \}.$$

Together with the vacuum vector $\Omega_{\mathbf{0}}$, the triple $(\mathcal{A}_{\mathfrak{h}}, U_{\mathbf{0}}, \Omega_{\mathbf{0}})$ constitutes a conformal net on S^1 . There is a natural unitary equivalence $\mathcal{A}_{\mathfrak{h}} \cong \mathcal{A}_{\mathbb{R}\boldsymbol{v}_1} \otimes \cdots \otimes \mathcal{A}_{\mathbb{R}\boldsymbol{v}_N}$, where each $\mathcal{A}_{\mathbb{R}\boldsymbol{v}_1}$ is a copy of the U(1)-current net from Section 2.2.2.

The sectors (equivalence classes of irreducible representations) of the net can be parametrized by \mathfrak{h} . Indeed, let us take a basis $\{\boldsymbol{v}_j\}$ of \mathfrak{h} . By [LT18, Proposition 3.11], which applies to the U(1)-current net by [LT18, Example 3.9], and by the classification in [BMT88, Section 2B, 3B] together with [CW] (cf. [Zel17]) any sector σ of $\mathcal{A}_{\mathfrak{h}} \cong \mathcal{A}_{\boldsymbol{v}_1\mathbb{R}} \otimes \cdots \otimes \mathcal{A}_{\boldsymbol{v}_N\mathbb{R}}$ is of the form $\sigma_{\lambda_1 h} \otimes \cdots \otimes \sigma_{\lambda_N h}$, where $h \in C^{\infty}(S^1)$ such that $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$, giving the parametrization $(\lambda_1, \dots, \lambda_N) \in \mathbb{R}^N \cong \mathfrak{h}$, whose Hilbert space we denote by $\mathcal{H}_{\boldsymbol{\lambda}h}$ (or $\mathcal{H}_{\mathfrak{h},\boldsymbol{\lambda}h}$ when necessary). These sectors correspond to the module $M_{\mathfrak{h}}(1,\boldsymbol{\lambda}), \boldsymbol{\lambda} = \sum_{j} \lambda_j \boldsymbol{v}_j$ (up to a unitary equivalence).

Let us fix $h \in C^{\infty}(\overline{S^1})$ such that $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$. We know that, for $\alpha, \beta \in \mathbb{R}$, the braiding of two sectors $\sigma_{\alpha h}$, $\sigma_{\beta h}$ of the U(1)-current net is $\varepsilon_{\alpha h,\beta h}^{\pm} = e^{\pm i\pi\alpha\beta}$. If \mathfrak{h} is N-dimensional, the braiding of the tensor product is the tensor product of the braidings, see Section 2.1.2. Let us write in the basis $\boldsymbol{\alpha} = \sum_{j} \alpha_{j} \boldsymbol{v}_{j}, \boldsymbol{\beta} = \sum_{j} \beta_{j} \boldsymbol{v}_{j} \in \mathfrak{h}$, and $\sigma_{\boldsymbol{\alpha} h} := \bigotimes_{j} \sigma_{\alpha_{j} h}, \sigma_{\boldsymbol{\beta} h} := \bigotimes_{j} \sigma_{\beta_{j} h}$. Then the braiding of the two sectors $\sigma_{\boldsymbol{\alpha} h}, \sigma_{\boldsymbol{\beta} h}$ of $\mathcal{A}_{\mathfrak{h}}$ is

$$\varepsilon_{\boldsymbol{\alpha}h,\boldsymbol{\beta}h}^{\pm} = \bigotimes_{j=1}^{N} \varepsilon_{\alpha_{j}\boldsymbol{v}_{j}h,\beta_{j}\boldsymbol{v}_{j}h}^{\pm} = e^{\pm i\pi\sum_{j}\alpha_{j}\beta_{j}} = e^{\pm i\pi(\boldsymbol{\alpha},\boldsymbol{\beta})_{\mathfrak{h}}}, \tag{2.3}$$

where we used that the braidings $\varepsilon_{\alpha_j \boldsymbol{v}_j h, \beta_j \boldsymbol{v}_j h}^{\pm}$ are scalar in this case. Furthermore, by the concrete construction, we have $\sigma_{\boldsymbol{\alpha}h}\sigma_{\boldsymbol{\beta}h} = \sigma_{(\boldsymbol{\alpha}+\boldsymbol{\beta})h}$. Although the Hilbert space of the representation $\sigma_{\boldsymbol{\alpha}h}$

remains the same $\mathcal{H}_{\mathbf{0}}$, when we stress that the net $\mathcal{A}_{\mathfrak{h}}$ is represented by $\sigma_{\boldsymbol{\alpha}h}$, we denote the Hilbert space by $\mathcal{H}_{\boldsymbol{\alpha}h}$.

Each representation $\sigma_{\alpha h}$ is diffeomorphism covariant [AGT25, Section 5.2]: there is a unitary multiplier representation $U_{\alpha h}$ of $\overline{\text{Diff}_+(S^1)}$ such that

$$\operatorname{Ad} U_{\boldsymbol{\alpha}h}(\gamma)(\sigma_{\boldsymbol{\alpha}h}(x)) = \sigma_{\boldsymbol{\alpha}h}(\operatorname{Ad} U_{\boldsymbol{0}}(\gamma)(x)), \qquad x \in \mathcal{A}_{\mathfrak{h}}(I_1), I_1 \in \mathcal{I}, \gamma \in \overline{\operatorname{Diff}_+(S^1)}. \tag{2.4}$$

By taking the constant function $\frac{\boldsymbol{\alpha}}{2\pi}$ (we put $\frac{1}{2\pi}$ to keep the normalization), we have $\sigma_{\boldsymbol{\alpha}h} \cong \sigma_{\frac{\boldsymbol{\alpha}}{2\pi}}$. While $\sigma_{\frac{\boldsymbol{\alpha}}{2\pi}}$ is not localized in any interval, it can be seen as the representation corresponding to the module $M_{\mathfrak{h}}(1,\boldsymbol{\lambda})$ without unitary equivalence.

3 Two-dimensional extensions

From now on, we fix an N-dimensional \mathbb{R} -vector space \mathfrak{h} together with a positive-definite symmetric bilinear form $(\cdot, \cdot)_{\mathfrak{h}}$ as in Section 2.3.1. In this Section, we construct a two-dimensional conformal net \mathcal{A}_Q that whose chiral components extend the Heisenberg nets, associated with a certain even lattice Q.

3.1 Chiral and antichiral components

Let p be an orthogonal projection on \mathfrak{h} with respect to $(\cdot, \cdot)_{\mathfrak{h}}$. We consider $p\mathfrak{h}$ to correspond to the chiral component, where $\bar{p} = 1 - p$ is the orthogonal complement of p. Let us denote the pair (\mathfrak{h}, p) as \mathfrak{h}_p (see Section 6.1 for the current algebra $\widehat{\mathfrak{h}}_p$ with the distinction between the chiral and antichiral parts). As each $p\mathfrak{h}, \bar{p}\mathfrak{h}$ are finite-dimensional \mathbb{R} -vector spaces, we have the associated Heisenberg conformal nets $\mathcal{A}_{p\mathfrak{h}}, \mathcal{A}_{\bar{p}\mathfrak{h}}$ on S^1 on $\mathcal{H}_{p\mathfrak{h},\mathbf{0}}, \mathcal{H}_{\bar{p}\mathfrak{h},\mathbf{0}}$, respectively. Then, on the Hilbert space $\mathcal{H}_{p\mathfrak{h},\mathbf{0}} \otimes \mathcal{H}_{\bar{p}\mathfrak{h},\mathbf{0}} \cong \mathcal{H}_{\mathfrak{h},\mathbf{0}}$, which we also denote by $\mathcal{H}_{\mathfrak{h}_p,\mathbf{0}}$, we have the two-dimensional conformal net $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ as in Section 2.1.3, which we shall also denote by $\mathcal{A}_{\mathfrak{h}_p}$.

We construct irreducible representations of the two-dimensional net $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ in the sense of Section 2.1.3: Let us fix $I \in \mathcal{I}$ and $h \in C^{\infty}(S^1)$ with $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$ and supp $h \subset I$. Then there is a one-to-one correspondence between equivalence classes of irreducible representations of $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ and $\lambda \in \mathfrak{h}$ (see Lemma 5.2 for classification), where the representation of the algebra $\mathcal{A}_{p\mathfrak{h}}(I_+) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I_-)$ is $\sigma_{p\lambda h,I_+} \otimes \sigma_{\bar{p}\lambda h,I_-}$. Again, the Hilbert spaces of these representations remain the same $\mathcal{H}_{\mathbf{0}}(=\mathcal{H}_{\mathfrak{h}_p,\mathbf{0}})$, but in order to indicate the representation, we denote it by $\mathcal{H}_{\mathfrak{h}_p,\lambda h} \cong \mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}$.

3.2 Twisted lattice algebra

For $\boldsymbol{\alpha}, \boldsymbol{\beta} \in \mathfrak{h}$, define $(\boldsymbol{\alpha}|\boldsymbol{\beta}) := (p\boldsymbol{\alpha}, p\boldsymbol{\beta})_{\mathfrak{h}} - (\bar{p}\boldsymbol{\alpha}, \bar{p}\boldsymbol{\beta})_{\mathfrak{h}}$. Let Q be an (not necessarily N-dimensional⁴) even lattice in \mathfrak{h} with respect to $(\cdot|\cdot)$, that is, an additive subgroup Q of $(\mathfrak{h}, +)$ spanned by a set of linearly independent elements of \mathfrak{h} , such that $(\boldsymbol{\alpha}|\boldsymbol{\beta}) \in \mathbb{Z}$ for any pair $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$, and $(\boldsymbol{\alpha}|\boldsymbol{\alpha}) \in 2\mathbb{Z}$ for any $\boldsymbol{\alpha} \in Q$.

⁴In the usual convention, a lattice in an N-dimensional vector space \mathfrak{h} is a \mathbb{Z} -span of a basis of \mathfrak{h} . Here we allow Q to be generated by linearly independent elements, not necessarily a basis. To stress this difference, we write "not necessarily N-dimensional".

As in [Kac98, Section 5.5], for an ordered basis v_1, \dots, v_{N_0} of $Q, N_0 \leq N$, let

$$\epsilon(\boldsymbol{v}_{i}, \boldsymbol{v}_{j}) := \begin{cases}
(-1)^{(\boldsymbol{v}_{i}|\boldsymbol{v}_{j}) + (\boldsymbol{v}_{i}|\boldsymbol{v}_{i})(\boldsymbol{v}_{j}|\boldsymbol{v}_{j})} = (-1)^{(\boldsymbol{v}_{i}|\boldsymbol{v}_{j})} & \text{if } i < j \\
(-1)^{((\boldsymbol{v}_{i}|\boldsymbol{v}_{i}) + (\boldsymbol{v}_{i}|\boldsymbol{v}_{i})^{2})/2} = (-1)^{(\boldsymbol{v}_{i}|\boldsymbol{v}_{i})/2} & \text{if } i = j , \\
1 & \text{if } i > j
\end{cases}$$
(3.1)

where the equalities hold as a special case of [Kac98, Equations (5.5.11), (5.5.12)] because the lattice Q is even, and extend ϵ to $Q \times Q$ biadditively (that is, $\epsilon(\boldsymbol{\alpha} + \boldsymbol{\beta}, \boldsymbol{\gamma}) = \epsilon(\boldsymbol{\alpha}, \boldsymbol{\gamma})\epsilon(\boldsymbol{\beta}, \boldsymbol{\gamma})$ and $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta} + \boldsymbol{\gamma}) = \epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta})\epsilon(\boldsymbol{\alpha}, \boldsymbol{\gamma})$). Then ϵ is a $\{\pm 1\}$ -valued 2-cocycle of Q, see [Kac98, Remark 5.5a], i.e. $\epsilon(\boldsymbol{\beta} + \boldsymbol{\gamma}, \boldsymbol{\alpha})\epsilon(\boldsymbol{\beta}, \boldsymbol{\gamma}) = \epsilon(\boldsymbol{\beta}, \boldsymbol{\gamma} + \boldsymbol{\alpha})\epsilon(\boldsymbol{\gamma}, \boldsymbol{\alpha}), \ \epsilon(\boldsymbol{\alpha}, \boldsymbol{0}) = \epsilon(\boldsymbol{0}, \boldsymbol{\alpha}) = 1, \ \epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta}) = (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})}\epsilon(\boldsymbol{\beta}, \boldsymbol{\alpha})$ for every $\boldsymbol{\alpha}, \boldsymbol{\beta}, \boldsymbol{\gamma} \in Q$, and it is such that $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\alpha}) = (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\alpha})/2}$ for every $\boldsymbol{\alpha} \in Q$.

We define the twisted group algebra $\mathbb{C}_{\epsilon}[Q]$ (twisted by the 2-cocycle ϵ), as the \mathbb{C} -algebra with basis $\{e_{\alpha}\}_{{\alpha}\in Q}$, with the product given by

$$e_{\alpha}e_{\beta} = \epsilon(\alpha, \beta)e_{\alpha+\beta}.$$
 (3.2)

For vectors $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$, if $\boldsymbol{\alpha} \neq \boldsymbol{\beta}$, again because the lattice is even,

$$e_{\alpha}e_{\beta} = \epsilon(\alpha, \beta)e_{\alpha+\beta} = (-1)^{(\alpha|\beta)}\epsilon(\beta, \alpha)e_{\beta+\alpha} = (-1)^{(\alpha|\beta)}e_{\beta}e_{\alpha}$$
(3.3)

and if $\alpha = \beta$, we trivially have $e_{\alpha}e_{\alpha} = (-1)^{(\alpha|\alpha)}e_{\alpha}e_{\alpha}$, as $(\alpha|\alpha) \in 2\mathbb{Z}$.

We define a (sesquilinear, linear in the second component) scalar product on $\mathbb{C}_{\epsilon}[Q]$ by $\langle e_{\alpha}, e_{\beta} \rangle := \delta_{\alpha,\beta}$. With this scalar product, the left multiplication operators by e_{α} on $\mathbb{C}_{\epsilon}[Q]$ are unitary:

$$\langle e_{\alpha}e_{\gamma}, e_{\alpha}e_{\beta} \rangle = \epsilon(\alpha, \beta)\overline{\epsilon(\alpha, \gamma)} \langle e_{\alpha+\gamma}, e_{\alpha+\beta} \rangle$$

$$= \epsilon(\alpha, \beta)\epsilon(\alpha, \gamma)\delta_{\alpha+\gamma,\alpha+\beta}$$

$$= \epsilon(\alpha, \beta)\epsilon(\alpha, \gamma)\delta_{\gamma,\beta}$$

$$= \epsilon(\alpha, \beta)^{2}\delta_{\gamma,\beta}$$

$$= \langle e_{\gamma}, e_{\beta} \rangle,$$

because $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta}), \epsilon(\boldsymbol{\alpha}, \boldsymbol{\gamma}) \in \{\pm 1\}$ for any $\boldsymbol{\alpha}, \boldsymbol{\beta}, \boldsymbol{\gamma} \in Q$.

Example 3.1. Let $\mathfrak{h} = \mathbb{R}^2 = \mathbb{R} \oplus \mathbb{R}$ with the Euclidean scalar product $(\cdot, \cdot)_{\mathfrak{h}}$. With respect to Section 3.1, we can take p as the orthogonal projection to the first component, and \bar{p} as the orthogonal projection to the second component, and define $(\cdot|\cdot)$ accordingly. Fix arbitrarily $R \in \mathbb{R}, R \neq 0$. We take the lattice $Q \subset \mathfrak{h}$ generated by $\frac{1}{\sqrt{2}}(R \oplus R)$ and $\frac{1}{\sqrt{2}}(R^{-1} \oplus (-R^{-1}))$. This is indeed an even lattice: for any $a, b, a_j, b_j \in \mathbb{Z}, j = 1, 2$, we have

$$\left(\frac{a}{\sqrt{2}}(R \oplus R) + \frac{b}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) \middle| \frac{a}{\sqrt{2}}(R \oplus R) + \frac{b}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) \right) = 2ab \in 2\mathbb{Z}, \quad (3.4)$$

$$\left(\frac{a_1}{\sqrt{2}}(R \oplus R) + \frac{b_1}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) \middle| \frac{a_2}{\sqrt{2}}(R \oplus R) + \frac{b_2}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) \right) = a_1b_2 + a_2b_1 \in \mathbb{Z}.$$
(3.5)

Note that, if $(x \oplus y) \in \mathfrak{h}$ satisfies $\left(x \oplus y \middle| \frac{1}{\sqrt{2}}(R \oplus R)\right), \left(x \oplus y \middle| \frac{1}{\sqrt{2}}(R^{-1} \oplus (-R^{-1}))\right) \in \mathbb{Z}$, then $(x \oplus y) \in \mathbb{Z} \frac{1}{\sqrt{2}}(R \oplus R) + \mathbb{Z} \frac{1}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) = Q$. Indeed, let us put

$$\left(x \oplus y \middle| \frac{1}{\sqrt{2}} (R \oplus R)\right) =: m, \left(x \oplus y \middle| \frac{1}{\sqrt{2}} (R^{-1} \oplus (-R^{-1}))\right) =: n,$$

then $x-y=\sqrt{2}R^{-1}m, x+y=\sqrt{2}Rn$, thus $x=\frac{1}{\sqrt{2}}mR^{-1}+\frac{1}{\sqrt{2}}nR, y=-\frac{1}{\sqrt{2}}mR^{-1}+\frac{1}{\sqrt{2}}nR$. This shows that Q is maximal, i.e. there is no integral lattice that includes Q as a proper sublattice.

3.3 Construction of two-dimensional conformal nets

3.3.1 The Hilbert space and basic operators

Let \mathfrak{h} and p be as in Section 3.1. For $\boldsymbol{\alpha} \in \mathfrak{h}$, we write $\boldsymbol{\alpha} = p\boldsymbol{\alpha} \oplus \bar{p}\boldsymbol{\alpha}$ the orthogonal decomposition. Let $Q \subset \mathfrak{h}$ be an (not necessarily N-dimensional) even lattice with respect to $(\boldsymbol{\alpha}|\boldsymbol{\beta}) = (p\boldsymbol{\alpha}, p\boldsymbol{\beta})_{\mathfrak{h}} - (\bar{p}\boldsymbol{\alpha}, \bar{p}\boldsymbol{\beta})_{\mathfrak{h}}$ for $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$ as before. Let us fix the 2-cocycle ϵ of Q as in Section 3.2, $I \in \mathcal{I}$ and $h \in C^{\infty}(S^1)$ such that supp $h \subset I$ and $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$ as in Section 2.3.2.

We consider the Hilbert space

$$\mathcal{H}_{Q} := \bigoplus_{\lambda \in Q} \mathcal{H}_{\mathfrak{h}_{p}, \lambda h} \cong \bigoplus_{\lambda \in Q} \mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}. \tag{3.6}$$

Using the unitary equivalence (only as Hilbert spaces) $\mathcal{H}_{\mathfrak{h}_p,\lambda h} \cong \mathcal{H}_{\mathfrak{h}_p,\mathbf{0}}$, we also have the following unitary equivalence

$$\mathcal{H}_Q \cong \mathcal{H}_{\mathfrak{h}_p,\mathbf{0}} \otimes \overline{\mathbb{C}_{\epsilon}[Q]},$$

where $\mathbb{C}_{\epsilon}[Q]$ is endowed with the scalar product $\langle e_{\alpha}, e_{\beta} \rangle = \delta_{\alpha,\beta}$ as in the previous section and completed to get a Hilbert space. The unitary left multiplication operators e_{α} act only on the second component, while the operators of the net $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ act on $\mathcal{H}_{\mathfrak{h}_p,\mathbf{0}} \otimes e_{\lambda} \cong \mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}$ by $\sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$ (see Section 2.1.3). Denote this representation of $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ on \mathcal{H}_Q by τ_Q . By (2.4), τ_Q is diffeomorphism covariant with the representation U_Q of $\overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ defined as follows

$$U_Q(\gamma_+ \times \gamma_-) := \bigoplus_{\lambda \in Q} U_{p\lambda h}(\gamma_+) \otimes U_{\bar{p}\lambda h}(\gamma_-). \tag{3.7}$$

We denote the twisted shift operator on \mathcal{H}_Q corresponding to $\boldsymbol{\alpha} \in Q$, induced by the left multiplication by $e_{\boldsymbol{\alpha}}$ on $\mathbb{C}_{\epsilon}[Q]$, as $\psi^{\boldsymbol{\alpha}}$. Let $\Psi = (\Psi_{\boldsymbol{\lambda}})_{\boldsymbol{\lambda} \in Q} \in \mathcal{H}_Q$ (where for each $\boldsymbol{\lambda}, \Psi_{\boldsymbol{\lambda}} \in \mathcal{H}_{\mathfrak{h}_p, \mathbf{0}} \otimes e_{\boldsymbol{\lambda}} \cong \mathcal{H}_{p\boldsymbol{\lambda}h} \otimes \mathcal{H}_{\bar{p}\boldsymbol{\lambda}h}$ but we parametrize the components by $\boldsymbol{\lambda} \in Q$, instead of the representation $\sigma_{p\boldsymbol{\lambda}h} \otimes \sigma_{\bar{p}\boldsymbol{\lambda}h}$). Then, definition (3.2) translates into

$$(\psi^{\alpha}\Psi)_{\alpha+\lambda} = \epsilon(\alpha, \lambda)\Psi_{\lambda}, \tag{3.8}$$

or equivalently, $(\psi^{\alpha}\Psi)_{\lambda} = \epsilon(\alpha, \lambda - \alpha)\Psi_{\lambda-\alpha}$. The commutation relations (3.3) translate into

$$\psi^{\alpha}\psi^{\beta} = (-1)^{(\alpha|\beta)}\psi^{\beta}\psi^{\alpha}, \tag{3.9}$$

where recall that $(\boldsymbol{\alpha}|\boldsymbol{\beta}) \in \mathbb{Z}$ and $(\boldsymbol{\alpha}|\boldsymbol{\alpha}) \in 2\mathbb{Z}$ for any $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$.

For a later use (Section 5), we also introduce simple shift operators without ϵ , that can be defined for any additive subgroup Q:

$$(\psi^{\alpha}\Psi)_{\alpha+\lambda} = \Psi_{\lambda}. \tag{3.10}$$

Moreover, we have

$$\operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\alpha}) = \tau_Q\left(z_{p\alpha h}(\gamma_+) \otimes z_{\bar{p}\alpha h}(\gamma_-)\right)\psi^{\alpha}$$
(3.11)

by calculations analogous to [AGT23, (2.2), Theorem 3.1]: differently from [AGT23], ψ^{α} is not a simple shift operator, but it is followed only by $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\lambda})$. As it is a scalar and $U_Q(\gamma_+ \times \gamma_-)$ preserves the spaces $\mathcal{H}_{\mathfrak{h}_p,\boldsymbol{\lambda}h}$, the results hold here as well.

Note that $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\lambda}) = \epsilon(-\boldsymbol{\alpha}, \boldsymbol{\lambda})$ by bimultiplicativity and $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\alpha}) \in \{1, -1\}$,

$$\epsilon(-\alpha, \lambda + \alpha) = \epsilon(-\alpha, \lambda)\epsilon(-\alpha, \alpha) = \epsilon(\alpha, \lambda)\epsilon(\alpha, \alpha).$$

Using this, for $\Psi, \Phi \in \mathcal{H}_Q$, we have

$$\langle \Psi, \psi^{\alpha} \Phi \rangle = \sum_{\lambda \in Q} \langle \Psi_{\lambda}, (\psi^{\alpha} \Phi)_{\lambda} \rangle$$

$$= \sum_{\lambda \in Q} \epsilon(\alpha, \lambda - \alpha) \langle \Psi_{\lambda}, \Phi_{\lambda - \alpha} \rangle$$

$$= \sum_{\lambda \in Q} \epsilon(\alpha, \lambda) \langle \Psi_{\lambda + \alpha}, \Phi_{\lambda} \rangle$$

$$= \epsilon(\alpha, \alpha) \sum_{\lambda \in Q} \epsilon(-\alpha, \lambda + \alpha) \langle \Psi_{\lambda + \alpha}, \Phi_{\lambda} \rangle$$

$$= \epsilon(\alpha, \alpha) \langle \psi^{-\alpha} \Psi, \Phi \rangle, \tag{3.12}$$

which shows that $(\psi^{\alpha})^* = \epsilon(\alpha, \alpha)\psi^{-\alpha}$.

3.3.2 The two-dimensional conformal net

In the notation of the previous section, we construct a conformal net $(\mathcal{A}_Q, U_Q, \Omega_Q)$ on \mathbb{R}^{1+1} on the Hilbert space \mathcal{H}_Q , defined above, as follows.

• The net \mathcal{A}_Q is given as follows: For the diamond $I \times I$, we set

$$\mathcal{A}_Q(I \times I) := \tau_Q(\mathcal{A}_{p\mathfrak{h}}(I) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I)) \vee \{\psi^{\boldsymbol{\alpha}}, (\psi^{\boldsymbol{\alpha}})^*\}_{\boldsymbol{\alpha} \in Q}.$$

For any other diamond $\gamma_+ I \times \gamma_- I$, where $\gamma_+ \times \gamma_- \in \overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ such that $\gamma_+ I \times \gamma_- I \subset \mathbb{R}^{1+1}$, we set

$$\mathcal{A}_Q(\gamma_+ I \times \gamma_- I) := \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\mathcal{A}_Q(I \times I)).$$

We show below that this definition does not depend on the choice of γ_+, γ_- fulfilling the above constraints.

- The $\overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ -covariance is implemented by U_Q as in the previous section.
- The vacuum vector is $\Omega_Q := \Omega_0 \otimes \Omega_0 \in \mathcal{H}_0 \otimes \mathcal{H}_0 \subset \mathcal{H}_Q$.

Theorem 3.2. The triple $(\mathcal{A}_Q, U_Q, \Omega_Q)$ is a conformal net on \mathbb{R}^{1+1} .

Proof. Let us show that the net \mathcal{A}_Q is well defined. We first take γ_{\pm} such that $\gamma_{+}I = I$, $\gamma_{-}I = I$. In this case, it is clear that $\operatorname{Ad} U_Q(\gamma_{+}I \times \gamma_{-}I)(\tau_Q(\mathcal{A}_{\mathfrak{h}}(I) \otimes \mathcal{A}_{\mathfrak{h}}(I))) = \tau_Q(\mathcal{A}_{\mathfrak{h}}(I) \otimes \mathcal{A}_{\mathfrak{h}}(I))$. By (3.11), we see that $\operatorname{Ad} U_Q(\gamma_{+} \times \gamma_{-})(\psi^{\boldsymbol{\alpha}})$ is contained in $\mathcal{A}_Q(I \times I)$, because in this case $z_{p\boldsymbol{\alpha}h} \in \mathcal{A}_{p\mathfrak{h}}(I)$, $z_{\bar{p}\boldsymbol{\alpha}h} \in \mathcal{A}_{\bar{p}\mathfrak{h}}(I)$. Similar for $(\psi^{\boldsymbol{\alpha}})^*$. Therefore, $\operatorname{Ad} U_Q(\gamma_{+} \times \gamma_{-})(\mathcal{A}_Q(I \times I)) \subset \mathcal{A}_Q(I \times I)$. By applying the same argument to $\gamma_{+}^{-1} \times \gamma_{-}^{-1}$, we see that $\operatorname{Ad} U_Q(\gamma_{+} \times \gamma_{-})(\mathcal{A}_Q(I \times I)) = \mathcal{A}_Q(I \times I)$.

Next, let us show that $\mathcal{A}_Q(\gamma_+ I \times \gamma_- I)$ is well-defined. For this, let $\gamma_{+,j}, \gamma_{-,j} \in \overline{\mathrm{Diff}_+(S^1)}, j=1,2$ and that $\gamma_{+,1}I = \gamma_{+,2}I$ and $\gamma_{-,1}I = \gamma_{-,2}I$. Then $\gamma_{+,2}^{-1}\gamma_{+,1}I = I$ and $\gamma_{-,2}^{-1}\gamma_{-,1}I = I$, and by the previous paragraph,

$$\operatorname{Ad} U_Q(\gamma_{+,1} \times \gamma_{-,1})(\mathcal{A}_Q(I \times I)) = \operatorname{Ad} U_Q(\gamma_{+,2} \times \gamma_{-,2}) \circ \operatorname{Ad} U_Q(\gamma_{+,2}^{-1} \gamma_{+,1} \times \gamma_{-,2}^{-1} \gamma_{-,1})(\mathcal{A}_Q(I \times I))$$

$$= \operatorname{Ad} U_Q(\gamma_{+,2} \times \gamma_{-,2})(\mathcal{A}_Q(I \times I)),$$

that is, $\mathcal{A}_Q(\gamma_+ I \times \gamma_- I)$ does not depend on the choice of γ_+, γ_- . Let us check the axioms (2dCN1)–(2dCN6).

- (2dCN3) follows by definition.
- To check (2dCN1), by covariance we may assume $I \subset I_+, I_-$. There are γ_+, γ_- such that $\gamma_+ I = I_+, \gamma_- I = I_-$. Then, as before, $\operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\tau_Q(\mathcal{A}_{p\mathfrak{h}}(I) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I))) = \tau_Q(\mathcal{A}_{p\mathfrak{h}}(I_+) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I_-))$ and $\operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\boldsymbol{\alpha}}) = \tau_Q(z_{p\boldsymbol{\alpha}}(\gamma_+) \otimes z_{\bar{p}\boldsymbol{\alpha}}(\gamma_-))\psi^{\boldsymbol{\alpha}}$. As we have $\tau_Q(z_{p\boldsymbol{\alpha}}(\gamma_+) \otimes z_{\bar{p}\boldsymbol{\alpha}}(\gamma_-)) \in \tau_Q(\mathcal{A}_{p\mathfrak{h}}(I_+) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I_-)) \subset \mathcal{A}_Q(I_+ \times I_-)$, it follows that $\psi^{\boldsymbol{\alpha}} \in \mathcal{A}_Q(I_+ \times I_-)$. To show that $(\psi^{\boldsymbol{\alpha}})^* \in \mathcal{A}_Q(I_+ \times I_-)$, it is enough to use (3.12).
- To check (2dCN2), again by covariance, we may assume that there are $\gamma_+, \gamma_- \in \overline{\mathrm{Diff}_+(S^1)}$ such that $\gamma_+ I$ is in the past of I, while $\gamma_-(I)$ is in the future of I (see Section 2.1.1, the other case is similar). We prove that $\mathcal{A}_Q(I \times I)$ and $\mathcal{A}_Q(\gamma_+ I \times \gamma_- I)$ commute, where

$$\mathcal{A}_{Q}(I \times I) = \tau_{Q}(\mathcal{A}_{ph}(I) \otimes \mathcal{A}_{\bar{p}h}(I)) \vee \{\psi^{\boldsymbol{\alpha}}, (\psi^{\boldsymbol{\alpha}})^{*}\},$$

$$\mathcal{A}_{Q}(\gamma_{+}I \times \gamma_{-}I) = \tau_{Q}(\mathcal{A}_{ph}(I_{+}) \otimes \mathcal{A}_{\bar{p}h}(I_{-})) \vee \{\operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi^{\boldsymbol{\alpha}}), \operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi^{\boldsymbol{\alpha}})^{*}\}.$$

- It is clear that $\tau_Q(\mathcal{A}_{ph}(I) \otimes \mathcal{A}_{\bar{p}h}(I))$ and $\tau_Q(\mathcal{A}_{ph}(I_+) \otimes \mathcal{A}_{\bar{p}h}(I_-))$ commute.
- As τ_Q is localized in $I \times I$, ψ^{α} commutes with $\tau_Q(\mathcal{A}_{p\mathfrak{h}}(I_+) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I_-))$. Similar for $(\psi^{\alpha})^*$ by using (3.12).
- Showing that $\tau_Q(\mathcal{A}_{p\mathfrak{h}}(I)\otimes\mathcal{A}_{\bar{p}\mathfrak{h}}(I))$ and $\operatorname{Ad} U_Q(\gamma_+\times\gamma_-)(\psi^{\boldsymbol{\alpha}})$ commute is equivalent to showing that $\operatorname{Ad} U_Q(\gamma_+^{-1}\times\gamma_-^{-1})(\tau_Q(\mathcal{A}_{p\mathfrak{h}}(I)\otimes\mathcal{A}_{\bar{p}\mathfrak{h}}(I)))=\tau_Q(\mathcal{A}_{p\mathfrak{h}}(\gamma_+^{-1}I)\otimes\mathcal{A}_{\bar{p}\mathfrak{h}}(\gamma_-^{-1}I)))$ and $\psi^{\boldsymbol{\alpha}}$ commute, and this holds because again τ_Q is localized in $I\times I$ and $\operatorname{Ad} \psi^{\boldsymbol{\alpha}}$ implements $\sigma_{\boldsymbol{\alpha}h}=\sigma_{p\boldsymbol{\alpha}h}\otimes\sigma_{\bar{p}\boldsymbol{\alpha}h}$, while $\gamma_+^{-1}I\times\gamma_-^{-1}I$ is spacelike from $I\times I$ and thus $\sigma_{\boldsymbol{\alpha}h}$ is trivial on $\mathcal{A}_{\mathfrak{h}}(\gamma_+^{-1}I)\otimes\mathcal{A}_{\mathfrak{h}}(\gamma_-^{-1}I)$. Similar for $\operatorname{Ad} U_Q(\gamma_+\times\gamma_-)(\psi^{\boldsymbol{\alpha}})^*$.
- Let us show that ψ^{α} and Ad $U_Q(\gamma_+ \times \gamma_-)(\psi^{\beta})$ commute (and their adjoints). By using $\psi^{\alpha}\psi^{\beta} = (-1)^{(\alpha|\beta)}\psi^{\beta}\psi^{\alpha}$ (3.9), compute

$$\psi^{\boldsymbol{\alpha}} \operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi^{\boldsymbol{\beta}}) = \psi^{\boldsymbol{\alpha}} \tau_{Q} (z_{p\boldsymbol{\beta}h}(\gamma_{+}) \otimes z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-})) \psi^{\boldsymbol{\beta}}$$

$$= \tau_{Q} (\sigma_{p\boldsymbol{\alpha}h}((z_{p\boldsymbol{\beta}h}(\gamma_{+})) \otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}(z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-}))) \psi^{\boldsymbol{\alpha}} \psi^{\boldsymbol{\beta}}$$

$$= \tau_{Q} (\sigma_{p\boldsymbol{\alpha}h}((z_{p\boldsymbol{\beta}h}(\gamma_{+})) \otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}(z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-}))) (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \psi^{\boldsymbol{\beta}} \psi^{\boldsymbol{\alpha}}$$

$$\operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi^{\boldsymbol{\beta}}) \psi^{\boldsymbol{\alpha}} = \tau_{Q} (z_{p\boldsymbol{\beta}h}(\gamma_{+}) \otimes z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-})) \psi^{\boldsymbol{\beta}} \psi^{\boldsymbol{\alpha}}.$$

To conclude that $\psi^{\alpha} \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\beta}) = \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\beta})\psi^{\alpha}$, it is enough to use (2.1) and to observe that

$$(-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} = e^{i\pi(\boldsymbol{\alpha}|\boldsymbol{\beta})}$$

$$= e^{i\pi((p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}} - (\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}})}$$

$$= \varepsilon^{+}_{p\boldsymbol{\alpha}h,p\boldsymbol{\beta}h} \varepsilon^{-}_{\bar{p}\boldsymbol{\alpha}h,\bar{p}\boldsymbol{\beta}h}$$

$$= \tau_{Q} (z_{p\boldsymbol{\beta}h}(\gamma_{+}) \otimes z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-}))^{*} \tau_{Q} (\sigma_{p\boldsymbol{\alpha}h}((z_{p\boldsymbol{\beta}h}(\gamma_{+})) \otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}(z_{\bar{p}\boldsymbol{\beta}h}(\gamma_{-}))),$$

using the definition of braiding between sectors in the last equality. As Q is even, in particular it is an integral lattice, we have $((-1)^{(\alpha|\beta)})^2 = 1$ for any pair α, β . Therefore,

$$\psi^{\boldsymbol{\alpha}} \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\boldsymbol{\beta}}) = (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\boldsymbol{\beta}}) \psi^{\boldsymbol{\alpha}}$$
$$= \operatorname{Ad} U_Q(\gamma_+ \times \gamma_-)(\psi^{\boldsymbol{\beta}}) \psi^{\boldsymbol{\alpha}}$$

as desired.

As we have $(\psi^{\alpha})^* = \epsilon(\alpha, \alpha)\psi^{-\alpha}$ and $\epsilon(\alpha, \alpha) \in \{1, -1\}$, the commutation of all other combinations follow from this.

- (2dCN4) holds because each $U_{p\lambda h} \otimes U_{\bar{p}\lambda h}$ has positive energy.
- For (2dCN5), $\Omega_{\mathbf{0}} \otimes \Omega_{\mathbf{0}} \in \mathcal{H}_{\mathbf{0}} \otimes \mathcal{H}_{\mathbf{0}} \subset \mathcal{H}_{Q}$ is cyclic for \mathcal{A}_{Q} because $\tau_{Q}(\mathcal{A}_{ph}(I) \otimes \mathcal{A}_{\bar{p}h}(I))\Omega_{\mathbf{0}} \otimes \Omega_{\mathbf{0}}$ spans $\mathcal{H}_{\mathbf{0}} \otimes \mathcal{H}_{\mathbf{0}}$, while $\psi^{\boldsymbol{\alpha}} \cdot \Omega_{\mathbf{0}} \otimes \Omega_{\mathbf{0}} \in \mathcal{H}_{p\boldsymbol{\alpha}h} \otimes \mathcal{H}_{\bar{p}\boldsymbol{\alpha}h}$ and $\mathcal{A}_{ph} \otimes \mathcal{A}_{\bar{p}h}$ is irreducibly represented there. As for uniqueness, among $U_{p\boldsymbol{\lambda}h}$ (respectively $U_{\bar{p}\boldsymbol{\lambda}h}$), only $p\boldsymbol{\lambda} = \mathbf{0}$ (respectively $\bar{p}\boldsymbol{\lambda} = \mathbf{0}$) has an invariant vector. They occur at the same time only for $\boldsymbol{\lambda} = \mathbf{0}$, and $\Omega_{\mathbf{0}} \otimes \Omega_{\mathbf{0}}$ is the unique (up to a scalar) invariant vector for $U_{p\boldsymbol{\lambda}h} \otimes U_{\bar{p}\boldsymbol{\lambda}h}$.
- To see (2dCN6), it is enough to show that the spectrum of U_Q with respect to the spacelike rotations is contained in \mathbb{N} . On each sector $\sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$, they are generated by $L_0 \otimes \mathbb{1} \mathbb{1} \otimes L_0$, where L_0 is given by (2.2), and their lowest weights are $\frac{\langle p\lambda, p\lambda \rangle}{2}$, $\frac{\langle \bar{p}\lambda, \bar{p}\lambda \rangle}{2}$. As $\lambda \in Q$, one has $\frac{\langle p\lambda, p\lambda \rangle}{2} \frac{\langle \bar{p}\lambda, \bar{p}\lambda \rangle}{2} = \frac{\langle \lambda|\lambda \rangle}{2} \in \mathbb{Z}$.

Example 3.3. The two-dimensional net \mathcal{A}_Q constructed above is an extension of $\mathcal{A}_{p\mathfrak{h}}\otimes\mathcal{A}_{\bar{p}\mathfrak{h}}$. Let now $Q\subset\mathfrak{h}\cong\mathbb{R}^2$ as in Example 3.1, generated by $\frac{1}{\sqrt{2}}(R\oplus R)$ and $\frac{1}{\sqrt{2}}(R^{-1}\oplus(-R^{-1}))$, where $R\in\mathbb{R}, R\neq 0$. The chiral conformal nets $\mathcal{A}_{p\mathfrak{h}}$ and $\mathcal{A}_{\bar{p}\mathfrak{h}}$ are not completely rational in the sense of [KLM01], in fact they are both isomorphic to the U(1)-current net $\mathcal{A}_{\mathbb{R}}$ [BMT88], where we denote $\mathcal{A}_{p\mathfrak{h}}\cong\mathcal{A}_{\mathbb{R}}\cong\mathcal{A}_{\bar{p}\mathfrak{h}}$, see Section 2.2.2.

For any $R \in \mathbb{R}$, $R \neq 0$ the (infinite index) extension $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}} \subset \mathcal{A}_{Q}$ is expected to be maximal among the local irreducible extensions of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$, cf. [Mor23, Section 6.3]. Indeed, supporting this idea, if there is a conformal extension $\mathcal{A}_{Q} \subset \mathcal{B}$ such that the decomposition of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}} \subset \mathcal{B}$ has discrete spectrum, then by Theorem 5.5 in Section 5, it must correspond to an even lattice extending Q, but we saw already that Q is maximal in Example 3.1. We do not know whether there is an extension \mathcal{B} of \mathcal{A}_{Q} with non-discrete spectrum. In general, extensions of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$ may have non-discrete spectrum, see Example 3.5.

Assume now that $R^2 \in \mathbb{Q}$, i.e. $R^2 = \frac{p}{q}$ with $p, q \in \mathbb{N}$ coprime. Then in Q there is an element $(\frac{q}{\sqrt{2}}R + \frac{p}{\sqrt{2}}R^{-1}) \oplus (\frac{q}{\sqrt{2}}R - \frac{p}{\sqrt{2}}R^{-1}) = \sqrt{2pq} \oplus 0$, and similarly $0 \oplus \sqrt{2pq}$. Therefore, there is an intermediate two-dimensional extension $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}} \subset \mathcal{A}_{\mathbb{R},\sqrt{2pq}} \otimes \mathcal{A}_{\mathbb{R},\sqrt{2pq}} \subset \mathcal{A}_{Q}$, where $\mathcal{A}_{\mathbb{R},\sqrt{2pq}}$ is the local extension of the conformal net $\mathcal{A}_{\mathbb{R}}$ on S^1 as in [BMT88, Section IV]. In particular, if R = 1, then $\mathcal{A}_{\mathbb{R},\sqrt{2}}$ is unitarily equivalent to the SU(2)-loop group net at level 1 [Was98]. Corresponding full Vertex Operator Algebras have been obtained in [Mor23, Section 6.3].

Example 3.4. Our construction depends only on the even lattice $Q \subset \mathbb{R}^N = \mathfrak{h}$ with respect to the bilinear form $(\cdot|\cdot) = (p\cdot,p\cdot)_{\mathfrak{h}} - (\bar{p}\cdot,\bar{p}\cdot)_{\mathfrak{h}}$. There are examples of such lattices where dim $p \neq \dim \bar{p}$, see [Mor23, Section 7.5] and references therein. In particular, the central charges of the chiral and antichiral components may be different.

Example 3.5. We fix an additive subgroup $G \subset \mathbb{R}$ and take $Q = \{(\alpha \oplus \alpha) \in \mathbb{R}^2 : \alpha \in G\}$, with p the projection onto the first component. Note that for any $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$, we have $(\boldsymbol{\alpha}|\boldsymbol{\beta}) = 0$. Therefore, by the same construction (here Q is not a lattice, but by the property $(\boldsymbol{\alpha}|\boldsymbol{\beta}) = 0$, we do not need the cocycle ϵ), we have a two-dimensional conformal net \mathcal{A}_Q . Clearly there are additive subgroups G such that $Q \subset \mathbb{R}^2$ is not discrete (in the relative topology), e.g. $G = \mathbb{Q}$, \mathbb{R} . These examples have been studied in [MTW18, Section 6].

4 Braided equivalences of the chiral components

4.1 Braided equivalences and extensions

We compare our construction of Example 3.3 with the results of [BKL15, Section 6.3]. Below, we construct the braided equivalence ϕ_Q corresponding to the extension $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}} \subset \mathcal{A}_Q$, cf. [BKL15, Proposition 6.7(6)], [DNO13, Theorem 3.6, Corollary 3.8], with due variations from the completely rational and finite index context. First, we recall the definition of braided tensor functor and equivalence, see e.g. [ML98, Chapter XI], [EGNO15, Chapter 4], between braided tensor categories, and of natural transformation between braided tensor functors. The categories will be strictly tensor, unitary (i.e. C^* , see e.g. [LR97]), with simple tensor unit object and not necessarily fusion.

Definition 4.1. Let $(C, \otimes_C, \mathrm{id}_C)$ and $(D, \otimes_D, \mathrm{id}_D)$ be two strict tensor categories, where we drop the subscripts in the tensor products \otimes and tensor unit objects id if no confusion arises. A **tensor** functor is a triple (ϕ, μ, η) , where $\phi : C \to D$ is a functor, μ is an invertible (unitary if C, D are C^*) natural transformation (see Definition 4.3) $\mu : \otimes \circ (\phi, \phi) \Rightarrow \phi \circ \otimes$ (i.e. a collection of invertible or unitary arrows $\mu_{x,y} : \phi(x) \otimes \phi(y) \to \phi(x \otimes y)$ for each pair of objects x, y in C), η is an invertible (unitary in the C^* case) arrow $\eta : \mathrm{id} \to \phi(\mathrm{id})$ (the identity arrow $\eta = 1$ if $\phi(\mathrm{id}) = \mathrm{id}$), such that the following diagrams commute

$$(\phi(x) \otimes \phi(y)) \otimes \phi(z) \longrightarrow \phi(x \otimes y) \otimes \phi(z) \longrightarrow \phi((x \otimes y) \otimes z)$$

$$\downarrow = \qquad \qquad \downarrow =$$

$$\phi(x) \otimes (\phi(y) \otimes \phi(z)) \longrightarrow \phi(x) \otimes \phi(y \otimes z) \longrightarrow \phi(x \otimes (y \otimes z))$$

$$id \otimes \phi(x) \longrightarrow \phi(id) \otimes \phi(x) \qquad \phi(x) \otimes id \longrightarrow \phi(x) \otimes \phi(id)$$

$$\downarrow = \qquad \qquad \downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\phi(x) \longrightarrow \phi(id \otimes x), \qquad \phi(x) \longrightarrow \phi(x \otimes id)$$

for every x, y, z objects in C, where the arrows labelled by = are identity arrows because C and D are assumed to be strict (i.e. $id \otimes x = x = x \otimes id$ and $(x \otimes y) \otimes z = x \otimes (y \otimes z)$ both in C and D). Note that the commutativity of the first diagram is a categorical 2-cocycle condition on $\mu = \{\mu_{x,y}\}_{x,y \in C}$. μ and η are respectively called the tensorator and unitor of the tensor functor.

A tensor functor is a **tensor equivalence** [EGNO15, Definition 2.4.1] if it is, in addition, an equivalence of categories [ML98, Section IV.4]. A tensor functor is called strict if all $\mu_{x,y}$ and η (hence all arrows in the diagrams) are identity arrows.

Definition 4.2. Let $(C, \otimes_C, \mathrm{id}_C, \varepsilon_C)$ and $(D, \otimes_D, \mathrm{id}_D, \varepsilon_D)$ be in addition braided with both braidings denoted by ε (hence by $\varepsilon_{x,y} : x \otimes y \to y \otimes x$ for every x, y in C or D) if no confusion arises (and assumed to be unitary in the C^* case). A braided tensor functor, or **braided functor** for short, is just a tensor functor such that, in addition, the following diagram commutes

$$\phi(x) \otimes \phi(y) \longrightarrow \phi(x \otimes y)
\downarrow \qquad \qquad \downarrow
\phi(y) \otimes \phi(x) \longrightarrow \phi(y \otimes x)$$

for every x, y objects in C, i.e. such that $\phi(\varepsilon_{x,y}) \mu_{x,y} = \mu_{y,x} \varepsilon_{\phi(x),\phi(y)}$.

A braided functor is a **braided equivalence** [EGNO15, Definition 8.1.7] if it is, in addition, an equivalence of categories. If C = D, braided/tensor equivalences are also called autoequivalences.

Note that being tensor for a functor is additional structure with constraints, while being braided for a tensor functor is just a constraint. Recall also the following

Definition 4.3. Let $\phi_1, \phi_2 : \mathcal{C} \to \mathcal{D}$ be two functors. A **natural transformation** ν from ϕ_1 to ϕ_2 , denoted by $\nu : \phi_1 \Rightarrow \phi_2$, is a collection of arrows $\nu = {\{\nu_x\}_{x \in \mathcal{C}} \text{ in } \mathcal{D}, \text{ one for every object of } \mathcal{C}, \text{ such that } \nu_x : \phi_1(x) \to \phi_2(x) \text{ in } \mathcal{D}, \text{ which is natural in } x \text{ in the sense that the following diagram commutes}$

$$\phi_1(x) \xrightarrow{\phi_1(t)} \phi_1(y)$$

$$\downarrow^{\nu_x} \qquad \qquad \downarrow^{\nu_y}$$

$$\phi_2(x) \xrightarrow{\phi_2(t)} \phi_2(y)$$

for every arrow $t: x \to y$ in C, i.e. such that $\nu_y \phi_1(t) = \phi_2(t) \nu_x$ (naturality). In words, ν intertwines ϕ_1 and ϕ_2 both on objects and on morphisms.

A natural transformation $\nu: \phi_1 \Rightarrow \phi_2$ is called a **natural isomorphism** if all the arrows ν_x are isomorphisms (i.e. invertible arrows, or unitary in the C^* case) in \mathcal{D} .

Let now $(\phi_1, \mu_1, \eta_1), (\phi_2, \mu_2, \eta_2) : \mathcal{C} \to \mathcal{D}$ be tensor functors as before. A natural transformation $\nu : \phi_1 \Rightarrow \phi_2$ is called a **tensor** natural transformation if, in addition, it makes the following diagrams commute

$$\phi_1(x) \otimes \phi_1(y) \longrightarrow \phi_1(x \otimes y) \quad \text{id} \longrightarrow \phi_1(\text{id})$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\phi_2(x) \otimes \phi_2(y) \longrightarrow \phi_2(x \otimes y), \qquad \qquad \phi_2(\text{id})$$

for every x, y objects in C, i.e. such that $\nu_{x \otimes y} \mu_{1_{x,y}} = \mu_{2_{x,y}} \nu_x \otimes \nu_y$ and $\nu_{id} \eta_1 = \eta_2$.

A tensor natural isomorphism is a tensor natural transformation which is also an isomorphism. If the tensor functors are also braided, adding the word **braided** to tensor natural transformations amounts to no additional constraint.

To construct the braided equivalence ϕ_Q associated with Example 3.3, we assume that $R^2 \notin \mathbb{Q}$. Let, as before, $h \in C^{\infty}(S^1)$ be chosen such that $\frac{1}{2\pi} \int_{-\pi}^{\pi} h(t) dt = 1$ and localized in some $I \in \mathcal{I}$. In this case,

$$\mathcal{H}_Q \cong \bigoplus_{a,b \in \mathbb{Z}} \mathcal{H}_{\mathbb{R},(\frac{a}{\sqrt{2}}R + \frac{b}{\sqrt{2}}R^{-1})h} \otimes \mathcal{H}_{\mathbb{R},(\frac{a}{\sqrt{2}}R - \frac{b}{\sqrt{2}}R^{-1})h}.$$

The representation of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$ on \mathcal{H}_{Q} is $\tau_{Q} = \bigoplus_{a,b \in \mathbb{Z}} \sigma_{(\frac{a}{\sqrt{2}}R + \frac{b}{\sqrt{2}}R^{-1})h} \otimes \sigma_{(\frac{a}{\sqrt{2}}R - \frac{b}{\sqrt{2}}R^{-1})h}$. We let \mathcal{C} be the full subcategory of $\operatorname{Rep}(\mathcal{A}_{\mathbb{R}})$ generated by the automorphisms (with categorical dimension one, thus irreducible) $\sigma_{\frac{1}{\sqrt{2}}Rh}$ and $\sigma_{\frac{1}{\sqrt{2}}R^{-1}h}$, closed under tensor products, conjugates, directs sums and subobjects in $\operatorname{Rep}(\mathcal{A}_{\mathbb{R}})$. Recall that fullness means that the hom-spaces between objects in \mathcal{C} coincide with those in $\operatorname{Rep}(\mathcal{A}_{\mathbb{R}})$. The unitary equivalence classes of irreducible objects in \mathcal{C} and $\operatorname{Rep}(\mathcal{A}_{\mathbb{R}})$, respectively, are in one-to-one correspondence with $\mathbb{Z}\frac{R}{\sqrt{2}} + \mathbb{Z}\frac{R^{-1}}{\sqrt{2}}$ and with \mathbb{R} . Moreover, by definition with h fixed, the composition (the categorical tensor product in \mathcal{C} , that we either

denote by \circ or by no symbol) of tensor powers of irreducibles in \mathcal{C} is given by $\sigma^n_{\frac{1}{\sqrt{2}}Rh}\sigma^m_{\frac{1}{\sqrt{2}}R^{-1}h}=\sigma_{(\frac{n}{\sqrt{2}}R+\frac{m}{\sqrt{2}}R^{-1})h}$ for every $n,m\in\mathbb{Z}$ (in particular the composition is commutative), and the inverse (categorical conjugate in the case of automorphisms) is $\bar{\sigma}_{\frac{1}{\sqrt{2}}Rh}=\sigma_{\frac{-1}{\sqrt{2}}Rh}, \bar{\sigma}_{\frac{1}{\sqrt{2}}R^{-1}h}=\sigma_{\frac{-1}{\sqrt{2}}R^{-1}h}$. On the irreducible objects of \mathcal{C} , which we denote below by $\sigma_{n,m}:=\sigma_{(\frac{n}{\sqrt{2}}R+\frac{m}{\sqrt{2}}R^{-1})h}$, we define ϕ_Q by

$$\phi_O(\sigma_{n,m}) := \sigma_{-n,m}.$$

On the tensor unit of \mathcal{C} (the identity automorphism id $= \sigma_{0,0}$) it holds $\phi_Q(\mathrm{id}) = \mathrm{id}$. On arrows between the irreducibles $\sigma_{n,m}$ and $\sigma_{n',m'}$ for $n,m,n',m' \in \mathbb{Z}$, which are either $\lambda 1_{\sigma_{n,m}}$ for some $\lambda \in \mathbb{C}$ if n=n' and m=m', or 0 otherwise (indeed, $\frac{n}{\sqrt{2}}R + \frac{m}{\sqrt{2}}R^{-1} = 0$ with $n,m \neq 0$ if and only if $R^2 = -\frac{m}{n} \in \mathbb{Q}$), we define ϕ_Q by linearity, namely by setting $\phi_Q(\lambda 1_{\sigma_{n,m}}) := \lambda 1_{\sigma_{-n,m}}$.

In this way, the representation τ_Q can be written as

$$\tau_Q = \bigoplus_{n,m \in \mathbb{Z}} \sigma_{n,m} \otimes \phi_Q(\bar{\sigma}_{n,m}),$$

cf. [DNO13, Section 3.2], [BKL15, Definition 4.1].

Note that $\sigma_{n,m}\sigma_{n',m'} = \sigma_{n',m'}\sigma_{n,m} = \sigma_{n+n',m+m'}$. We define ϕ_Q to be a non-strict tensor functor with tensorator μ given by the 2-cocycle $\epsilon: Q \times Q \to \{\pm 1\}$ from (3.1), namely

$$\phi_Q(\sigma_{n,m})\phi_Q(\sigma_{n',m'}) \xrightarrow{\mu_{(n,m),(n',m')}1} \phi_Q(\sigma_{n,m}\sigma_{n',m'}),$$

where 1 above is a short-hand notation for $1_{\sigma_{-n-n',m+m'}}$, the identity arrow between the common source and target automorphism $\sigma_{-n-n',m+m'}$, and the scalar with values in $\{\pm 1\}$ defined by

$$\mu_{(n,m),(n',m')} := \epsilon(\boldsymbol{\alpha}_{n,m},\boldsymbol{\alpha}_{n',m'}),$$

where $\boldsymbol{\alpha}_{n,m} := \frac{n}{\sqrt{2}}(R \oplus R) + \frac{m}{\sqrt{2}}(R^{-1} \oplus (-R^{-1})) \in Q$ for $n,m \in \mathbb{Z}$. In this notation, note that $\boldsymbol{\alpha}_{n,m} + \boldsymbol{\alpha}_{n',m'} = \boldsymbol{\alpha}_{n+n',m+m'}$. In order to exploit the definition of ϵ , we choose $\boldsymbol{v}_1 := \boldsymbol{\alpha}_{1,0} = \frac{1}{\sqrt{2}}(R \oplus R)$, $\boldsymbol{v}_2 := \boldsymbol{\alpha}_{0,1} = \frac{1}{\sqrt{2}}(R^{-1} \oplus (-R^{-1}))$ as an ordered basis of Q, and compute, using the bimultiplicative extension of (3.1) together with (3.4), (3.5), its value $\epsilon(\boldsymbol{\alpha}_{n,m},\boldsymbol{\alpha}_{n',m'}) = \epsilon(\boldsymbol{\alpha}_{1,0},\boldsymbol{\alpha}_{0,1})^{nm'} = (-1)^{nm'}$. The scalars $\mu_{(n,m),(n',m')}$ clearly define a tensorator (see Definition 4.1 and [ML98, Section XI.2, Equation (3)]) for the (non-strict) tensor functor $\phi_Q : \mathcal{C} \to \mathcal{C}$ between strict tensor categories (after extending ϕ_Q to direct sums of irreducible objects in \mathcal{C}), because ϵ is a 2-cocycle on Q. Note that ϕ_Q is a tensor autoequivalence $\mathcal{C} \to \mathcal{C}$ with inverse $\phi_Q^{-1} = \phi_Q$. Furthermore, ϕ_Q is also braided (see Definition 4.2 and [ML98, Section XI.2, Equation (10)]), namely the following diagram commutes for every $n, m, n', m' \in \mathbb{Z}$

$$\phi_{Q}(\sigma_{n,m})\phi_{Q}(\sigma_{n',m'}) \longrightarrow \phi_{Q}(\sigma_{n,m}\sigma_{n',m'})$$

$$\downarrow \qquad \qquad \downarrow$$

$$\phi_{Q}(\sigma_{n',m'})\phi_{Q}(\sigma_{n,m}) \longrightarrow \phi_{Q}(\sigma_{n',m'}\sigma_{n,m}),$$

where the horizontal arrows are $\mu_{(n,m),(n',m')}1 = (-1)^{nm'}\mathbb{1}$ above and $\mu_{(n',m'),(n,m)}\mathbb{1} = (-1)^{n'm}\mathbb{1}$ below, and the vertical arrows are given by the braidings (2.3), namely on the left

$$\varepsilon_{\phi_Q(\sigma_{n,m}),\phi_Q(\sigma_{n',m'})}^+ = \varepsilon_{\sigma_{-n,m},\sigma_{-n',m'}}^+ = \mathrm{e}^{\mathrm{i}\pi(\frac{nn'}{2}R^2 + \frac{mm'}{2}R^{-2} - \frac{nm'}{2} - \frac{mn'}{2})}\mathbb{1},$$

and on the right

$$\phi_Q(\varepsilon_{\sigma_{n,m},\sigma_{n',m'}}^+) = e^{i\pi(\frac{nn'}{2}R^2 + \frac{mm'}{2}R^{-2} + \frac{nm'}{2} + \frac{mn'}{2})} \mathbb{1}.$$

Thus, the commutativity of the previous diagram, namely the following equality of scalars

$$(-1)^{nm'}\mathrm{e}^{\mathrm{i}\pi(\frac{nn'}{2}R^2+\frac{mm'}{2}R^{-2}+\frac{nm'}{2}+\frac{nn'}{2})}=(-1)^{n'm}\mathrm{e}^{\mathrm{i}\pi(\frac{nn'}{2}R^2+\frac{mm'}{2}R^{-2}-\frac{nm'}{2}-\frac{mn'}{2})}$$

is readily checked by using e.g. $(-1) = e^{i\pi}$ and $e^{i\frac{3\pi}{2}} = e^{-i\frac{\pi}{2}}$, indeed

$$e^{i\pi(\frac{nn'}{2}R^2 + \frac{mm'}{2}R^{-2} + \frac{3nm'}{2} + \frac{mn'}{2})} = e^{i\pi(\frac{nn'}{2}R^2 + \frac{mm'}{2}R^{-2} - \frac{nm'}{2} + \frac{mn'}{2})}.$$

Thus ϕ_Q is a braided functor.

Remark 4.4. The braided autoequivalence ϕ_Q extends to a braided autoequivalence of the replete completion of \mathcal{C} in Rep $(\mathcal{A}_{\mathbb{R}})$, denoted by $\tilde{\mathcal{C}}$, obtained by including all objects in Rep $(\mathcal{A}_{\mathbb{R}})$ that are (unitarily) equivalent to some object in \mathcal{C} . If σ is irreducible in $\tilde{\mathcal{C}}$, choose a unitary arrow (intertwining operator, unique up to scalar factors) $u:\sigma\to\sigma_{n,m}$ for a unique pair $n,m\in\mathbb{Z}$. We set $\tilde{\phi}_Q(\sigma):=\sigma_{-n,m}$. If σ_1 and σ_2 are two irreducibles in $\tilde{\mathcal{C}}$, then they are either mutually inequivalent and the only intertwiner is 0, or every arrow $t:\sigma_1\to\sigma_2$ equals $\lambda u_2^*u_1$ where $\lambda\in\mathbb{C}$ is some scalar depending on t and $u_1:\sigma_1\to\sigma_{n,m}, u_2:\sigma_2\to\sigma_{n,m}$ are the unitaries chosen as before. In the latter case, we set $\tilde{\phi}_Q(t):=\lambda 1_{\sigma_{-n,m}}$. One can check that functoriality holds, i.e. $\tilde{\phi}_Q(st)=\tilde{\phi}_Q(s)\tilde{\phi}_Q(t)$ if $t:\sigma_1\to\sigma_2$ and $s:\sigma_2\to\sigma_3$. Extending to direct sums of irreducible objects gives a braided autoequivalence $\tilde{\phi}_Q:\tilde{\mathcal{C}}\to\tilde{\mathcal{C}}$, whose isomorphism class among the braided autoequivalences of $\tilde{\mathcal{C}}$ does not depend on the choice of function $h\in\mathcal{C}^\infty(S^1)$ made at the beginning.

Indeed, let $g \in \mathcal{C}^{\infty}(S^1)$ be another function with $\frac{1}{2\pi} \int_{-\pi}^{\pi} g(t) dt = 1$ and localized in some other proper interval of S^1 . There is a unitary tensor (i.e. braided tensor) natural transformation (see Definition 4.3) from $\tilde{\phi}_Q$ to the functor $\tilde{\phi}_{Q,g}:\tilde{\mathcal{C}}\to\tilde{\mathcal{C}}$ defined in the same way using g instead of h. Let $M \in \mathcal{C}^{\infty}(S^1)$ be a primitive of g - h, namely $M(t) := \frac{1}{2\pi} \int_{-\pi}^{t} (g(s) - h(s)) ds$ so that $M(-\pi) = M(\pi) = 0$. Let $u := W(\frac{-1}{\sqrt{2}}RM)$ be the Weyl unitary that intertwines $\sigma_{\frac{-1}{\sqrt{2}}Rh} (= \sigma_{-1,0})$ in our previous notation) with $\sigma_{\frac{-1}{\sqrt{2}}Rg}$, and $v := W(\frac{1}{\sqrt{2}}R^{-1}M)$ between $\sigma_{\frac{1}{\sqrt{2}}R^{-1}h}(=\sigma_{0,1})$ with $\sigma_{\frac{1}{\sqrt{2}}R^{-1}g}$. Observe that u and v commute because $\int_{-\pi}^{\pi} M(t)M'(t)dt = \frac{1}{2}[M^2]_{-\pi}^{\tilde{v}^2} = 0$, hence uv = vu $W((\frac{-1}{\sqrt{2}}R+\frac{1}{\sqrt{2}}R^{-1})M)$. For σ irreducible in $\tilde{\mathcal{C}}$, let $\nu_{\sigma}:=u\otimes\stackrel{(n)}{\dots}\otimes u\otimes v\stackrel{(m)}{\dots}\otimes v$ be the unitary intertwiner between $\tilde{\phi}_Q(\sigma) = \sigma_{-n,m}$ and $\tilde{\phi}_{Q,g}(\sigma) = \sigma_{-n,m,g} (:= \sigma_{(\frac{-n}{\sqrt{2}}R + \frac{m}{\sqrt{2}}R^{-1})g})$, where the symbol \otimes denotes the tensor product of intertwiners in the category $\text{Rep}(\mathcal{A}_{\mathbb{R}})$. Naturality of the collection of arrows $\{\nu_{\sigma}\}\$ can be checked as for functoriality, using that all arrows in the naturality diagram are either zero or scalars and that $\sigma_{n,m}$ and $\sigma_{n,m,g}$ are unitarily equivalent. To check tensoriality of the natural transformation ν , observe that the tensorators $\mu_{(n,m),(n',m')}$ only depend on n,m,n',m'(they are the 2-cocycle ε) and not on h, g, hence they are the same for the two functors. Thus, we only have to check that $\nu_{\sigma} \otimes \nu_{\sigma'} = \nu_{\sigma \otimes \sigma'}$, where recall that the tensor product of automorphisms in Rep $(\mathcal{A}_{\mathbb{R}})$ is just composition. Observe that $v \otimes u = v\sigma_{0,1}(u) = e^{\frac{i}{2\pi}\int \frac{1}{\sqrt{2}}R^{-1}h(t)\frac{-1}{\sqrt{2}}RM(t)dt}vu$ and $u\otimes v=u\sigma_{-1,0}(v)=\mathrm{e}^{\frac{\mathrm{i}}{2\pi}\int\frac{-1}{\sqrt{2}}Rh(t)\frac{1}{\sqrt{2}}R^{-1}M(t)dt}uv$, thus the phase factor is the same and we conclude $v \otimes u = u \otimes v$. Consequently, $\nu_{\sigma} \otimes \nu_{\sigma'} = \nu_{\sigma \otimes \sigma'}$ as unitary intertwiners between $\sigma_{-n-n',m+m'}$ and $\sigma_{-n-n',m+m',g}$, showing tensoriality of $\nu:\phi_Q\Rightarrow \overline{\phi}_{Q,q}$.

4.2 Braided equivalences and unitary (non-)equivalence of extensions

When one considers the problem of classifying extensions $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{B}$ up to unitary equivalence, there are two possibilities: either one requires that the "unitary equivalence" intertwines also the representation $\tau_{\mathcal{B}}$ of $\mathcal{A}_{+} \otimes \mathcal{A}_{-}$ on the Hilbert space of \mathcal{B} or not. The stronger notion of equivalence for extensions is implicitly used in [BKL15, Proposition 6.7(1)], cf. [DR90, Definition 3.4]. In [BKL15, Section 6], two extensions $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{B}_{1}$, \mathcal{B}_{2} are called unitarily equivalent if the nets \mathcal{B}_{1} , \mathcal{B}_{2} are unitarily equivalent via some unitary operator U, and the same operator intertwines the representations τ_{1} , τ_{2} of $\mathcal{A}_{+} \otimes \mathcal{A}_{-}$ obtained by restricting the vacuum representations of \mathcal{B}_{1} , \mathcal{B}_{2} , respectively, i.e. Ad $U \circ \tau_{1} = \tau_{2}$. It may happen that, even if two extensions $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{B}_{j}$, j = 1, 2 are not unitarily equivalent, two nets \mathcal{B}_{i} are unitarily equivalent. We exhibit this in examples. In order to stress that we consider the stronger notion, we say that the extensions $\mathcal{A}_{+} \otimes \mathcal{A}_{-} \subset \mathcal{B}_{j}$ are unitarily equivalent.

Let us consider the extensions of Example 3.5 with $G = \mathbb{Z}$: $\mathfrak{h} = \mathbb{R}^2$, p is the projection onto the first component, $Q = \{(\alpha \oplus \alpha) \in \mathbb{R}^2 : \alpha \in \mathbb{Z}\}$, and we obtain a two-dimensional conformal net \mathcal{A}_Q by Theorem 3.2. On the other hand, we can also take $Q' = \{(\alpha \oplus -\alpha) \in \mathbb{R}^2 : \alpha \in \mathbb{Z}\}$, and we obtain another two-dimensional conformal net $\mathcal{A}_{Q'}$ (the same net constructed in [AGT23, Section 6]). Both nets extend the tensor product of U(1)-current nets $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$. We discuss the equivalence of these two-dimensional conformal extensions at the end of the section.

Note that on $\mathcal{A}_{\mathbb{R}}(I)$ there is an automorphism $\theta(W(f)) = W(-f)$, This is implemented by the unitary operator V which acts as 1 on the even particle number space and -1 on the odd particle number space, in particular, $V^2 = 1$. Fix $h \in C^{\infty}(S^1)$ localized in some $I \in \mathcal{I}$ as before.

We have $\operatorname{Ad} V \circ \sigma_{\alpha h} = \sigma_{-\alpha h} \circ \operatorname{Ad} V$ elementwise on $\mathcal{A}_{\mathbb{R}}$, indeed

$$\operatorname{Ad} V \circ \sigma_{\alpha h}(W(f)) = e^{\frac{\mathrm{i}\alpha}{2\pi} \int f(t)h(t)dt} \operatorname{Ad} V(W(f)) = e^{\frac{\mathrm{i}\alpha}{2\pi} \int f(t)h(t)dt} (W(-f))$$
$$= e^{-\frac{\mathrm{i}\alpha}{2\pi} \int (-f(t))h(t)dt} (W(-f)) = \sigma_{-\alpha h}(W(-f)) = \sigma_{-\alpha h} \circ \operatorname{Ad} V(W(f)).$$

As Ad V is a vacuum-preserving automorphism of the net $\mathcal{A}_{\mathbb{R}}$, it commutes with its implementation of the diffeomorphisms $U(\gamma)$ [CW05], [CLTW12, Proposition 4.2], [CKLW18, Theorem 6.10] (one can also see this from Ad $(J_n) = -J_n$ and the fact that L_n in (2.2) are quadratic in J). As the representations $U_{\alpha h}$ on $\mathcal{H}_{\alpha h}$ is (locally) given by $U_{\alpha h}(\gamma) = \sigma_{\alpha h}(U(\gamma))$, we have

$$\operatorname{Ad}V(U_{\alpha h}(\gamma)) = \operatorname{Ad}V \circ \sigma_{\alpha h}(U(\gamma)) = \sigma_{-\alpha h} \circ \operatorname{Ad}V(U(\gamma)) = \sigma_{-\alpha h}(U(\gamma)) = U_{-\alpha h}(\gamma).$$

Let $\tilde{V} := \bigoplus_{\alpha \in \mathbb{Z}} \mathbb{1} \otimes V$, we clearly have $\tilde{V}(\bigoplus_{\alpha \in \mathbb{Z}} \mathcal{H}_{\alpha h} \otimes \mathcal{H}_{\alpha h}) = \bigoplus_{\alpha \in \mathbb{Z}} \mathcal{H}_{\alpha h} \otimes \mathcal{H}_{-\alpha h}$ i.e. $\tilde{V} : \mathcal{H}_Q \to \mathcal{H}_{Q'}$, and recall the notation for the representations of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$ respectively $\tau_Q = \bigoplus_{\alpha \in \mathbb{Z}} \sigma_{\alpha h} \otimes \sigma_{\alpha h}$ on \mathcal{H}_Q and $\tau_{Q'} = \bigoplus_{\alpha \in \mathbb{Z}} \sigma_{\alpha h} \otimes \sigma_{-\alpha h}$ on $\mathcal{H}_{Q'}$. Furthermore, let us denote by $\psi_Q^{\boldsymbol{\alpha}}$, $\psi_{Q'}^{\boldsymbol{\alpha}}$, where $\boldsymbol{\alpha} \in Q$ or Q', the corresponding shift operators (twisted by the 2-cocycle of Q or Q') as in Section 3.1. Then we have

- Ad $\tilde{V} \circ \tau_Q = \tau_{Q'} \circ \text{Ad } \tilde{V}$ elementwise on $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$ (but not Ad $\tilde{V} \circ \tau_Q = \tau_{Q'}$, i.e. \tilde{V} is not a unitary intertwiner between the two representations of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$, as V is clearly not a unitary intertwiner between the irreducibles $\rho_{\alpha h}$ and $\rho_{-\alpha h}$, which is impossible unless $\alpha = 0$).
- Ad $\tilde{V}(U_Q(\gamma_+ \times \gamma_-)) = U_{Q'}(\gamma_+ \times \gamma_-).$
- $\tilde{V}\Omega_Q = \Omega_{Q'}$.
- Ad $\tilde{V}(\psi_Q^{\alpha}) = \psi_{Q'}^{\alpha}$ (to show this, we identify $\mathcal{H}_{\alpha h} = \mathcal{H}_0$ as Hilbert spaces, and note that V acts on the same unitary on each of them and $\psi_Q^{\alpha}, \psi_{Q'}^{\alpha}$ do only the shift).

• For $I \times I$, where I is the distinguished interval containing supp h, we have

$$\operatorname{Ad} \tilde{V}(\mathcal{A}_{Q}(I \times I)) = \operatorname{Ad} \tilde{V}(\{\tau_{Q}(W(f_{+}) \otimes W(f_{-})) : \operatorname{supp} f_{\pm} \subset I, \psi_{Q}^{\boldsymbol{\alpha}}\}'')$$

$$= \{\tau_{Q'}(W(f_{+}) \otimes W(-f_{-})) : \operatorname{supp} f_{\pm} \subset I, \psi_{Q'}^{\boldsymbol{\alpha}}\}''$$

$$= \{\tau_{Q'}(W(f_{+}) \otimes W(f_{-})) : \operatorname{supp} f_{\pm} \subset I, \psi_{Q'}^{\boldsymbol{\alpha}}\}''$$

$$= \mathcal{A}_{Q'}(I \times I),$$

where the equality is global on the local algebras over $I \times I$ but not elementwise.

- For any other double cone, we have the equality by covariance, hence \mathcal{A}_Q and $\mathcal{A}_{Q'}$ are unitarily equivalent via \tilde{V} as two-dimensional conformal nets per se.
- Moreover, the two copies of the two-dimensional Heisenberg net (two subnets of \mathcal{A}_Q and $\mathcal{A}_{Q'}$) are also unitarily equivalent via \tilde{V} , globally on the local algebras over $I \times I$ hence for every double cone, indeed

$$\operatorname{Ad} \tilde{V}(\tau_{Q}(\mathcal{A}_{\mathbb{R}}(I) \otimes \mathcal{A}_{\mathbb{R}}(I)) = \operatorname{Ad} \tilde{V}(\{\tau_{Q}(W(f_{+}) \otimes W(f_{-})) : \operatorname{supp} f_{\pm} \subset I\}'')$$

$$= \{\tau_{Q'}(W(f_{+}) \otimes W(-f_{-})) : \operatorname{supp} f_{\pm} \subset I\}'')$$

$$= \{\tau_{Q'}(W(f_{+}) \otimes W(f_{-})) : \operatorname{supp} f_{\pm} \subset I\}'')$$

$$= \tau_{Q'}(\mathcal{A}_{\mathbb{R}}(I) \otimes \mathcal{A}_{\mathbb{R}}(I)).$$

Hence $\mathcal{A}_Q \supset \tau_Q(\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}})$ and $\mathcal{A}_{Q'} \supset \tau_{Q'}(\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}})$ are not unitarily equivalent as extensions of nets, as τ_Q and $\tau_{Q'}$ are not unitarily equivalent representations of $\mathcal{A}_{\mathbb{R}} \otimes \mathcal{A}_{\mathbb{R}}$. However, the two nets $\mathcal{A}_Q, \mathcal{A}_{Q'}$ are unitarily equivalent as two-dimensional conformal nets.

Remark 4.5. The braided autoequivalences (cf. [BKL15, Proposition 6.7(1)] in the completely rational and finite index context) corresponding to the extensions are, respectively, $\phi_Q(\sigma_{\alpha h}) = \sigma_{-\alpha h}$ and $\phi_{Q'}(\sigma_{\alpha h}) = \sigma_{\alpha h}$ (the identity autoequivalence). In particular, there is no natural transformation $\nu: \phi_Q \Rightarrow \phi_{Q'}$ (see Definition 4.3) as there is no unitary intertwiner between $\sigma_{-\alpha h}(=\bar{\sigma}_{\alpha h})$ and $\sigma_{\alpha h}$ for $\alpha \in \mathbb{R}$, unless $\alpha = 0$.

Let again $\mathcal{A}_{+}\otimes\mathcal{A}_{-}\subset\mathcal{B}_{1}$, \mathcal{B}_{2} be two extensions realized on the Hilbert spaces \mathcal{H}_{1} , \mathcal{H}_{2} , such that \mathcal{A}_{+} , \mathcal{A}_{-} are the maximal chiral components of \mathcal{B}_{1} , \mathcal{B}_{2} in the sense of [Reh00]. Assume that there is a unitary operator $V:\mathcal{H}_{1}\to\mathcal{H}_{2}$ such that $\operatorname{Ad}V(\mathcal{B}_{1}(D))=\mathcal{B}_{2}(D)$ for any double cone D, $\operatorname{Ad}V(U_{1}(\gamma))=U_{2}(\gamma)$ for any $\gamma\in\overline{\operatorname{Diff}_{+}(S^{1})}\times\overline{\operatorname{Diff}_{+}(S^{1})}$ and $V\Omega_{1}=\Omega_{2}$. By the second property, $\operatorname{Ad}V$ intertwines the actions of diffeomorphisms, thus the maximal chiral components are intertwined as well, in particular $\operatorname{Ad}V(\tau_{1}(\mathcal{A}_{+}(I)\otimes\mathbb{1}))=\tau_{2}(\mathcal{A}_{+}(I)\otimes\mathbb{1})$, $\operatorname{Ad}V(\tau_{1}(\mathbb{1}\otimes\mathcal{A}_{-}(I)))=\tau_{2}(\mathbb{1}\otimes\mathcal{A}_{-}(I))$, where τ_{1},τ_{2} are the representations of $\mathcal{A}_{+}\otimes\mathcal{A}_{-}$ on $\mathcal{H}_{1},\mathcal{H}_{2}$, respectively. Denoting the restriction of the latter to their vacuum parts by $\tau_{1,0},\tau_{2,0}$, respectively, we have that $\operatorname{Ad}V(\tau_{1,0}(\mathcal{A}_{+}(I)\otimes\mathbb{1}))=\tau_{2,0}(\mathcal{A}_{+}(I)\otimes\mathbb{1})$ and $\operatorname{Ad}V(\tau_{1,0}(\mathbb{1}\otimes\mathcal{A}_{-}(I)))=\tau_{2,0}(\mathbb{1}\otimes\mathcal{A}_{-}(I))$. If we identify $\tau_{1,0}=\tau_{2,0}$, this means that $\operatorname{Ad}V$ gives a vacuum-preserving automorphism of $\mathcal{A}_{+}\otimes\mathcal{A}_{-}$. This then must intertwine all representations that appear in τ_{1} and τ_{2} .

5 Classification of two-dimensional extensions

We classify two-dimensional extensions $\mathcal{A}_{\mathfrak{h}_p} \subset \mathcal{B}$ of Heisenberg nets. More precisely, we consider the following situation: let \mathfrak{h} be an N-dimensional real Hilbert space, p an orthogonal projection

on \mathfrak{h} , and $\mathcal{A}_{\mathfrak{h}_p} := \mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ the two-dimensional conformal net with its chiral components both being Heisenberg nets as in Section 3.1. Furthermore, we assume that \mathcal{B} is a conformal extension of $\mathcal{A}_{\mathfrak{h}_p}$ in the sense of Section 2.1.3. This implies that, for (lightlike) translations T_t , we have $U(T_t \times \iota), U(\iota \times T_t) \in \overline{\bigcup_{I_{\pm} \in \mathcal{I}} \tau_{\mathcal{B}}(\mathcal{A}_{p\mathfrak{h}}(I_+) \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}(I_-))}$, where $\tau_{\mathcal{B}}$ is the representation of $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$ on the Hilbert space \mathcal{H} of the net \mathcal{B} . We wish to classify two-dimensional conformal extensions $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}} \subset \mathcal{B}$, in the sense of Section 2.1.3, up to unitary equivalence (see Section 4.2).

In full generality, the problem looks rather difficult. Indeed, by [Lon03, Section 3.3], [Car04, Section 2.3], \mathcal{H} can be seen as a representation of $\mathcal{A}_{\mathfrak{h}_p}$ and assuming that it decomposes into irreducible representations, their unitary equivalence classes should form a subgroup of the additive group \mathfrak{h} with respect to the fusion rule. Thus we would need to know all possible additive subgroups of $\mathfrak{h} \cong \mathbb{R}^N$, but we are unaware of a nice such parametrization.

Instead, if we require that such a subgroup forms a discrete subgroup of \mathfrak{h} (with the induced topology), then it is a (not necessarily N-dimensional) lattice in $\mathfrak{h} \cong \mathbb{R}^N$ [Neu99, (4.2) Proposition]. Under this assumption, we classify all such conformal extensions $\mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}} \subset \mathcal{B}$. The result (see Theorem 5.5) is that any such extension corresponds to a (not necessarily N-dimensional) lattice which is even (thus necessarily integral, i.e. \mathbb{Z} -valued) with respect to $(\cdot|\cdot)$ defined as in Section 3.2, and to any such even lattice Q, there is a unique extension \mathcal{A}_Q from Section 3.3.

The discreteness assumption has appeared also in [BMT88] in their classification of relatively local extensions of the U(1)-current net. Here, for some basic parts, our arguments closely follow the one-dimensional case of [BMT88, Proposition 3.1].

Lemma 5.1. Let \mathcal{A}_+ , \mathcal{A}_- be conformal nets on S^1 , $\mathcal{A}_+ \otimes \mathcal{A}_-$ be the conformal net on \mathbb{R}^{1+1} as in Section 2.1.3 and \mathcal{B} be a conformal extension of $\mathcal{A}_+ \otimes \mathcal{A}_-$ in the sense specified above. Let $I_+ \times I_- \subset \mathbb{R}^{1+1}$ (in the sense of Section 2.1.1). Then $\mathcal{A}_+(I'_+) \otimes \mathbb{I}$ commutes with $\mathcal{B}(I_+ \times I_-)$.

Proof. We take $I \in \mathcal{I}$ such that $\overline{I} \subset I'_+$. There is $\gamma \in \overline{\text{M\"ob}}$ (acting on \mathbb{R}) that maps I_+ onto itself and $\gamma I \subset \mathbb{R}$. With such γ , it holds that $\operatorname{Ad} U(\gamma \times \iota)(\mathcal{B}(I_+ \times I_-)) = \mathcal{B}(I_+ \times I_-)$, while $\operatorname{Ad} U(\gamma \times \iota)(\mathcal{A}(I) \otimes \mathbb{1}) = \mathcal{A}_+(\gamma I) \otimes \mathbb{1}$. Hence, as $\mathcal{B}(I_+ \times I_-)$ and $\mathcal{A}_+(\gamma I) \otimes \mathbb{1}$ commute, so do $\mathcal{B}(I_+ \times I_-)$ and $\mathcal{A}_+(I) \otimes \mathbb{1}$. As $I \subset I_+$ was arbitrary under the condition $\overline{I} \subset I'_+$, we conclude that $\mathcal{B}(I_+ \times I_-)$ and $\mathcal{A}_+(I'_+) \otimes \mathbb{1}$ commute.

Recall that a representation ρ of a conformal net \mathcal{A}_{κ} on S^1 is said to be **factorial** if $\rho(\mathcal{A}_{\kappa})'' := \overline{\bigcup_{I \in \mathcal{I}} \rho_I(\mathcal{A}_{\kappa}(I))}$ is a factor.

Proposition 5.2. Assume that A_{\pm} are type I conformal nets [LT18, Section 3.2], that is, all their factorial representations ρ are irreducible, thus $\rho(A_{\kappa})''$ is a type I factor. Then any factorial representation (in particular, irreducible) of the two-dimensional conformal net $A_{+} \otimes A_{-}$ is of the form $\rho_{+} \otimes \rho_{-}$ for some representations ρ_{+}, ρ_{-} of A_{+}, A_{-} , respectively.

Proof. A representation ρ of $\mathcal{A}_{+} \otimes \mathcal{A}_{-}$ can be restricted to $\mathcal{A}_{+} \otimes \mathbb{1}$ and $\mathbb{1} \otimes \mathcal{A}_{-}$, respectively, giving representations of \mathcal{A}_{+} , \mathcal{A}_{-} as conformal nets on S^{1} (see Section 2.1.3). For any pair I_{+} , I_{-} such that $|I_{\pm}| < \pi$, so that $I_{+} \cup I_{-} \subset I$ for some $I \in \mathcal{I}$, we see that the images of $\mathcal{A}_{+}(I_{+})$ and $\mathcal{A}_{-}(I_{-})$ under the respective restrictions of ρ commute. By additivity, $\rho(\mathcal{A}_{+}(I_{+}) \otimes \mathbb{1})$ and $\rho(\mathbb{1} \otimes \mathcal{A}_{-}(I_{-}))$ commute for arbitrary $I_{\pm} \in \mathcal{I}$.

The center $\rho(\mathcal{A}_+\otimes \mathbb{1})''\cap \rho(\mathcal{A}_+\otimes \mathbb{1})'$ is included in the center

$$\rho(\mathcal{A}_{+}\otimes\mathcal{A}_{-})''\cap\rho(\mathcal{A}_{+}\otimes\mathcal{A}_{-})'=\rho(\mathcal{A}_{+}\otimes\mathcal{A}_{-})''\cap\rho(\mathcal{A}_{+}\otimes\mathbb{1})'\cap\rho(\mathbb{1}\otimes\mathcal{A}_{-})',$$

as we saw that $\rho(\mathcal{A}_{+} \otimes \mathbb{1})'' \cap \rho(\mathcal{A}_{+} \otimes \mathbb{1})' \subset \rho(\mathcal{A}_{+} \otimes \mathbb{1})'' \subset \rho(\mathbb{1} \otimes \mathcal{A}_{-})'$. As ρ is factorial, both centers must be trivial. That is, the restriction of ρ to \mathcal{A}_{+} is factorial. The same holds for \mathcal{A}_{-} .

By assumption, \mathcal{A}_+ , \mathcal{A}_- are of type I. Thus $\rho(\mathcal{A}_+ \otimes \mathbb{1})''$ and $\rho(\mathbb{1} \otimes \mathcal{A}_-)''$ are type I factors and, as they commute, the representation ρ is of the tensor product form.

Let $\mathcal{A}_{\mathfrak{h}_p} \subset \mathcal{B}$ be a two-dimensional conformal extension, with \mathcal{B} not necessarily local. Assume that the restriction $\tau_{\mathcal{B}}$ of the vacuum representation of \mathcal{B} on the Hilbert space \mathcal{H} to the subnet $\mathcal{A}_{\mathfrak{h}_p}$ is a direct sum of irreducible representations of $\mathcal{A}_{\mathfrak{h}_p}$. Combining Proposition 5.2 and [LT18, Example 3.9, Proposition 3.11] with the classification of irreducible representations of $\mathcal{A}_{\mathbb{R}}$ in [BMT88, Section 2B, 3B][CW] (see Section 2.3.2), each such irreducible representation is of the form $\sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$ up to unitary equivalence. Thus we may and do assume that $\mathcal{H} = \bigoplus_{\lambda \in Q} \mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}$ as the representation $\tau_{\mathcal{B}} = \bigoplus_{\lambda \in Q} \sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$ of $\mathcal{A}_{\mathfrak{h}_p} = \mathcal{A}_{p\mathfrak{h}} \otimes \mathcal{A}_{\bar{p}\mathfrak{h}}$, and $Q \subset \mathfrak{h} = \mathbb{R}^N$ is a collection of points $\lambda \in Q$, possibly counted with multiplicity, for which $\sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$ appears in the decomposition. Note that $\mathbf{0} \in Q$, as $\mathcal{A}_{\mathfrak{h}_p}$ and \mathcal{B} have the same vacuum vector (cyclic for $\mathcal{A}_{\mathfrak{h}_p}$ on the subspace $\mathcal{H}_{\mathbf{0}} \otimes \mathcal{H}_{\mathbf{0}}$). We take $h \in \mathcal{C}^{\infty}(S^1)$ as in Section 2.3.2 in such a way that supp $h \subset I$, for some fixed $I \in \mathcal{I}$.

The following is an adaptation of parts of [BMT88, Proposition 3.1] to the two-dimensional case (locality of \mathcal{B} is not used here, as in the original proof).

Lemma 5.3. Let $\mathcal{A}_{\mathfrak{h}_p} \subset \mathcal{B}$ be as above, and assume that $Q \subset \mathbb{R}^N$ is a discrete subset of \mathbb{R}^N with respect to the usual topology of \mathbb{R}^N . Then for each $\boldsymbol{\alpha} \in Q$ and double cone $O = I_+ \times I_-$, there is a $\phi^{\boldsymbol{\alpha}} \in \mathcal{B}(O)$ such that $\phi^{\boldsymbol{\alpha}} \Omega \in \mathcal{H}_{p\boldsymbol{\alpha}h} \otimes \mathcal{H}_{\bar{p}\boldsymbol{\alpha}h}$. Moreover, each $\boldsymbol{\lambda} \in Q$ comes with multiplicity 1 (i.e. $\sigma_{p\boldsymbol{\alpha}h} \otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}$ appears with multiplicity 1 in $\tau_{\mathcal{B}}$), and $Q \subset \mathbb{R}^N$ is an additive subgroup.

Proof. By taking a basis on $p\mathfrak{h}, \bar{p}\mathfrak{h}$ we identify \mathfrak{h} with \mathbb{R}^N and define the operators \mathfrak{q}_j on \mathcal{H} , $j=1,\cdots,N$, that act by multiplication with λ_j on each direct summand $\mathcal{H}_{p\lambda h}\otimes\mathcal{H}_{\bar{p}\lambda h}$ of \mathcal{H} , $\lambda\in Q$.

For $\mathbf{s} \in \mathbb{R}^N$, let $\mathbf{s} \cdot \mathbf{q} := \sum_j s_j \mathbf{q}_j$ and take the unitary $e^{i\mathbf{s} \cdot \mathbf{q}}$ on \mathcal{H} . We claim that it holds that $\operatorname{Ad} e^{i\mathbf{s} \cdot \mathbf{q}}(\mathcal{B}(O)) = \mathcal{B}(O)$ for any double cone $O = I_+ \times I_-$. Indeed, let us pick a component j, the j-th canonical basis vector $\mathbf{e}_j = (0, \dots, 0, \frac{1}{j-\text{th}}, 0, \dots, 0)$ and $f_1, f_2 \in C^{\infty}(S^1)$ such that $f_1 + f_2 = 1$, supp $f_i \subset I_i, i = 1, 2$ and $\{I_1, I_2\}$ is a cover of S^1 , and $s \in \mathbb{R}$.

In the vacuum direct summand $\mathcal{H}_{p\mathbf{0}h} \otimes \mathcal{H}_{\bar{p}\mathbf{0}h}$, recalling the notation from Section 2.3.2, it holds that $e^{\frac{1}{4\pi i}\int sf_1'(t)sf_2(t)dt}e^{i\mathbf{e}_j(sf_1)}e^{i\mathbf{e}_j(sf_2)}=e^{i\mathbf{e}_j(s)}=1$, where s is the constant function, hence $\mathbf{e}_j(s)=J_0$ which is the zero operator on $\mathcal{H}_{p\mathbf{0}h} \otimes \mathcal{H}_{\bar{p}\mathbf{0}h}$. Therefore, in the representation (automorphism) denoted for simplicity $\sigma_{\lambda h} := \sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$, for $s \in \mathbb{R}$, it holds that

$$\begin{split} & \mathrm{e}^{\frac{1}{4\pi\mathrm{i}}\int sf_1'(t)sf_2(t)dt} \sigma_{\pmb{\lambda}h} \big(\mathrm{e}^{\mathrm{i}\pmb{e}_j(sf_1)} \big) \sigma_{\pmb{\lambda}h} \big(\mathrm{e}^{\mathrm{i}\pmb{e}_j(sf_2)} \big) \\ & = \mathrm{e}^{\frac{1}{4\pi\mathrm{i}}\int sf_1'(t)sf_2(t)dt} \mathrm{e}^{\frac{\mathrm{i}}{2\pi}\int sf_1(t)\lambda_j h(t)dt} \mathrm{e}^{\mathrm{i}\pmb{e}_j(sf_1)} \mathrm{e}^{\frac{\mathrm{i}}{2\pi}\int sf_2(t)\lambda_j h(t)dt} \mathrm{e}^{\mathrm{i}\pmb{e}_j(sf_2)} \\ & = \mathrm{e}^{\frac{\mathrm{i}s\lambda_j}{2\pi}\int (f_1(t)+f_2(t))h(t)dt} \mathbbm{1} \\ & = \sigma_{\pmb{\lambda}h} \big(\mathrm{e}^{\mathrm{i}s\lambda_j} \, \mathbbm{1} \big). \end{split}$$

This shows that, on \mathcal{H} , for $s_i \in \mathbb{R}$ it holds that

$$e^{\frac{1}{4\pi i}\int s_j f_1'(t)s_j f_2(t)dt} \tau_{\mathcal{B}}(e^{i\mathbf{e}_j(s_j f_1)}) \tau_{\mathcal{B}}(e^{i\mathbf{e}_j(s_j f_2)}) = e^{is_j \mathfrak{q}_j}.$$

Note that this holds for any decomposition $1 = f_1 + f_2$ under the above conditions.

As \mathcal{B} is an extension of $\mathcal{A}_{\mathfrak{h}_p}$, for any $\mathbf{s} \in \mathbb{R}^N$, it holds that $\operatorname{Ad} \operatorname{e}^{\mathrm{i}\mathbf{s}\cdot\mathfrak{q}}(\mathcal{B}(I_+ \times I_-)) \subset \mathcal{B}(I_+ \times I_-)$. Indeed, we may assume that $\mathbf{s} = s\mathbf{e}_j$ for some j in the first $\dim p\mathfrak{h}$ entries (the other case is similar). Then we can write $e^{i\mathbf{s}\cdot\mathbf{q}}$, where then $\mathbf{s}\cdot\mathbf{q}=s\mathbf{q}_j$, as a product of two Weyl operators as above. We can take f_1, f_2 in such a way that supp $f_2 \subset I'_+$, and $f_1+f_2=1$ on S^1 . By Lemma 5.1, $\operatorname{Ad} e^{i\mathbf{s}\cdot\mathbf{q}}(\mathcal{B}(I_+\times I_-))=\operatorname{Ad} \tau_{\mathcal{B}}(e^{is\mathbf{e}_j(f_1)})(\mathcal{B}(I_+\times I_-))\subset \mathcal{B}(\operatorname{supp} f_1\times I_-)$. As the decomposition $f_1+f_2=1$ is arbitrary under the condition that $\operatorname{supp} f_2\subset I'_+$, $\operatorname{supp} f_1$ can be any closed interval containing I_+ . In particular, $\operatorname{Ad} e^{i\mathbf{s}\cdot\mathbf{q}}(\mathcal{B}(I_+\times I_-))=\mathcal{B}(I_+\times I_-)$.

We claim that the set Q forms an additive subgroup in \mathbb{R}^N . Let us fix $O = I_+ \times I_-$. Now we have the situation where the locally compact group \mathbb{R}^N has a unitary representation $V(\mathbf{s}) := \mathrm{e}^{\mathrm{i}\mathbf{s}\cdot\mathbf{q}}$, Ad $V(\mathbf{s})$ is an automorphism of $\mathcal{B}(O)$, $V(\mathbf{s})\Omega = \Omega$ and Ω is cyclic and separating for $\mathcal{B}(O)$. As in [Bau95, Section 1.8], we can define the Arveson spectrum of Ad V, and under these conditions, it coincides with the joint spectrum of $\mathfrak{q}_1, \ldots, \mathfrak{q}_N$. By assumption, the joint spectrum is discrete, and we can define the Arveson spectrum $\mathcal{B}(O, \operatorname{Ad} V, \lambda)$ of Ad V for one point λ , and by (an analogue of) [Bau95, Theorem 1.8.4], we have

$$\mathcal{B}(O, \operatorname{Ad} V, \lambda) = \{x \in \mathcal{B}(O), x\Omega \in \mathcal{H}_{\lambda h}\},\$$

where we denote for simplicity $\mathcal{H}_{\lambda h} := \mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}$. In particular, for each $\alpha \in Q$, we can choose $\phi^{\alpha} \in \mathcal{B}(O, \operatorname{Ad} V, \alpha), \ \phi^{\alpha} \neq 0$, thus $\phi^{\alpha} \Omega \in \mathcal{H}_{\alpha h}$.

We claim that $x = \phi^{\boldsymbol{\alpha}*}\phi^{\boldsymbol{\alpha}}$ belongs to $\tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(O))$, $O = I_+ \times I_-$. Indeed, it is invariant under $\operatorname{Ad}V(\boldsymbol{s})$, and it can be restricted to $\mathcal{H}_{\boldsymbol{0}h}$. The restriction $x|_{\mathcal{H}_{\boldsymbol{0}h}}$ commutes with $\mathcal{A}_{\mathfrak{h}_p}(I'_+ \times I'_-)$, thus by Haag duality on $\mathcal{H}_{\boldsymbol{0}h}$, there is an element y of $\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-)$ which coincides with the restriction of $x|_{\mathcal{H}_{\boldsymbol{0}h}}$ to $\mathcal{H}_{\boldsymbol{0}h}$. By the separating property of Ω for $\mathcal{B}(I_+ \times I_-)$, $x = \tau_{\mathcal{B}}(y)$, proving the claim.

Next, we claim that we can take ϕ^{α} to be an isometry. Let $\phi^{\alpha} = v^{\alpha}(\phi^{\alpha*}\phi^{\alpha})^{\frac{1}{2}}$ be the polar decomposition. We may assume that the phase v^{α} is an isometry. Indeed, if it is only a partial isometry, then $v^{\alpha*}v^{\alpha}$ is a projection and it belongs to $\tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-))$ by the same argument as above. As $\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-)$ is a type III factor, we can find an isometry $u \in \mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-)$ such that $\tau_{\mathcal{B}}(uu^*) = v^*_{\alpha}v_{\alpha}$, and $v_{\alpha}\tau_{\mathcal{B}}(u) \in \mathcal{B}(O)$ satisfies the desired property.

We show that Q is an additive subgroup of \mathbb{R}^N . For this purpose, let $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$ and take $\phi^{\boldsymbol{\alpha}}, \phi^{\boldsymbol{\beta}} \in \mathcal{B}(O)$ isometries as above. As $\phi^{\boldsymbol{\beta}*}\phi^{\boldsymbol{\alpha}*}\phi^{\boldsymbol{\alpha}}\phi^{\boldsymbol{\beta}} = 1$, we have $\phi^{\boldsymbol{\alpha}}\phi^{\boldsymbol{\beta}}\Omega \neq 0$, and it is in $\mathcal{H}_{(\boldsymbol{\alpha}+\boldsymbol{\beta})h}$. Next, Ω is separating for $\mathcal{B}(I)$, $\phi^{\boldsymbol{\alpha}*}\Omega \neq 0$, and it is in $\mathcal{H}_{-\boldsymbol{\alpha}h}$. Clearly, $\mathbf{0} \in Q$. This shows that Q is an additive subgroup of \mathbb{R}^N .

Finally, we show that λ has multiplicity 1. To see this, let Ψ, Φ , with $\Phi \neq 0$, be vectors that belong to $\mathcal{H}_{\lambda h} \oplus \ldots \oplus \mathcal{H}_{\lambda h}$, where the direct sum is taken over a single value $\lambda \in Q$ possibly with multiplicity (that is, $\lambda \in Q$ may be counted with multiplicity). Assume that $\langle \Psi, \tilde{\sigma}_{\lambda h}(x) \Phi \rangle = 0$ for all $x \in \mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-)$ and $I_+, I_- \in \mathcal{I}$, where $\tilde{\sigma}_{\lambda h}$ is the direct sum of copies of $\sigma_{\lambda h}$ over the single value λ repeated in Q in the sense above. Then $\langle \Psi, y \Phi \rangle = 0$ holds for all $y \in \overline{\bigcup_{I_{\pm} \in \mathcal{I}} \tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-))}$, where the closure is taken in the weak operator topology on \mathcal{H} .

Let $\phi_1, \phi_2 \in \mathcal{B}(O)$ with charge λ as in the previous paragraph. As lightlike translations $U(T_t \times \iota)$ belong to $\bigcup_{I_{\pm} \in \mathcal{I}} \tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-))$ as \mathcal{B} is a conformal extension of $\mathcal{A}_{\mathfrak{h}_p}$ (see the beginning of this Section), and

$$\phi_1 U(T_t \times \iota) \phi_2^* = U(T_t \times \iota) \cdot U(T_t \times \iota)^* \phi_1 U(T_t \times \iota) \phi_2^* \in \overline{\bigcup_{I_+ \in \mathcal{I}} \tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-))},$$

thus $\langle \Psi, \phi_1 U(T_t \times \iota) \phi_2^* \Phi \rangle = 0$ for all t. By positivity of energy, this extends to the upper half plane, and then by Schwarz reflection principle, it is 0 for all $t \in \mathbb{R}$. Apply this also for $U(\iota \times T_t)$, and by the uniqueness of Ω , it must hold that $\langle \Psi, \phi_1 \Omega \rangle \langle \Omega \phi_2^*, \Phi \rangle = 0$. As Ω is cyclic for $\mathcal{B}(O)$ and ϕ_1, ϕ_2 were arbitrary operators with charge λ , we conclude that $\Psi = 0$.

By Lemma 5.3, Q is an additive subgroup of \mathbb{R}^N . If we assume that it is discrete, then by [Neu99, (4.2) Proposition], it is a (not necessarily N-dimensional) lattice in \mathfrak{h} . Recall that we defined simple (not twisted by the 2-cocycle) shift operators $\underline{\psi}^{\boldsymbol{\alpha}}$ for any lattice Q in (3.10). For any $x \in \mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-)$, it holds that $\psi^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \psi^{\boldsymbol{\alpha}*} = \tau_{\mathcal{B}}(\sigma_{-\boldsymbol{\alpha}h}(x)) \in \tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(I_+ \times I_-))$.

Lemma 5.4. Let Q be a lattice in \mathbb{R}^N and let \mathcal{B} be a (local) conformal extensions of $\tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p})$ on \mathcal{H}_Q defined in (3.6), where $\tau_{\mathcal{B}} = \bigoplus_{\boldsymbol{\lambda} \in Q} \sigma_{p\boldsymbol{\lambda}h} \otimes \sigma_{\bar{p}\boldsymbol{\lambda}h}$ and $h \in \mathcal{C}^{\infty}(S^1)$ has supp $h \subset I$ as before. Then for each $\boldsymbol{\alpha} \in Q$, there is a $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \in \mathcal{B}(I \times I)$ such that $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} = \bigoplus_{\boldsymbol{\lambda} \in Q} \epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda})$ for some $\epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda}) \in S^1$. Moreover, Q is an even lattice with respect to $(\cdot|\cdot)$.

Proof. We take an isometry $\phi^{\alpha} \in \mathcal{B}(I \times I, \operatorname{Ad} V, \lambda)$ as in Lemma 5.3, and we use the same notation introduced in its proof for $\sigma_{\lambda h}$ and $\mathcal{H}_{\lambda h}$. By assumption, $\underline{\psi}^{\alpha}\phi^{\alpha*}$ preserves each $\mathcal{H}_{\lambda h}$ and commutes with $\sigma_{\lambda h}(\mathcal{A}_{\mathfrak{h}_p}(I' \times I'))$, therefore, as each $\sigma_{\lambda h}$ is an automorphism, by Haag duality of $\mathcal{A}_{\mathfrak{h}_p}$, it holds that $\underline{\psi}^{\alpha}\phi^{\alpha*} = \bigoplus_{\lambda \in Q} \sigma_{\lambda h}(W_{\lambda})$ for some $W_{\lambda} \in \mathcal{A}_{\mathfrak{h}_p}(I \times I)$ depending on $\lambda \in Q$.

Let us consider $\mathbf{0} \in Q$ and set $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} := \tau_{\mathcal{B}}(W_{\mathbf{0}})\phi^{\boldsymbol{\alpha}}$. Then, denoted by $P_{\mathbf{0}}$ the orthogonal projection onto $\mathcal{H}_{\mathbf{0}h}$, it holds that

$$P_{\mathbf{0}}\psi_{\mathcal{B}}^{\alpha} = P_{\mathbf{0}}\sigma_{\mathbf{0}}(W_{\mathbf{0}})\phi^{\alpha}$$
$$= P_{\mathbf{0}}\psi^{\alpha}$$

therefore, for any $x \in \mathcal{A}_{\mathfrak{h}_p}$ we have

$$P_{\mathbf{0}}(\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*}) = P_{\mathbf{0}}(\psi^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \psi^{\boldsymbol{\alpha}*}).$$

Note that, by taking $x \in \mathcal{A}_{\mathfrak{h}_p}(\tilde{O})$, where $\overline{I \times I} \subset \tilde{O}$, we have $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*}, \underline{\psi}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \underline{\psi}^{\boldsymbol{\alpha}*} \in \mathcal{A}_{\mathfrak{h}_p}(\tilde{O})$. By the separating property of the vacuum, we have that $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x) \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} = \underline{\psi}^{\boldsymbol{\alpha}} \cdot \overline{\tau_{\mathcal{B}}}(x) \cdot \underline{\psi}^{\boldsymbol{\alpha}*}$, or equivalently,

$$\tau_{\mathcal{B}}(x) \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} = \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x).$$

Moreover, if $x' \in \mathcal{A}_{\mathfrak{h}_p}(I' \times I')$, then by relative locality and the fact that $\tau_{\mathcal{B}}$ is localized in $I \times I$ (by the choice of localization of the function h), we have

$$\tau_{\mathcal{B}}(x') \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} = \psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} \cdot \tau_{\mathcal{B}}(x').$$

As $\mathcal{A}_{\mathfrak{h}_p}(\tilde{O})$ and $\mathcal{A}_{\mathfrak{h}_p}(I' \times I')$ generate the whole net $\mathcal{A}_{\mathfrak{h}_p}$, this shows that $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} \underline{\psi}^{\boldsymbol{\alpha}} = \bigoplus_{\boldsymbol{\lambda} \in Q} \epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda})$ for some $\epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda}) \in \mathbb{C}$. As $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}$ is an isometry and $\underline{\psi}^{\boldsymbol{\alpha}}$ is unitary, we have $|\epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda})| = 1$, thus $\epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda}) \in S^1$.

Recall that \mathcal{B} is local. Then, for any $\gamma_+, \gamma_- \in \overline{\mathrm{Diff}_+(S^1)} \times \overline{\mathrm{Diff}_+(S^1)}$ that map $I \times I$ to $I_+ \times I_-$, where I_+ is in the past of I and I_- is in the future of I, Ad $U(\gamma_+ \times \gamma_-)(\psi^{\alpha}_{\mathcal{B}})$ and $\psi^{\alpha}_{\mathcal{B}}$ must commute. On the other hand, as each $\epsilon_{\mathcal{B}}(\alpha, \lambda)$ clearly commutes with $U(\gamma_+ \times \gamma_-)|_{\mathcal{H}_{\lambda h}}$ for every $\lambda \in Q$, arguing as for (3.11), we get

$$\psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \cdot \operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}) = \tau_{\mathcal{B}} \left(\sigma_{p\boldsymbol{\alpha}h}((z_{p\boldsymbol{\alpha}h}(\gamma_{+})) \otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}(z_{\bar{p}\boldsymbol{\alpha}h}(\gamma_{-})) \right) \psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \psi_{\mathcal{B}}^{\boldsymbol{\alpha}}$$

$$\operatorname{Ad} U_{Q}(\gamma_{+} \times \gamma_{-})(\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}) \cdot \psi_{\mathcal{B}}^{\boldsymbol{\alpha}} = \tau_{\mathcal{B}} \left(z_{p\boldsymbol{\alpha}h}(\gamma_{+}) \otimes z_{\bar{p}\boldsymbol{\alpha}h}(\gamma_{-}) \right) \psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \psi_{\mathcal{B}}^{\boldsymbol{\alpha}}$$

$$(5.1)$$

By the definition of braiding between the automorphisms of $\mathcal{A}_{\mathfrak{h}_n}$, see (2.1) and (2.3),

$$\tau_{\mathcal{B}}(z_{p\boldsymbol{\alpha}h}(\gamma_{+})\otimes z_{\bar{p}\boldsymbol{\alpha}h}(\gamma_{-}))^{*}\tau_{\mathcal{B}}(\sigma_{p\boldsymbol{\alpha}h}((z_{p\boldsymbol{\alpha}h}(\gamma_{+}))\otimes \sigma_{\bar{p}\boldsymbol{\alpha}h}(z_{\bar{p}\boldsymbol{\alpha}h}(\gamma_{-})))=\varepsilon_{p\boldsymbol{\alpha}h,p\boldsymbol{\alpha}h}^{+}\varepsilon_{\bar{p}\boldsymbol{\alpha}h,\bar{p}\boldsymbol{\alpha}h}^{-}=e^{\mathrm{i}\pi(\alpha|\alpha)}.$$

Thus, it must hold that $(\alpha|\alpha) \in 2\mathbb{Z}$ to have locality of \mathcal{B} .

For any pair $\boldsymbol{\alpha}, \boldsymbol{\beta} \in Q$, it holds that $(\boldsymbol{\alpha} + \boldsymbol{\beta} | \boldsymbol{\alpha} + \boldsymbol{\beta}) = (\boldsymbol{\alpha} | \boldsymbol{\alpha}) + 2(\boldsymbol{\alpha} | \boldsymbol{\beta}) + (\boldsymbol{\beta} | \boldsymbol{\beta}) \in 2\mathbb{Z}$, hence $(\boldsymbol{\alpha} | \boldsymbol{\beta}) \in \mathbb{Z}$. That is, Q is an even lattice with respect to $(\cdot | \cdot)$.

Theorem 5.5. Let $\mathcal{A}_{ph} \otimes \mathcal{A}_{\bar{p}h} \subset \mathcal{B}$ be a local conformal extension and assume that $\mathcal{A}_{ph} \otimes \mathcal{A}_{\bar{p}h}$ on \mathcal{H} is a direct sum of $\mathcal{H}_{p\lambda h} \otimes \mathcal{H}_{\bar{p}\lambda h}$, for $\lambda \in Q \subset \mathbb{R}^N$ possibly counted with multiplicity and corresponding to the representations $\sigma_{p\lambda h} \otimes \sigma_{\bar{p}\lambda h}$. Assume that the set Q is discrete in \mathbb{R}^N . Then each λ appears with multiplicity 1, Q is a (not necessarily N-dimensional) even lattice in \mathbb{R}^N with respect to $(\cdot|\cdot)$ defined in Section 3.2, and \mathcal{B} is unitarily equivalent to the two-dimensional lattice conformal net \mathcal{A}_Q constructed in Section 3.3.

Proof. By assumption $\tau_{\mathcal{B}} = \bigoplus_{\boldsymbol{\lambda} \in Q} \sigma_{\boldsymbol{\lambda}h}$ and Q (disregarding multiplicities) is a discrete subset of \mathbb{R}^N containing $\boldsymbol{0}$. By Lemma 5.3 and Lemma 5.4, Q is an even lattice and the multiplicity of each $\boldsymbol{\lambda} \in Q$ in the decomposition of $\tau_{\mathcal{B}}$ is 1. Recall the short-hand notation introduced earlier for $\sigma_{\boldsymbol{\lambda}h}$ and $\mathcal{H}_{\boldsymbol{\lambda}h}$. To show that \mathcal{B} is unitarily equivalent to \mathcal{A}_Q , for each $\boldsymbol{\alpha} \in Q$ we take $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}$ as in Lemma 5.4, such that $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}*}\underline{\psi}^{\boldsymbol{\alpha}} = \bigoplus_{\boldsymbol{\lambda} \in Q} \epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda})$ and $\epsilon_{\mathcal{B}}(\boldsymbol{\alpha}, \boldsymbol{\lambda}) \in S^1$.

From this it follows that, for any $x \in \mathcal{A}_{\mathfrak{h}_p}(O)$, O a double cone in \mathbb{R}^{1+1} , we have $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}\tau_{\mathcal{B}}(x)\psi_{\mathcal{B}}^{\boldsymbol{\alpha}*} = \tau_{\mathcal{B}}(\sigma_{-\boldsymbol{\alpha}h}(x)) \in \tau_{\mathcal{B}}(\mathcal{A}_{\mathfrak{h}_p}(O))$, just as with $\underline{\psi}^{\boldsymbol{\alpha}}$. Then it follows that $\psi_{\mathcal{B}}^{\boldsymbol{\alpha}}\psi_{\mathcal{B}}^{\boldsymbol{\beta}} = Z(\boldsymbol{\alpha},\boldsymbol{\beta})\psi_{\mathcal{B}}^{\boldsymbol{\alpha}+\boldsymbol{\beta}}$, and a priori $Z(\boldsymbol{\alpha},\boldsymbol{\beta}) = \bigoplus_{\boldsymbol{\gamma} \in Q} Z(\boldsymbol{\alpha},\boldsymbol{\beta},\boldsymbol{\gamma})$ with $Z(\boldsymbol{\alpha},\overline{\boldsymbol{\beta}},\boldsymbol{\gamma}) \in \mathbb{C}$, but by the separating property of the vacuum, actually $Z(\boldsymbol{\alpha},\boldsymbol{\beta},\boldsymbol{\gamma})$ does not depend on $\boldsymbol{\gamma}$. Moreover, $Z(\boldsymbol{\alpha},\boldsymbol{\beta})$ satisfies the 2-cocycle equation. By repeating the calculation of (5.1) for possibly different $\boldsymbol{\alpha},\boldsymbol{\beta}$, and imposing locality, we obtain $Z(\boldsymbol{\alpha},\boldsymbol{\beta})Z(\boldsymbol{\beta},\boldsymbol{\alpha})^{-1} = \mathrm{e}^{\mathrm{i}(\boldsymbol{\alpha}|\boldsymbol{\beta})}$.

Recall that, on the same Hilbert space \mathcal{H}_Q , we defined in (3.8) the twisted shift operators $\psi^{\boldsymbol{\alpha}}$, $\boldsymbol{\alpha} \in Q$, which satisfy $\psi^{\boldsymbol{\alpha}}\psi^{\boldsymbol{\beta}} = \epsilon(\boldsymbol{\alpha},\boldsymbol{\beta})\psi^{\boldsymbol{\alpha}+\boldsymbol{\beta}}$, and we also have $\epsilon(\boldsymbol{\alpha},\boldsymbol{\beta})\epsilon(\boldsymbol{\beta},\boldsymbol{\alpha})^{-1} = e^{i(\boldsymbol{\alpha}|\boldsymbol{\beta})}$, see below (3.1).

The quotient of 2-cocycles is again a 2-cocycle, and $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta}) Z(\boldsymbol{\alpha}, \boldsymbol{\beta})^{-1}$ satisfies

$$\epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta}) Z(\boldsymbol{\alpha}, \boldsymbol{\beta})^{-1} (\epsilon(\boldsymbol{\beta}, \boldsymbol{\alpha}) Z(\boldsymbol{\beta}, \boldsymbol{\alpha})^{-1})^{-1} = 1.$$

By [Bau95, Lemma 3.4.2], it is 2-coboundary: $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\beta}) Z(\boldsymbol{\alpha}, \boldsymbol{\beta})^{-1} = \chi(\boldsymbol{\alpha}) \chi(\boldsymbol{\beta}) \chi(\boldsymbol{\alpha} + \boldsymbol{\beta})^{-1}$ for some function $\chi: Q \to S^1$. This means that $\psi_{\mathcal{B}}^{\prime \boldsymbol{\alpha}} := \chi(\boldsymbol{\alpha}) \psi_{\mathcal{B}}^{\boldsymbol{\alpha}} \in \mathcal{B}(I \times I)$ satisfy

$$\psi_{\mathcal{B}}^{\prime \alpha} \psi_{\mathcal{B}}^{\prime \beta} = \epsilon(\alpha, \beta) \psi_{\mathcal{B}}^{\prime \alpha + \beta}. \tag{5.2}$$

Note that $\psi_{\mathcal{B}}^{\mathbf{0}} = \mathbb{1} = \psi^{\mathbf{0}}$. Furthermore, by (5.2),

$$\psi_{\mathcal{B}}^{\prime \alpha} = \epsilon(\alpha, \beta) \psi_{\mathcal{B}}^{\prime \alpha + \lambda} (\psi_{\mathcal{B}}^{\prime \lambda})^*, \tag{5.3}$$

therefore, $\psi_{\mathcal{B}}^{\prime \alpha}$ is completely determined by $\{\psi_{\mathcal{B}}^{\prime \lambda}|_{\mathcal{H}_{\mathbf{0}h}}\}_{\lambda \in Q}$ and ϵ .

Now we can write $\psi_{\mathcal{B}}^{\alpha} = \bigoplus_{\lambda \in Q} \epsilon_{\mathcal{B}}^{\prime}(\alpha, \lambda) \psi^{\alpha}$ for some $\epsilon_{\mathcal{B}}^{\prime}(\alpha, \lambda) \in \mathbb{C}$. It follows that $\epsilon_{\mathcal{B}}^{\prime}(\mathbf{0}, \mathbf{0}) = 1$. Therefore, with the unitary $V := \bigoplus_{\lambda} \epsilon_{\mathcal{B}}^{\prime}(\lambda, \lambda)$, it holds that

$$\operatorname{Ad}V(\psi^{\lambda})|_{\mathcal{H}_{\mathbf{0}h}} = \epsilon'_{\mathcal{B}}(\lambda, \lambda)\overline{\epsilon'_{\mathcal{B}}(\mathbf{0}, \mathbf{0})}\psi^{\lambda}|_{\mathcal{H}_{\mathbf{0}h}} = \epsilon'_{\mathcal{B}}(\lambda, \lambda)\psi^{\lambda}|_{\mathcal{H}_{\mathbf{0}h}} = \psi'^{\lambda}_{\mathcal{B}}|_{\mathcal{H}_{\mathbf{0}h}}$$

As $\operatorname{Ad} V(\psi^{\boldsymbol{\alpha}})|_{\mathcal{H}_{\boldsymbol{0}h}}$ has the same cocycle ϵ , it follows from the remark below (5.3) that $\operatorname{Ad} V(\psi^{\boldsymbol{\alpha}}) = \psi_{\mathcal{B}}^{\prime \boldsymbol{\alpha}}$. As $\operatorname{Ad} V$ commutes with $\tau_{\mathcal{B}}(x) = \tau_{Q}(x)$ for any $x \in \mathcal{A}_{\mathfrak{h}_{p}}(O)$, we conclude that two extensions $\mathcal{A}_{\mathfrak{h}_{p}} \subset \mathcal{B}$ and $\mathcal{A}_{\mathfrak{h}_{p}} \subset \mathcal{A}_{Q}$ are unitarily equivalent.

Remark 5.6. In general, our technical condition, that the representation of $\mathcal{A}_{ph} \otimes \mathcal{A}_{\bar{p}h}$ is a direct sum of irreducible representations and its spectrum is discrete in \mathbb{R}^N , is not true: We have counterexamples from Example 3.5 (see [MTW18, Section 6] for details), where the spectrum is the diagonal embedding of \mathbb{Q} in \mathbb{R}^2 , which is clearly not discrete. We do not know whether one can exclude direct integral instead of direct sum, which can happen e.g. in the Liouville theory (see [GKR24, Theorem 6.4, spectral decomposition]).

On the other hand, we observe that some compactness or nuclearity conditions imply our technical condition. For example, if one assumes that $e^{-\beta H}$ is compact for $\beta > 0$, where $H = L_0 \otimes \mathbb{1} + \mathbb{1} \otimes L_0$ is the conformal Hamiltonian, then in particular it must have discrete spectrum and in particular the representation of $\mathcal{A}_{ph} \otimes \mathcal{A}_{\bar{p}h}$ must be a direct sum (otherwise H and $e^{-\beta H}$ would have continuous decomposition and the latter cannot be compact) and the sectors must be discrete in \mathbb{R}^N (otherwise $e^{-\beta H}$ would not be compact).

6 Two-dimensional Wightman fields

Let $\mathfrak{h}, p, Q, \epsilon$ as in Section 3. Using the charge structures studied for conformal nets, we construct Wightman fields on the Hilbert space \mathcal{H}_Q . For this purpose, it is more convenient to take

$$\mathcal{K}_{Q} := \bigoplus_{\lambda \in Q} \mathcal{K}_{p\lambda} \otimes \mathcal{K}_{\bar{p}\lambda}, \tag{6.1}$$

where $\mathcal{K}_{p\lambda}$, $\mathcal{K}_{\bar{p}\lambda}$ are the completion of $M_{ph}(1,p\lambda)$, $M_{\bar{p}h}(1,\bar{p}\lambda)$ including the actions of the operators $p\boldsymbol{\alpha}(m)$, $\bar{p}\boldsymbol{\alpha}(m)$, thus $\mathcal{K}_{p\lambda} = \mathcal{H}_{\frac{p\lambda}{2\pi}}$ (the constant function $\frac{p\lambda}{2\pi}$), see Section 2.3.2. As we remarked there, $\mathcal{H}_{p\lambda h} \cong \mathcal{H}_{\frac{p\lambda}{2\pi}} = \mathcal{K}_{p\lambda}$ as representations of \mathcal{A}_{ph} , and similar for $\mathcal{A}_{\bar{p}h}$. In this sense, there is a natural unitary equivalence

$$\mathcal{H}_{Q} = \bigoplus_{\boldsymbol{\lambda} \in Q} \mathcal{H}_{p\boldsymbol{\lambda}h} \otimes \mathcal{H}_{\bar{p}\boldsymbol{\lambda}h} \cong \bigoplus_{\boldsymbol{\lambda} \in Q} \mathcal{K}_{p\boldsymbol{\lambda}} \otimes \mathcal{K}_{\bar{p}\boldsymbol{\lambda}} = \mathcal{K}_{Q}$$

intertwining the actions of $\mathcal{A}_{\mathfrak{h}_p}$.

We use the following convention of formal series of vectors or operators, cf. [AGT23, Section 5.2], [AMT24, Section 2.1]. Let z, \bar{z} be two independent formal variables⁵. For a family $\{A_{r,s}\}_{r,s\in\mathbb{R}}$ of vectors in W (we often take $W=\operatorname{End}(V)$ for some vector space V), we write $A(\underline{z})=\sum_{r,s}A_{s,r}z^{-r-1}\bar{z}^{-s-1}$. The elements $A_{r,s}$ are referred to as **Fourier components**. We note that in many cases $A_{r,s}=0$ whenever $r-s\notin\mathbb{Z}$.

6.1 Vertex operators

On each $M_{\mathfrak{h}}(1, \lambda)$ of Section 2.3.1, we use the same symbol for the operators $\boldsymbol{\alpha}(n)$, irrespectively of λ . We define formal series (with a slight abuse of notations " $\boldsymbol{\alpha}(\cdot)$ ")

$$oldsymbol{lpha}(z) := \sum_{n \in \mathbb{Z}} oldsymbol{lpha}(n) z^{-n-1},$$

$$E^+(oldsymbol{lpha}, z) := \exp\left(-\sum_{n \in \mathbb{Z}, n \geq 1} \frac{oldsymbol{lpha}(n)}{n} z^{-n}\right),$$

$$E^-(oldsymbol{lpha}, z) := \exp\left(\sum_{n \in \mathbb{Z}, n \geq 1} \frac{oldsymbol{lpha}(-n)}{n} z^n\right).$$

By [TL97, Chapter VI (1.2.2)], it holds that

$$\left(1 - \frac{z}{w}\right)^{-(\boldsymbol{\alpha}, \boldsymbol{\beta})_{\mathfrak{h}}} E^{+}(\boldsymbol{\alpha}, w) E^{-}(\boldsymbol{\beta}, z) = E^{-}(\boldsymbol{\beta}, z) E^{+}(\boldsymbol{\alpha}, w), \tag{6.2}$$

 $^{^5\}bar{z}$ is not the complex conjugate of z, although they are natural notations when one considers correlation functions, see [AMT24].

where $(1-u)^a = \sum_{n \in \mathbb{Z}, n > 0} {a \choose n} (-u)^n$ by definition, giving only integer powers of u.

Let $\widehat{\mathfrak{h}}_p$ be the infinite-dimensional Lie algebra generated by $p\alpha(m)$, $\bar{p}\alpha(n)$ and a central element K, where $\alpha \in \mathfrak{h}$, with the following commutation relations

$$[p\boldsymbol{\alpha}_{1}(m), p\boldsymbol{\alpha}_{2}(n)] = \delta_{m,-n}m \cdot (p\boldsymbol{\alpha}_{1}, p\boldsymbol{\alpha}_{2})_{\mathfrak{h}}K,$$

$$[\bar{p}\boldsymbol{\alpha}_{1}(m), \bar{p}\boldsymbol{\alpha}_{2}(n)] = \delta_{m,-n}m \cdot (\bar{p}\boldsymbol{\alpha}_{1}, \bar{p}\boldsymbol{\alpha}_{2})_{\mathfrak{h}}K,$$

$$[p\boldsymbol{\alpha}_{1}(m), \bar{p}\boldsymbol{\alpha}_{2}(n)] = 0.$$

There are modules of this algebra parametrized by $\lambda = p\lambda \oplus \bar{p}\lambda$, where $\lambda \in \mathfrak{h}$ again. Indeed, for each $\lambda \in \mathfrak{h}$, we denote $M_{\mathfrak{h}_p}(1,\lambda) = U(\mathfrak{h}_p)/J_{\lambda}$, where J_{λ} is the ideal generated by $p\alpha(m), \bar{p}\alpha(n), m, n > 0, p\alpha(0) - (p\lambda, p\alpha)_{\mathfrak{h}}, \bar{p}\alpha(0) - (\bar{p}\lambda, \bar{p}\alpha)_{\mathfrak{h}}, K - 1$. As $\mathfrak{h}_p = p\mathfrak{h} \oplus \bar{p}\mathfrak{h}$, there is a natural isomorphism $M_{\mathfrak{h}_p}(1,\lambda) \cong M_{p\mathfrak{h}}(1,p\lambda) \otimes M_{\bar{p}\mathfrak{h}}(1,\bar{p}\lambda)$ see Section 2.3.1.

On each $M_{\mathfrak{h}_p}(1,\boldsymbol{\lambda})$, we define formal series with two variables z,\bar{z} by

$$\boldsymbol{\alpha}(\underline{z}) = \sum_{z \in \mathbb{Z}} (p\boldsymbol{\alpha}(n)z^{-n-1} + \bar{p}\boldsymbol{\alpha}(n)\bar{z}^{-n-1}),$$

$$E^{+}(\boldsymbol{\alpha}, \underline{z}) = \exp\left(-\sum_{z \geq 1} \left(\frac{p\boldsymbol{\alpha}(n)}{n}z^{-n} + \frac{\bar{p}\boldsymbol{\alpha}(n)}{n}\bar{z}^{-n}\right)\right),$$

$$E^{-}(\boldsymbol{\alpha}, \underline{z}) = \exp\left(\sum_{z > 1} \left(\frac{p\boldsymbol{\alpha}(-n)}{n}z^{n} + \frac{\bar{p}\boldsymbol{\alpha}(-n)}{n}\bar{z}^{n}\right)\right).$$
(6.3)

With respect to the tensor product structure $M_{\mathfrak{h}_p}(1, \lambda) \cong M_{p\mathfrak{h}}(1, p\lambda) \otimes M_{\bar{p}\mathfrak{h}}(1, \bar{p}\lambda)$, we can write

$$\boldsymbol{\alpha}(\underline{z}) = p\boldsymbol{\alpha}(z) \otimes \mathbb{1} + \mathbb{1} \otimes \bar{p}\boldsymbol{\alpha}(\bar{z}),$$

$$E^{+}(\boldsymbol{\alpha}, \underline{z}) = E^{+}(p\boldsymbol{\alpha}, z) \otimes E^{+}(\bar{p}\boldsymbol{\alpha}, \bar{z}),$$

$$E^{-}(\boldsymbol{\alpha}, \underline{z}) = E^{-}(p\boldsymbol{\alpha}, z) \otimes E^{-}(\bar{p}\boldsymbol{\alpha}, \bar{z}).$$

$$(6.4)$$

With respect to this decomposition, we can construct a pair of the Virasoro algebra representations by the Sugawara formula, which we denote $L_0 \otimes \mathbb{1}, \mathbb{1} \otimes L_0$, respectively (where we omitted the dependence on λ).

Let us write $\underline{z}^{\boldsymbol{\alpha}(0)} := z^{p\boldsymbol{\alpha}(0)} \bar{z}^{\bar{p}\boldsymbol{\alpha}(0)}$, where $p\boldsymbol{\alpha}(0)|_{\mathcal{K}_{p\boldsymbol{\lambda}}} = (p\boldsymbol{\alpha}, p\boldsymbol{\lambda})_{\mathfrak{h}}$ and $\bar{p}\boldsymbol{\alpha}(0)|_{\mathcal{K}_{\bar{p}\boldsymbol{\lambda}}} = (\bar{p}\boldsymbol{\alpha}, \bar{p}\boldsymbol{\lambda})_{\mathfrak{h}}$ on each $\boldsymbol{\lambda} \in Q$. We introduce the pre-vertex operators as in [AGT23, Section 6.4]:

$$\underline{Y_{p\boldsymbol{\alpha}}}(z) = E^{-}(p\boldsymbol{\alpha}, z)E^{+}(p\boldsymbol{\alpha}, z)z^{p\boldsymbol{\alpha}(0)}
\underline{Y_{\bar{p}\boldsymbol{\alpha}}}(\bar{z}) = E^{-}(\bar{p}\boldsymbol{\alpha}, \bar{z})E^{+}(\bar{p}\boldsymbol{\alpha}, \bar{z})\bar{z}^{\bar{p}\boldsymbol{\alpha}(0)}
\underline{Y_{\boldsymbol{\alpha}}}(\underline{z}) = E^{-}(\boldsymbol{\alpha}, \underline{z})E^{+}(\boldsymbol{\alpha}, \underline{z})\underline{z}^{p\boldsymbol{\alpha}(0)} \cong \underline{Y_{p\boldsymbol{\alpha}}}(z) \otimes \underline{Y_{\bar{p}\boldsymbol{\alpha}}}(\bar{z}),$$
(6.5)

where the last tensor product refers to $\left(\bigoplus_{p_{\lambda}} M_{p\mathfrak{h}}(1, p_{\lambda})\right) \otimes \left(\bigoplus_{\bar{p}_{\lambda}} M_{\bar{p}\mathfrak{h}}(1, \bar{p}_{\lambda})\right)$, but it can be restricted to $\bigoplus_{\lambda \in Q} M_{\mathfrak{h}_p}(1, \lambda) \cong \bigoplus_{\lambda \in Q} M_{p\mathfrak{h}}(1, p_{\lambda}) \otimes M_{\bar{p}\mathfrak{h}}(1, \bar{p}_{\lambda})$.

6.2 Formal Wightman fields

On $\Psi = (\Psi_{\lambda})_{\lambda \in Q} \in \mathcal{K}_Q$ (see (6.1)), we take the twisted shift operator c_{α} defined by

$$(c_{\alpha}\Psi)_{\alpha+\lambda} = \epsilon(\alpha, \lambda)\Psi_{\lambda} \tag{6.6}$$

(we use a notation different from (3.8), in order to distinguish the specific equivalence classes \mathcal{K}_{λ} from $\mathcal{H}_{\lambda h}$). Then it holds that (cf. [AGT23, Section 6.4])

$$c_{\boldsymbol{\alpha}}c_{\boldsymbol{\beta}} = (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})}c_{\boldsymbol{\beta}}c_{\boldsymbol{\alpha}},$$

$$c_{\boldsymbol{\alpha}}\underline{z}^{\boldsymbol{\beta}(0)} = z^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}}\bar{z}^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}}\underline{z}^{\boldsymbol{\beta}(0)}c_{\boldsymbol{\alpha}}.$$
(6.7)

The formal series for our Wightman fields are given by

$$Y_{\alpha}(\underline{z}) := c_{\alpha} Y_{\underline{\alpha}}(\underline{z}). \tag{6.8}$$

There are only countably many non-zero Fourier components $Y_{\boldsymbol{\alpha},r,s}$ of $Y_{\boldsymbol{\alpha}}(\underline{z}) = \sum_{r,s} Y_{\boldsymbol{\alpha},r,s} z^{-r-1} \bar{z}^{-s-1}$ and when restricted to $M_{\mathfrak{h}_p}(1,\boldsymbol{\lambda})$, its image is $M_{\mathfrak{h}_p}(1,\boldsymbol{\alpha}+\boldsymbol{\lambda})$ and only the components $Y_{\boldsymbol{\alpha},r,s}$ such that $r + \frac{(p\boldsymbol{\alpha},p\boldsymbol{\lambda})_{\mathfrak{h}}}{2}$, $s + \frac{(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\lambda})_{\mathfrak{h}}}{2} \in \mathbb{Z}$ are non-zero.

Compared with [AGT23, Section 6.4], our Y_{α} has more tensor components in each chiral and antichiral components, and c_{α} is twisted by the cocycle ϵ . Note that the cocycle ϵ takes values in $\{-1,1\}$, and we have the action of the pair of the Virasoro algebras $L_m \otimes 1, 1 \otimes L_m$ on \mathcal{K}_Q , thus the commutation relations remain the same (cf. [AGT23, below (6.5)]):

$$\boldsymbol{\alpha}(m)c_{\boldsymbol{\beta}} = c_{\boldsymbol{\beta}}(\boldsymbol{\alpha}(m) + (\boldsymbol{\alpha}, \boldsymbol{\beta})_{\mathfrak{h}}\delta_{m,0}),$$

$$L_{m} \otimes \mathbb{1} \cdot c_{\boldsymbol{\alpha}} = c_{\boldsymbol{\alpha}}(L_{m} \otimes \mathbb{1} + p\boldsymbol{\alpha}(m)) + \frac{(p\boldsymbol{\alpha}, p\boldsymbol{\alpha})_{\mathfrak{h}}}{2}\delta_{m,0},$$

$$\mathbb{1} \otimes L_{m} \cdot c_{\boldsymbol{\alpha}} = c_{\boldsymbol{\alpha}}(\mathbb{1} \otimes L_{m} + \bar{p}\boldsymbol{\alpha}(m)) + \frac{(\bar{p}\boldsymbol{\alpha}, \bar{p}\boldsymbol{\alpha})_{\mathfrak{h}}}{2}\delta_{m,0}.$$

From this, it is analogous to [AGT23, Section 6.4] to show that Y_{α} are primary with the scaling dimensions $(\frac{(p\alpha,p\alpha)_{\mathfrak{h}}}{2},\frac{(\bar{p}\alpha,\bar{p}\alpha)_{\mathfrak{h}}}{2})$:

$$[L_m \otimes 1, Y_{\alpha}(\underline{z})] = \partial_z Y_{\alpha}(\underline{z}) z^{m+1} + \frac{(p\alpha, p\alpha)_{\mathfrak{h}}}{2} (m+1) Y_{\alpha}(\underline{z}), \tag{6.9}$$

$$[\mathbb{1} \otimes L_m, Y_{\alpha}(\underline{z})] = \partial_{\bar{z}} Y_{\alpha}(\underline{z}) \bar{z}^{m+1} + \frac{(\bar{p}\alpha, \bar{p}\alpha)_{\mathfrak{h}}}{2} (m+1) Y_{\alpha}(\underline{z}). \tag{6.10}$$

Using (6.8)(6.5)(6.4)(6.2)(6.7), we have (cf. [AGT23, (6.4)])

$$\left(1 - \frac{z}{w}\right)^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \left(1 - \frac{\bar{z}}{\bar{w}}\right)^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} w^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \bar{w}^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} Y_{\boldsymbol{\alpha}}(\underline{w}) Y_{\boldsymbol{\beta}}(\underline{z})$$

$$= (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \cdot \left(1 - \frac{w}{z}\right)^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \left(1 - \frac{\bar{w}}{\bar{z}}\right)^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} z^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \bar{z}^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} Y_{\boldsymbol{\beta}}(\underline{z}) Y_{\boldsymbol{\alpha}}(\underline{w}) \tag{6.11}$$

6.3 Wightman fields

We say that a formal series A(z) or $A(\bar{z})$ with coefficients in $\operatorname{End}(M_{p\mathfrak{h}}(1,p\boldsymbol{\lambda}))$ or $\operatorname{End}(M_{\bar{p}\mathfrak{h}}(1,\bar{p}\boldsymbol{\lambda}))$ satisfies polynomial energy bounds if $||A_s\Psi|| \leq C(|s|+1)^p||(L_0+1)^q\Psi||$. Similarly, we say that a formal series $B(\underline{z})$ in $\underline{z} = (z,\bar{z})$ with coefficients in $\operatorname{End}(\bigoplus_{\boldsymbol{\lambda}\in Q} M_{p\mathfrak{h}}(1,p\boldsymbol{\lambda})\otimes M_{\bar{p}\mathfrak{h}}(1,\bar{p}\boldsymbol{\lambda}))$ satisfies polynomial energy bounds if $||B_{s,t}\Psi|| \leq C(|s|+|t|+1)^p||(L_0\otimes \mathbb{I}+\mathbb{I}\otimes L_0)^q\Psi||$ for some C,p,q>0.

By [TL97, Proposition VI.1.2.1] [Gui19, Theorem A.2], $\underline{Y_{p\alpha}}(z)$, $\underline{Y_{\bar{p}\alpha}}(\bar{z})$ satisfies polynomial energy bounds. Thus by the arguments of [AGT23, Lemma 5.5] $\underline{Y_{\alpha}}(z)$ satisfies polynomial energy bounds as well.

Let us write

$$\mathscr{D}_{\lambda} = C^{\infty}(L_0|_{\mathcal{K}_{p\lambda}} \otimes \mathbb{1} + \mathbb{1} \otimes L_0|_{\mathcal{K}_{\bar{p}\lambda}}) \subset \mathcal{K}_{p\lambda} \otimes \mathcal{K}_{\bar{p}\lambda} \cong \mathcal{K}_{\lambda}$$

$$(6.12)$$

and $\mathscr{D}_Q = \bigoplus_{\lambda \in Q} \mathscr{D}_{\lambda}$ (the algebraic direct sum). By polynomial energy bounds, the Fourier components of $Y_{\alpha}(\underline{z})$, which are first defined on $M_{ph}(1, p\lambda) \otimes M_{\bar{p}h}(1, \bar{p}\lambda)$, extend to \mathscr{D}_{λ} , thus to \mathscr{D}_Q . Moreover, for a test function f on $(-\pi, \pi) \times (-\pi, \pi)$, define

$$f_{r,s} = \frac{1}{(2\pi)^2} \int_{-\pi}^{\pi} \int_{-\pi}^{\pi} f(\theta_+, \theta_-) e^{-i(r\theta_+ + s\theta_-)} d\theta_+ d\theta_-.$$

Then the sum

$$Y_{\boldsymbol{\alpha}}(f) := \sum_{r,s \in \mathbb{R}} Y_{\boldsymbol{\alpha},r,s} f_{r,s}$$

converges on each \mathcal{D}_{λ} of (6.12), thus on \mathcal{D}_{Q} .

Recall that we defined conformal Wightman fields on the Einstein cylinder in [AGT23, Section 5.1], by requiring locality, diffeomorphism covariance, positivity of energy, vacuum and the Reeh-Schlieder property and polynomial energy bounds for a family of such fields. Among them, the cyclicity of vacuum is required for a family of fields and not for single fields (see [AGT23, (2dW4)]). For simplicity, we include the fields $\alpha(f)$, $\alpha \in \mathfrak{h}_p$.

Theorem 6.1. The family $Y_{\alpha}(f)$, $\alpha \in Q$ with $\alpha(f)$, $\alpha \in \mathfrak{h}_p$, together with $U_Q = \bigoplus_{\lambda \in Q} U_{p\lambda} \otimes U_{\bar{p}\lambda}$ (cf. Section 3.3) and Ω_0 , satisfies the above axioms for two-dimensional conformal Wightman fields.

Proof. As most of the proofs are analogous to [AGT23, Theorem 5.9, Theorem 5.7], we will be brief except for locality. We have checked polynomial energy bounds of all these fields. Positivity of energy follows by construction, cf. Theorem 3.2. Moreover, as we have (2dCN6), by covariance all the fields extend to the Einstein cylinder. As we include chiral and antichiral fields $\alpha(f)$, they span \mathcal{K}_0 from Ω_Q , and as $Y_{\alpha}(f)$ maps \mathcal{K}_{λ} to $\mathcal{K}_{\alpha+\lambda}$, their image exhausts \mathcal{K}_Q .

To verify diffeomorphism covariance, it is enough to consider diffeomorphisms of the form $\gamma_+ \times \iota$ or $\iota \times \gamma_-$. For this, we apply the arguments of [CKLW18, Proposition 6.4], [AGT23, Lemma 5.4]: from (6.9) the commutation relations with the scaling dimension $\frac{(p\alpha,p\alpha)_{\mathfrak{h}}}{2}$ follow:

$$i[L_1(f_1) \otimes \mathbb{1}, Y_{\alpha}(f_2)] = Y_{\alpha}(\frac{(p\alpha, p\alpha)_{\mathfrak{h}}}{2} f_1' f_2 - f_1 \partial_{\theta_+} f_2),$$

where $f_1 \in C_0^{\infty}(S^1)$, f_2 is a compactly supported test function on $(-\pi, \pi) \times (-\pi, \pi)$. From this, the covariance with respect to $U(\gamma_+ \times \iota)$ follows exactly as in [AGT23, Lemma 5.4]. The covariance with respect to $U(\iota \times \gamma_-)$ is similar. The covariance of the chiral and antichiral fields are known, e.g. [CKLW18, Proposition 6.4].

Locality between $\alpha(f), \beta(g)$ and between $\alpha(f), Y_{\beta}(g)$ is analogous to [AGT23]. In order to show locality between $Y_{\alpha}(f)$ and $Y_{\beta}(g)$, we take technical elements of [AGT23, Lemma 5.3]: On $\bigoplus_{\lambda \in Q} \mathcal{K}_{p\lambda}$, with $c'_{p\alpha}$ the simple shift operator by $p\alpha$ (without cocycle), we have the commutation relations between formal series, just as we did for $Y_{\alpha}(\underline{z})$ (even more easily)

$$\left(1 - \frac{z}{w}\right)^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} w^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} c'_{p\boldsymbol{\alpha}} \underline{Y_{p\boldsymbol{\alpha}}}(w) \cdot c'_{p\boldsymbol{\beta}} \underline{Y_{p\boldsymbol{\beta}}}(z) = \left(1 - \frac{w}{z}\right)^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} z^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} c'_{p\boldsymbol{\beta}} \underline{Y_{p\boldsymbol{\beta}}}(z) \cdot c'_{p\boldsymbol{\alpha}} \underline{Y_{p\boldsymbol{\alpha}}}(w).$$

By the same arguments as in [AGT23, Lemma 5.3], this translates into the commutation relations between operators: Let $f_+, g_+ \in C_0^{\infty}((-\pi, \pi))$ such that supp $f_+ < \text{supp } g_+$, then

$$c'_{p\boldsymbol{\alpha}}\underline{Y_{p\boldsymbol{\alpha}}}(f_+)c'_{p\boldsymbol{\beta}}\underline{Y_{p\boldsymbol{\beta}}}(g_+) = \lim_{\mathrm{Im}\,\zeta>0,\zeta\to-1}\zeta^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}}\cdot c'_{p\boldsymbol{\beta}}\underline{Y_{p\boldsymbol{\beta}}}(g_+)\underline{Y_{p\boldsymbol{\alpha}}}(f_+),$$

where ζ^s is defined as the analytic continuation of positive roots on $\mathbb{R}_{>0}$ to $\mathbb{C}\setminus\mathbb{R}_{\leq 0}$. These relations hold even when restricted to $\mathcal{K}_{p\lambda}$, and thus to $\mathcal{K}_{p\lambda}\otimes\mathcal{K}_{\bar{p}\lambda}$. Similarly, if supp $f_->$ supp g_- ,

$$c'_{\bar{p}\alpha}\underline{Y_{\bar{p}\alpha}}(f_{-})c'_{\bar{p}\beta}\underline{Y_{\bar{p}\alpha}}(g_{-}) = \lim_{\mathrm{Im}\,\zeta < 0,\zeta \to -1} \zeta^{-(\bar{p}\alpha,\bar{p}\beta)_{\mathfrak{h}}} \cdot c'_{p\beta}\underline{Y_{\bar{p}\beta}}(g_{-})c'_{p\alpha}\underline{Y_{\bar{p}\alpha}}(f_{-}).$$

On \mathcal{K}_{λ} , we have $Y_{\alpha}(\underline{z}) = c_{\alpha} \underline{Y_{p\alpha}}(z) \otimes \underline{Y_{\bar{p}\alpha}}(\bar{z})$, where c_{α} is the shift $c'_{p\alpha} \otimes c'_{p\alpha} : \mathcal{K}_{\lambda} \cong \mathcal{K}_{p\lambda} \otimes \mathcal{K}_{\bar{p}\lambda} \to \mathcal{K}_{\alpha+\lambda} \cong \mathcal{K}_{p(\alpha+\lambda)} \otimes \mathcal{K}_{\bar{p}(\alpha+\lambda)}$ followed by the scalar $\epsilon(\alpha, \lambda)$.

Recall that $c_{\alpha}c_{\beta} = (-1)^{(\alpha|\beta)}c_{\beta}c_{\alpha}$. Altogether, for test functions $f, g \in C_0^{\infty}((-\pi, \pi) \times (-\pi, \pi))$ with compact supports such that supp f is on the left of supp g, (6.11) implies on each \mathcal{K}_{λ} that

$$\begin{split} Y_{\boldsymbol{\alpha}}(f)Y_{\boldsymbol{\beta}}(g) &= (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \cdot \lim_{\operatorname{Im} \zeta > 0, \zeta \to -1} \zeta^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \cdot \lim_{\operatorname{Im} \zeta < 0, \zeta \to -1} \zeta^{-(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} \cdot Y_{\boldsymbol{\beta}}(g)Y_{\boldsymbol{\alpha}}(f) \\ &= (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \cdot \lim_{\operatorname{Im} \zeta > 0, \zeta \to -1} \zeta^{-(p\boldsymbol{\alpha},p\boldsymbol{\beta})_{\mathfrak{h}}} \cdot \lim_{\operatorname{Im} \zeta > 0, \zeta \to -1} \zeta^{(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\beta})_{\mathfrak{h}}} \cdot Y_{\boldsymbol{\beta}}(g)Y_{\boldsymbol{\alpha}}(f) \\ &= (-1)^{(\boldsymbol{\alpha}|\boldsymbol{\beta})} \cdot (-1)^{-(\boldsymbol{\alpha}|\boldsymbol{\beta})}Y_{\boldsymbol{\beta}}(g)Y_{\boldsymbol{\alpha}}(f) \\ &= Y_{\boldsymbol{\beta}}(g)Y_{\boldsymbol{\alpha}}(f), \end{split}$$

where in the second equality we calculate the limit on the circle $S^1 \setminus \{-1\}$ and used the fact that the limit $\zeta \to -1$, Im $\zeta < 0$ amounts to the limit $\zeta^{-1} \to -1$, Im $\zeta > 0$.

Example 6.2. Let us take the case of Example 3.3. The restriction of $\epsilon(\boldsymbol{\alpha}, \boldsymbol{\lambda})^{-1}Y_{\boldsymbol{\alpha}}(\underline{z})$ to $M_{\mathfrak{h}_p}(1, \boldsymbol{\lambda}) \subset \mathcal{K}_{\boldsymbol{\lambda}}$ has the form

$$\epsilon(\boldsymbol{\alpha}, \boldsymbol{\lambda})^{-1} Y_{\boldsymbol{\alpha}}(\underline{z}) = Y_{p\boldsymbol{\alpha}}(z) \otimes Y_{\bar{p}\boldsymbol{\alpha}}(\bar{z})$$

and the tensor component $Y_{p\boldsymbol{\alpha}}(z)$ can be identified (up to a change of notations) with $\mathcal{Y}_{p\boldsymbol{\alpha}}(c_{p\boldsymbol{\alpha}},x)$ of [Gui19, Theorem A.2(a)]. Thus its Fourier components are bounded uniformly in $\boldsymbol{\lambda}$, say by C, if $(p\boldsymbol{\alpha},p\boldsymbol{\alpha})_{\mathfrak{h}} \leq 1$ (similar for $Y_{\bar{p}\boldsymbol{\alpha}}(\bar{z})$).

If we have both $(p\boldsymbol{\alpha},p\boldsymbol{\alpha})_{\mathfrak{h}} \leq 1$, $(\bar{p}\boldsymbol{\alpha},\bar{p}\boldsymbol{\alpha})_{\mathfrak{h}} \leq 1$, then the smeared two-dimensional field $Y_{\boldsymbol{\alpha}}(f)$ is bounded (cf. [Reh97]). This holds if $\boldsymbol{\alpha} = \frac{R}{\sqrt{2}} \oplus \frac{R}{\sqrt{2}}$ with $R^2 \leq 2$ or $\boldsymbol{\alpha} = \frac{R^{-1}}{\sqrt{2}} \oplus \frac{R^{-1}}{\sqrt{2}}$ with $R^{-2} \leq 2$. Therefore, if $\frac{1}{\sqrt{2}} \leq R \leq \sqrt{2}$, they together generate the two-dimensional conformal net. By the uniqueness result, Theorem 5.5, the generated net must be \mathcal{A}_Q of Theorem 3.2.

Acknowledgements

We thank Sebastiano Carpi and Yasuyuki Kawahigashi for discussions about braided equivalences, Yuto Moriwaki for informing us of his results in [Mor23] and Mizuki Oikawa for discussions regarding [BKL15]. Y.T. gratefully acknowledges the hospitality of Department of Mathematics, Friedrich-Alexander-Universität at Erlangen-Nürnberg, where part of this work was carried out.

M.S.A. is a Humboldt Research Fellow supported by the Alexander von Humboldt Foundation and is partially supported by INdAM–GNAMPA Project "Ricostruzione di segnali, tramite operatori e frame, in presenza di rumore" CUP E53C23001670001 and GNAMPA–INdAM. L.G.

and Y.T. acknowledge support from the GNAMPA–INdAM project "Operator algebras and infinite quantum systems", CUP E53C23001670001, project GNAMPA 2024 "Probabilità Quantistica e Applicazioni", CUP E53C23001670001, from the MIUR Excellence Department Project MatMod@TOV awarded to the Department of Mathematics, University of Rome "Tor Vergata", CUP E83C23000330006 and from the Tor Vergata University of Rome funding OANGQS, CUP E83C25000580005.

References

- [AGT23] Maria Stella Adamo, Luca Giorgetti, and Yoh Tanimoto. Wightman fields for two-dimensional conformal field theories with pointed representation category. *Comm. Math. Phys.*, 404(3):1231–1273, 2023. https://arxiv.org/abs/2301.12310.
- [AGT25] Maria Stella Adamo, Luca Giorgetti, and Yoh Tanimoto. Representations of conformal nets associated with infinite-dimensional groups. 2025.
- [AMT24] Maria Stella Adamo, Yuto Moriwaki, and Yoh Tanimoto. Osterwalder—Schrader axioms for unitary full vertex operator algebras. 2024. https://arxiv.org/abs/2407.18222.
- [ANS22] Maria Stella Adamo, Karl-Hermann Neeb, and Jonas Schober. Reflection positivity and Hankel operators—the multiplicity free case. J. Funct. Anal., 283(2):Paper No. 109493, 50, 2022. https://arxiv.org/abs/2105.08522.
- [ANS25] Maria Stella Adamo, Karl-Hermann Neeb, and Jonas Schober. Reflection positivity and its relation to disc, half plane and the strip. *Expo. Math.*, 43(4):Paper No. 125660, 50, 2025. https://arxiv.org/abs/2407.21123.
- [Ara99] Huzihiro Araki. Mathematical theory of quantum fields, volume 101 of International Series of Monographs on Physics. Oxford University Press, New York, 1999. https://doi.org/10.1093/oso/9780198517733.001.0001.
- [Bau95] Hellmut Baumgärtel. Operator algebraic methods in quantum field theory. Akademie Verlag, Berlin, 1995. https://books.google.com/books?id=rq3vAAAAMAAJ.
- [BKL15] Marcel Bischoff, Yasuyuki Kawahigashi, and Roberto Longo. Characterization of 2D rational local conformal nets and its boundary conditions: the maximal case. *Doc. Math.*, 20:1137–1184, 2015. https://arxiv.org/abs/1410.8848.
- [BLM11] Henning Bostelmann, Gandalf Lechner, and Gerardo Morsella. Scaling limits of integrable quantum field theories. Rev. Math. Phys., 23(10):1115–1156, 2011. https://arxiv.org/abs/1105.2781.
- [BMT88] Detlev Buchholz, Gerhard Mack, and Ivan Todorov. The current algebra on the circle as a germ of local field theories. *Nuclear Phys. B Proc. Suppl.*, 5B:20-56, 1988. https://www.researchgate.net/publication/222585851.
- [BSM90] Detlev Buchholz and Hanns Schulz-Mirbach. Haag duality in conformal quantum field theory. Rev. Math. Phys., 2(1):105–125, 1990. https://www.researchgate.net/publication/246352668.
- [Car04] Sebastiano Carpi. On the representation theory of Virasoro nets. Comm. Math. Phys., 244(2):261–284, 2004. https://arxiv.org/abs/math/0306425.

- [CKLW18] Sebastiano Carpi, Yasuyuki Kawahigashi, Roberto Longo, and Mihály Weiner. From vertex operator algebras to conformal nets and back. *Mem. Amer. Math. Soc.*, 254(1213):vi+85, 2018. https://arxiv.org/abs/1503.01260.
- [CLTW12] Paolo Camassa, Roberto Longo, Yoh Tanimoto, and Mihály Weiner. Thermal states in conformal QFT. I. Comm. Math. Phys., 309(3):703-735, 2012. https://arxiv.org/abs/1101.2865.
- [CW] Sebastiano Carpi and Mihály Weiner. Local energy bounds and representations of conformal nets. in preparation.
- [CW05] Sebastiano Carpi and Mihály Weiner. On the uniqueness of diffeomorphism symmetry in conformal field theory. *Comm. Math. Phys.*, 258(1):203–221, 2005. https://arxiv.org/abs/math/0407190.
- [DFK04] C.D'Antoni, K.Fredenhagen, S.Köster. Implementation of conformal covariance by diffeomorphism symmetry. Lett. Math. Phys., 67(2004):239–247. https://arxiv.org/abs/math-ph/0312017.
- [DG13] Jan Dereziński and Christian Gérard. *Mathematics of quantization and quantum fields*. Cambridge Monographs on Mathematical Physics. Cambridge University Press, Cambridge, 2013. https://doi.org/10.1017/CB09780511894541.
- [DHR71] Sergio Doplicher, Rudolf Haag, and John E. Roberts. Local observables and particle statistics. I. Comm. Math. Phys., 23:199-230, 1971. http://projecteuclid.org/euclid.cmp/1103857630.
- [DMNO13] Alexei Davydov, Michael Müger, Dmitri Nikshych, and Victor Ostrik. The Witt group of non-degenerate braided fusion categories. *J. Reine Angew. Math.*, 677:135–177, 2013. https://arxiv.org/abs/1009.2117.
- [DNO13] Alexei Davydov, Dmitri Nikshych, and Victor Ostrik. On the structure of the Witt group of braided fusion categories. Selecta Math. (N.S.), 19(1):237–269, 2013. https://arxiv.org/abs/1109.5558.
- [DR90] Sergio Doplicher and John E. Roberts. Why there is a field algebra with a compact gauge group describing the superselection structure in particle physics. *Comm. Math. Phys.*, 131(1):51–107, 1990. http://projecteuclid.org/euclid.cmp/1104200703.
- [EGNO15] Pavel Etingof, Shlomo Gelaki, Dmitri Nikshych, and Victor Ostrik. *Tensor categories*, volume 205 of *Mathematical Surveys and Monographs*. American Mathematical Society, Providence, RI, 2015. https://math.mit.edu/~etingof/egnobookfinal.pdf.
- [FJ96] Klaus Fredenhagen and Martin Jörß. Conformal Haag-Kastler nets, pointlike localized fields and the existence of operator product expansions. *Comm. Math. Phys.*, 176(3):541–554, 1996. https://projecteuclid.org/euclid.cmp/1104286114.
- [FLM88] Igor Frenkel, James Lepowsky, and Arne Meurman. Vertex operator algebras and the Monster, volume 134 of Pure and Applied Mathematics. Academic Press, Inc., Boston, MA, 1988. https://www.sciencedirect.com/bookseries/pure-and-applied-mathematics/vol/134.
- [FRS89] K. Fredenhagen, K.-H. Rehren, and B. Schroer. Superselection sectors with braid group statistics and exchange algebras. I. General theory. *Comm. Math. Phys.*, 125(2):201–226, 1989. http://projecteuclid.org/euclid.cmp/1104179464.

- [FRS02] Jürgen Fuchs, Ingo Runkel, and Christoph Schweigert. TFT construction of RCFT correlators. I. Partition functions. Nuclear Phys. B, 646(3):353-497, 2002. https://arxiv.org/abs/hep-th/0204148.
- [GF93] Fabrizio Gabbiani and Jürg Fröhlich. Operator algebras and conformal field theory. *Comm. Math. Phys.*, 155(3):569–640, 1993. http://projecteuclid.org/euclid.cmp/1104253398.
- [GKR24] Colin Guillarmou, Antti Kupiainen, and Rémi Rhodes. Review on the probabilistic construction and conformal bootstrap in Liouville theory. 2024. https://arxiv.org/abs/2403.12780.
- [GLW98] D. Guido, R. Longo, and H.-W. Wiesbrock. Extensions of conformal nets and superselection structures. Comm. Math. Phys., 192(1):217–244, 1998. https://arxiv.org/abs/hep-th/9703129.
- [Gui19] Bin Gui. Energy bounds condition for intertwining operators of types B, C, and G_2 unitary affine vertex operator algebras. Trans. Amer. Math. Soc., 372(10):7371-7424, 2019. https://arxiv.org/abs/1809.07003.
- [Gui21] Bin Gui. Categorical extensions of conformal nets. Comm. Math. Phys., 383(2):763–839, 2021. https://arxiv.org/abs/1812.04470.
- [Haa96] Rudolf Haag. Local quantum physics. Springer-Verlag, Berlin, second edition, 1996. https://books.google.com/books?id=OlLmCAAAQBAJ.
- [HK07] Yi-Zhi Huang and Liang Kong. Full field algebras. Comm. Math. Phys., 272(2):345–396, 2007. https://arxiv.org/abs/math/0511328.
- [Kac98] Victor Kac. Vertex algebras for beginners, volume 10 of University Lecture Series. American Mathematical Society, Providence, RI, second edition, 1998. https://books.google.com/books?id=e-jxBwAAQBAJ.
- [KL04a] Yasuyuki Kawahigashi and Roberto Longo. Classification of local conformal nets. Case c < 1. Ann. of Math. (2), 160(2):493-522, 2004. https://arxiv.org/abs/math-ph/0201015.
- [KL04b] Yasuyuki Kawahigashi and Roberto Longo. Classification of two-dimensional local conformal nets with c < 1 and 2-cohomology vanishing for tensor categories. Comm. Math. Phys., 244(1):63–97, 2004. https://arxiv.org/abs/math-ph/0304022.
- [KLM01] Yasuyuki Kawahigashi, Roberto Longo, and Michael Müger. Multi-interval subfactors and modularity of representations in conformal field theory. *Comm. Math. Phys.*, 219(3):631–669, 2001. https://arxiv.org/abs/math/9903104.
- [Kon07] Liang Kong. Full field algebras, operads and tensor categories. Adv. Math., 213(1):271–340, 2007. https://arxiv.org/abs/math/0603065.
- [LM75] M. Lüscher and G. Mack. Global conformal invariance in quantum field theory. *Comm. Math. Phys.*, 41:203–234, 1975. http://projecteuclid.org/euclid.cmp/1103898909.
- [Lon03] Roberto Longo. Conformal subnets and intermediate subfactors. Comm. Math. Phys., 237(1-2):7–30, 2003. https://arxiv.org/abs/math/0102196.
- [LR97] R. Longo and J. E. Roberts. A theory of dimension. K-Theory, 11(2):103-159, 1997. https://arxiv.org/abs/funct-an/9604008.

- [LT18] Roberto Longo and Yoh Tanimoto. Rotational KMS states and type I conformal nets. Comm. Math. Phys., 357(1):249–266, 2018. https://arxiv.org/abs/1608.08903.
- [ML98] Saunders Mac Lane. Categories for the working mathematician, volume 5 of Graduate Texts in Mathematics. Springer-Verlag, New York, second edition, 1998. https://doi.org/10.1007/978-1-4757-4721-8.
- [Mor23] Yuto Moriwaki. Two-dimensional conformal field theory, full vertex algebra and current-current deformation. Adv. Math., 427:Paper No. 109125, 74, 2023. https://arxiv.org/abs/2007.07327.
- [MTW18] Vincenzo Morinelli, Yoh Tanimoto, and Mihály Weiner. Conformal covariance and the split property. Comm. Math. Phys., 357(1):379–406, 2018. https://arxiv.org/abs/1609.02196.
- [Neu99] Jürgen Neukirch. Algebraic number theory, volume 322 of Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, Berlin, 1999. https://doi.org/10.1007/978-3-662-03983-0.
- [Ost03] Victor Ostrik. Module categories, weak Hopf algebras and modular invariants. *Transform*. *Groups*, 8(2):177–206, 2003. https://arxiv.org/abs/math/0111139.
- [Reh97] K.-H. Rehren. Bounded Bose fields. Lett. Math. Phys., 40(4):299-306, 1997. https://arxiv.org/abs/hep-th/9605156.
- [Reh00] K.-H. Rehren. Chiral observables and modular invariants. Comm. Math. Phys., 208(3):689–712, 2000. https://arxiv.org/abs/hep-th/9903262.
- [RFFS07] Ingo Runkel, Jens Fjelstad, Jürgen Fuchs, and Christoph Schweigert. Topological and conformal field theory as Frobenius algebras. In *Categories in algebra*, geometry and mathematical physics, volume 431 of *Contemp. Math.*, pages 225–247. Amer. Math. Soc., Providence, RI, 2007. https://doi.org/10.1090/conm/431/08275.
- [Seg04] Graeme Segal. The definition of conformal field theory. In *Topology, geometry and quantum field theory*, volume 308 of *London Math. Soc. Lecture Note Ser.*, pages 421–577. Cambridge Univ. Press, Cambridge, 2004. https://doi.org/10.1017/CB09780511526398.
- [SFR06] Christoph Schweigert, Jürgen Fuchs, and Ingo Runkel. Categorification and correlation functions in conformal field theory. In *International Congress of Mathematicians. Vol. III*, pages 443–458. Eur. Math. Soc., Zürich, 2006. https://arxiv.org/abs/math/0602079.
- [SW00] R. F. Streater and A. S. Wightman. *PCT*, spin and statistics, and all that. Princeton Landmarks in Physics. Princeton University Press, Princeton, NJ, 2000. https://www.jstor.org/stable/j.ctt1cx3vcq.
- [TL97] Valerio Toledano Laredo. Fusion of positive energy representations of LSpin(2n). 1997. Ph.D. Thesis, University of Cambridge. https://arxiv.org/abs/math/0409044.
- [Was98] Antony Wassermann. Operator algebras and conformal field theory. III. Fusion of positive energy representations of LSU(N) using bounded operators. *Invent. Math.*, 133(3):467–538, 1998. https://arxiv.org/abs/math/9806031.
- [Wei06] Mihály Weiner. Conformal covariance and positivity of energy in charged sectors. Comm. Math. Phys., 265(2):493–506, 2006. https://arxiv.org/abs/math-ph/0507066.
- [Xu06] Feng Xu. Some computations in the cyclic permutations of completely rational nets. Comm. Math. Phys., 267(3):757–782, 2006. https://arxiv.org/abs/math/0511662.

[Zel17] Christoph Zellner. On the existence of regular vectors. In *Representation theory—current trends and perspectives*, EMS Ser. Congr. Rep., pages 747–763. Eur. Math. Soc., Zürich, 2017. https://arxiv.org/abs/1510.08727.